# UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

|  | FORM 10-Q   |   |  |                               |
|--|---|---|--|-------------------------------|
| QUARTERLY REPORT PURSUANT  □ 1934  | TO SECTION 13 OR 15(d) OF THE SE  | CURITIES EXCHA                                  | NGE ACT OF   |                               |
|  | For the quarterly period ended Sept   | ember 28, 2019                                  |  |                               |
|  | OR  |   |  |                               |
| TRANSITION REPORT PURSUANT   □ 1934  | TO SECTION 13 OR 15(d) OF THE SEC   | CURITIES EXCHA                                  | NGE ACT OF   |                               |
|  | For the transition period from _  | to  |  |                               |
|  | Commission File Number 00   | 1-35849   |  |                               |
|  | NV5 Global, Inc (Exact name of registrant as specifie   |   |  |                               |
| Delaware   |   |   | 45-3458017   |                               |
| (State or other jurisdiction   | on of   |   | (I.R.S. Employer   |                               |
| incorporation or organiza  | tion)   |   | Identification No.)  |                               |
| 200 South Park Road, Suite 35  | 50  |   | 33021  |                               |
| Hollywood, Florida<br>(Address of principal executiv   |   |   | (Zip Code)   |                               |
| _  | (954) 495-2112 (Registrant's telephone number, inclusion of the securities registered pursuant to Section |   | _  |                               |
| Title of each class  | Trading Symbol(s)   | Name of eac                                     | h exchange on which registered   |                               |
| Common Stock, \$0.01 par value   | NVEE  |   | Q Stock Market   |                               |
| during the preceding 12 months (or for such shorter for the past 90 days. Yes ⊠ No □   | strant has submitted electronically every Int   | file such reports), and teractive Data File req | (2) has been subject to such fili<br>uired to be submitted pursuant to | ng requirements o Rule 405 of |
| files). Yes ⊠ No □   | the preceding 12 months (or for such short  | er period that the regis                        | stant was required to subline suc                                      | 11                            |
| Indicate by check mark whether the regis<br>emerging growth company. See definitions of "lar<br>Rule 12b-2 of the Exchange Act.  | strant is a large accelerated filer, an accelerated eaccelerated filer," "accelerated filer," "s          |   |  |                               |
| Large accelerated filer  |   | Ac  | celerated Filer  | X                             |
| Non-accelerated filer  |   | Sm  | naller reporting company   |                               |
| Emerging growth company $\Box$   |   |   |  |                               |
| If an emerging growth company, indicate by check revised financial accounting standards provided pure Indicate by check mark whether the registrant is a As of November 6, 2019, there were 12,818,872 s | ursuant to Section 13(a) of the Exchange A shell company (as defined in Rule 12b-2 o                      | ct. □ f the Exchange Act).                      | Yes □ No ⊠   | ny new or                     |
|  |   |   |  |                               |

# NV5 GLOBAL, INC. INDEX

|                   |   | Page |
|-------------------|---|------|
| PART I – FINA     | ANCIAL INFORMATION  |      |
| ITEM 1            | FINANCIAL STATEMENTS  | 1    |
|                   | Consolidated Balance Sheets (unaudited)   | 1    |
|                   | Consolidated Statements of Net Income and Comprehensive Income (unaudited)            | 2    |
|                   | Consolidated Statement of Changes in Stockholders' Equity (unaudited)                 | 3    |
|                   | Consolidated Statements of Cash Flows (unaudited)                                     | 4    |
|                   | Notes to Consolidated Financial Statements (unaudited)                                | 6    |
| ITEM 2            | MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS | 24   |
| ITEM 3            | QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK                            | 34   |
| ITEM 4            | CONTROLS AND PROCEDURES   | 34   |
| PART II – OTI     | HER INFORMATION   |      |
| ITEM 1            | LEGAL PROCEEDINGS   | 35   |
| ITEM 1A           | RISK FACTORS  | 35   |
| ITEM 2            | UNREGISTERED SALES OF EQUITY SECURITIES AND USE OF PROCEEDS                           | 36   |
| ITEM 3            | <u>DEFAULTS UPON SENIOR SECURITIES</u>  | 36   |
| ITEM 4            | MINE SAFETY DISCLOSURES   | 36   |
| ITEM 5            | OTHER INFORMATION   | 36   |
| ITEM 6            | <u>EXHIBITS</u>   | 37   |
| <u>SIGNATURES</u> |   | 39   |

# PART I – FINANCIAL INFORMATION ITEM 1. FINANCIAL STATEMENTS.

## NV5 Global, Inc. and Subsidiaries CONSOLIDATED BALANCE SHEETS (UNAUDITED)

(in thousands, except share data)

|  | Septer | September 28, 2019 |    | December 29,<br>2018 |
|--|--------|--------------------|----|----------------------|
| Assets   |        |                    |    |                      |
| Current assets:  |        |                    |    |                      |
| Cash and cash equivalents  | \$     | 31,425             | \$ | 40,739               |
| Billed receivables, net  |        | 109,590            |    | 98,324               |
| Unbilled receivables, net  |        | 53,818             |    | 43,411               |
| Prepaid expenses and other current assets  |        | 9,198              |    | 2,582                |
| Total current assets   |        | 204,031            |    | 185,056              |
| Property and equipment, net  |        | 12,349             |    | 11,677               |
| Right-of-use lease asset, net  |        | 42,366             |    | _                    |
| Intangible assets, net   |        | 100,688            |    | 99,756               |
| Goodwill   |        | 158,423            |    | 140,930              |
| Other assets   |        | 2,886              |    | 2,002                |
| Total Assets   |        | 520,743            | \$ | 439,421              |
| Liabilities and Stockholders' Equity   |        |                    |    |                      |
| Current liabilities:   |        |                    |    |                      |
| Accounts payable   | \$     | 23,082             | \$ | 22,588               |
| Accrued liabilities  | Ψ      | 33,654             | Ψ  | 20,853               |
| Income taxes payable   |        |                    |    | 2,697                |
| Billings in excess of costs and estimated earnings on uncompleted contracts  |        | 2,241              |    | 7,625                |
| Client deposits  |        | 276                |    | 208                  |
| Current portion of contingent consideration  |        | 3,351              |    | 1,845                |
| Current portion of notes payable and other obligations   |        | 17,578             |    | 17,139               |
| Total current liabilities  |        | 80,182             |    | 72,955               |
| Contingent consideration, less current portion   |        | 2,195              |    | 2,853                |
| Long-term lease liability  |        | 32,781             |    | 2,633                |
| Notes payable and other obligations, less current portion  |        | 40,638             |    | 29,847               |
| Deferred income tax liabilities, net   |        | 16,881             |    | 16,224               |
| Total liabilities  |        | 172,676            |    | 121,879              |
| Commitments and contingencies  |        |                    |    |                      |
| Stockholders' equity:  |        |                    |    |                      |
| Preferred stock, \$0.01 par value; 5,000,000 shares authorized, no shares issued and outstanding   |        | _                  |    | _                    |
| Common stock, \$0.01 par value; 45,000,000 shares authorized, 12,818,919 and 12,550,711 shares issued and outstanding as of September 28, 2019 and December 29, 2018, respectively |        | 128                |    | 126                  |
| Additional paid-in capital   |        | 246,869            |    | 236,525              |
| Retained earnings  |        | 101,070            |    | 80,891               |
| Total stockholders' equity   |        | 348,067            |    | 317,542              |
| Total liabilities and stockholders' equity   | \$     | 520,743            | \$ | 439,421              |
| Tom. memiles and stockholders equity   | Ψ      | 320,173            | ψ  | 737,7421             |

# NV5 Global, Inc. and Subsidiaries CONSOLIDATED STATEMENTS OF NET INCOME AND COMPREHENSIVE INCOME (UNAUDITED)

(in thousands, except share data)

|   | <b>Three Months Ended</b> |                     |    |                      |    | Nine Months Ended     |    |                      |  |  |  |
|---|---------------------------|---------------------|----|----------------------|----|-----------------------|----|----------------------|--|--|--|
|   | Se                        | ptember 28,<br>2019 | S  | eptember 29,<br>2018 | S  | September 28,<br>2019 | Se | eptember 29,<br>2018 |  |  |  |
| Gross revenues  | \$                        | 131,032             | \$ | 104,185              | \$ | 376,340               | \$ | 302,737              |  |  |  |
| Direct costs (excluding depreciation and amortization): |                           |                     |    |                      |    |                       |    |                      |  |  |  |
| Salaries and wages                                      |                           | 40,426              |    | 34,475               |    | 113,762               |    | 98,542               |  |  |  |
| Sub-consultant services                                 |                           | 19,972              |    | 14,989               |    | 56,969                |    | 43,349               |  |  |  |
| Other direct costs                                      |                           | 7,139               |    | 4,747                |    | 25,244                |    | 13,539               |  |  |  |
| Total direct costs                                      |                           | 67,536              |    | 54,211               |    | 195,975               |    | 155,430              |  |  |  |
| Gross Profit  |                           | 63,496              |    | 49,974               |    | 180,365               |    | 147,307              |  |  |  |
| Operating Expenses:                                     |                           |                     |    |                      |    |                       |    |                      |  |  |  |
| Salaries and wages, payroll taxes and benefits          |                           | 33,428              |    | 24,897               |    | 93,431                |    | 76,122               |  |  |  |
| General and administrative                              |                           | 11,028              |    | 7,556                |    | 30,786                |    | 23,348               |  |  |  |
| Facilities and facilities related                       |                           | 4,664               |    | 3,490                |    | 12,407                |    | 10,552               |  |  |  |
| Depreciation and amortization                           |                           | 6,551               |    | 4,057                |    | 18,908                |    | 11,660               |  |  |  |
| Total operating expenses                                |                           | 55,671              |    | 40,000               |    | 155,533               |    | 121,682              |  |  |  |
| Income from operations                                  |                           | 7,825               |    | 9,974                |    | 24,832                |    | 25,625               |  |  |  |
| Interest expense  |                           | (421)               |    | (451)                |    | (1,230)               |    | (1,712)              |  |  |  |
| Income before income tax expense                        |                           | 7,403               |    | 9,523                |    | 23,602                |    | 23,913               |  |  |  |
| Income tax expense                                      |                           | (1,560)             |    | (2,238)              |    | (3,422)               |    | (4,716)              |  |  |  |
| Net Income and Comprehensive Income                     | \$                        | 5,843               | \$ | 7,285                | \$ | 20,180                | \$ | 19,197               |  |  |  |
| Earnings per share:                                     |                           |                     |    |                      |    |                       |    |                      |  |  |  |
| Basic   | \$                        | 0.48                | \$ | 0.65                 | \$ | 1.67                  | \$ | 1.80                 |  |  |  |
| Diluted   | \$                        | 0.46                | \$ | 0.62                 | \$ | 1.62                  | \$ | 1.71                 |  |  |  |
| Weighted average common shares outstanding:             |                           |                     |    |                      |    |                       |    |                      |  |  |  |
| Basic   |                           | 12,191,405          |    | 11,256,946           |    | 12,086,588            |    | 10,686,040           |  |  |  |
| Diluted   |                           | 12,566,966          |    | 11,701,394           |    | 12,485,049            |    | 11,205,748           |  |  |  |

## NV5 Global, Inc. and Subsidiaries CONSOLIDATED STATEMENTS OF CHANGES IN STOCKHOLDERS' EQUITY (UNAUDITED)

(in thousands, except share data)

|  | Comm         | Common Stock |        |    | Additional         |    |                      |               |
|--|--------------|--------------|--------|----|--------------------|----|----------------------|---------------|
|  | Shares       |              | Amount | •  | Paid-In<br>Capital |    | Retained<br>Earnings | Total         |
| Balance, June 30, 2018                         | 11,129,082   | \$           | 111    | \$ | 131,746            | \$ | 65,947               | \$<br>197,804 |
| Stock compensation                             | _            |              | _      |    | 1,902              |    | _                    | 1,902         |
| Restricted stock issuance, net                 | 25,672       |              | _      |    | (1)                |    | _                    | _             |
| Stock issuance for acquisitions                | 37,146       |              | _      |    | 2,886              |    | _                    | 2,886         |
| Proceeds from secondary offering, net of costs | 1,270,000    |              | 13     |    | 93,456             |    | _                    | 93,469        |
| Net income                                     | _            |              | _      |    | _                  |    | 7,285                | 7,285         |
| Balance, September 29, 2018                    | 12,461,900   | \$           | 125    | \$ | 229,989            | \$ | 73,232               | \$<br>303,346 |
|  |              |              |        |    |                    |    |                      |               |
| Balance, June 29, 2019                         | 12,657,841   | \$           | 127    | \$ | 243,646            | \$ | 95,228               | \$<br>339,001 |
| Stock compensation                             | <del>_</del> |              | _      |    | 2,819              |    | _                    | 2,819         |
| Restricted stock issuance, net                 | 155,307      |              | 1      |    | _                  |    | _                    | 1             |
| Stock issuance for acquisitions                | 5,771        |              | _      |    | 403                |    | _                    | 403           |
| Net income                                     | _            |              | _      |    | _                  |    | 5,843                | 5,843         |
| Balance, September 28, 2019                    | 12,818,919   | \$           | 128    | \$ | 246,869            | \$ | 101,070              | \$<br>348,067 |

|   | Common Stock |     |        | Additional<br>Paid-In |         |    | Retained |               |
|---|--------------|-----|--------|-----------------------|---------|----|----------|---------------|
|   | Shares       |     | Amount |                       | Capital |    | Earnings | Total         |
| Balance, December 30, 2017                            | 10,834,770   | \$  | 108    | \$                    | 125,954 | \$ | 54,035   | \$<br>180,097 |
| Stock compensation                                    | _            |     | _      |                       | 4,541   |    | _        | 4,541         |
| Restricted stock issuance, net                        | 144,118      |     | 2      |                       | (2)     |    | _        | _             |
| Stock issuance for acquisitions                       | 73,012       |     | 1      |                       | 4,949   |    |          | 4,950         |
| Proceeds from secondary offering, net of costs        | 1,270,000    |     | 13     |                       | 93,456  |    | _        | 93,469        |
| Proceeds from exercise of warrants, net of costs      | 140,000      |     | 1      |                       | 1,091   |    | _        | 1,092         |
| Net income  | _            |     | _      |                       | _       |    | 19,197   | 19,197        |
| Balance, September 29, 2018                           | 12,461,900   | \$  | 125    | \$                    | 229,989 | \$ | 73,232   | \$<br>303,346 |
|   |              | · · |        |                       |         |    |          |               |
| Balance, December 29, 2018                            | 12,550,711   | \$  | 126    | \$                    | 236,525 | \$ | 80,891   | \$<br>317,542 |
| Stock compensation                                    | _            |     | _      |                       | 6,989   |    |          | 6,989         |
| Restricted stock issuance, net                        | 215,431      |     | 2      |                       | (2)     |    |          | _             |
| Stock issuance for acquisitions                       | 41,592       |     | _      |                       | 2,632   |    |          | 2,632         |
| Payment of contingent consideration with common stock | 11,185       |     |        |                       | 725     |    |          | 725           |
| Net income  | _            |     | _      |                       | _       |    | 20,180   | 20,180        |
| Balance, September 28, 2019                           | 12,818,919   | \$  | 128    | \$                    | 246,869 | \$ | 101,070  | \$<br>348,067 |

## NV5 Global, Inc. and Subsidiaries CONSOLIDATED STATEMENTS OF CASH FLOWS (UNAUDITED) (in thousands)

| Net income         \$ 20,180 \$ 19,197           Adjustments to reconcile net income to net eash provided by operating activities:         Secondary of the provided of the p   |  | Nine Months Ended |               |                    |           |  |
|---|--|-------------------|---------------|--------------------|-----------|--|
| Net income         \$ 20,180 \$ 19,197           Adjustments to reconcile net income to net cash provided by operating activities:         8           Depreciation and amortization         18,808         11,600           Non-eash lease expense         6,770         —           Provision for doubtful accounts         1,725         843           Stock based compensation         6,989         4,544           Change in fair value of contingent consideration         49         267           Gain on disposals of property and equipment         (48)         —           Deferred income taxes         (3,389)         564           Changes in operating assets and liabilities, net of impact of acquisitions:         8         6,396           Billed receivables         58         6,396           Unbrilled receivables         4,490         1,375           Prepated expenses and other assets         1,279         819           Accounts payable         2,279         819           Accounts payable         2,197         485           Deposits         6,97         1,570           Net cash provided by operating activities         2,155         1,570           Cash Flows From Investing Activities:         2,2155         1,552           Cash paid for acqui   |  | Septen            | nber 28, 2019 | September 29, 2018 |           |  |
| Adjustments to reconcile net income to net cash provided by operating activities:   Depreciation and amortization   18,908   11,600   16,770   16,770   17,255   1843   17,255   1843   17,255   1843   17,255   1843   17,255   1843   17,255   1843   17,255   1843   17,255   1843   17,255   1843   17,255   1843   17,255   1843   17,255   1843   17,255   1843   17,255   1843   17,255   1843   17,255   1843   17,255   1843   18,000 | Cash Flows From Operating Activities:  |                   |               |                    |           |  |
| Depreciation and amortization         18,908         11,600           Non-cash lease expense         6,770         —           Provision for doubtful accounts         1,725         843           Stock based compensation         6,989         4,541           Change in fair value of contingent consideration         49         2,67           Caisn on disposals of property and equipment         (3,839)         564           Changes in operating assets and liabilities, net of impact of acquisitions:         80         6,396           Billed receivables         508         (6,396           Unbilled receivables         508         (6,396           Unbilled receivables         (4,490)         (3,759         819           Accounts payable         (2,053)         (6,799         Accounts payable         (2,789)         (6,713           Accused liabilities         (4,490)         (3,259         4,834         4,936         4,932           Recuest a provided by operating activities         (2,789)         (6,713         4,536         4,536         4,536         4,536         4,536         4,536         4,536         4,536         4,536         4,536         4,536         4,536         4,536         4,536         4,536         4,536         4,536   | Net income   | \$                | 20,180        | \$                 | 19,197    |  |
| Non-cash lease expense         6,770           Provision for doubful accounts         1,725         843           Stock based compensation         6,989         4,541           Change in fair value of contingent consideration         49         267           Gain on disposals of property and equipment         (88)         —           Deferred income taxes         (3,839)         564           Changes in operating assets and liabilities, net of impact of acquisitions:         818         (3,399)         63,396           Billed receivables         (4,490)         (3,759)         819           Account fail in the ceivables         (4,490)         (3,759)         819           Accounts payable         (2,053)         (679         819           Accounts payable         (2,789)         (6,713         69,713           Billings in excess of costs and estimated earnings on uncompleted contracts         (5,972)         485           Deposits         6         —         1,557         17,570           Cash Provided by operating activities         (2,365)         (2,840)         6,713           Cash paid for acquisitions (net of cash received from acquisitions)         (29,365)         (28,400)           Purchase of property and equipment         (1,810)         (1,582)   | Adjustments to reconcile net income to net cash provided by operating activities:  |                   |               |                    |           |  |
| Provision for doubtful accounts         1,725         843           Stock based compensation         6,989         4,541           Change in fair value of contingent consideration         (48)         —           Gain on disposals of property and equipment         (48)         —           Deferred income taxes         (3,839)         564           Changes in operating assets and liabilities, net of impact of acquisitions:         508         (6,396           Billed receivables         508         (6,396           Unbilled receivables         (4,490)         3,759           Prepaid expenses and other assets         (2,053)         (679           Accounts payable         (2,053)         (679           Accured liabilities         (9,170)         (3,259           Income taxes payable         (2,789)         (6,713           Billings in excess of costs and estimated earnings on uncompleted contracts         (5,972)         485           Deposits         68         —           Net cash provided by operating activities         21,557         17,570           Cash Flows From Investing Activities         (29,365)         (28,460           Purchase of property and equipment         (1,810)         (1,582           Net cash provided by operating activities   | Depreciation and amortization  |                   | 18,908        |                    | 11,660    |  |
| Stock based compensation         6,989         4,541           Change in fair value of contingent consideration         49         267           Gain on disposals of property and equipment         (48)         ————————————————————————————————————  | Non-cash lease expense   |                   | 6,770         |                    | _         |  |
| Change in fiair value of contingent consideration         49         267           Gain on disposals of property and equipment         (48)         —           Deferred income taxes         (3,839)         564           Changes in operating assets and liabilities, net of impact of acquisitions:         ***           Billed receivables         508         (6,396           Unbilled receivables         (5,279)         819           Prepaid expenses and other assets         (5,279)         819           Accounts payable         (2,053)         (679           Accrued liabilities         (9,170)         (3,259)           Income taxes payable         (2,789)         (6,713)           Billings in excess of costs and estimated earnings on uncompleted contracts         (5,972)         485           Deposits         68         —           Net eash provided by operating activities         21,557         17,570           Cash Flows From Investing Activities:           Cash paid for acquisitions (net of cash received from acquisitions)         (29,365)         (28,460)           Purchase of property and equipment         (1,810)         (1,582)           Net cash used in investing activities         31,175)         30,402           Cash Flows From Financing Activities:         <   | Provision for doubtful accounts  |                   | 1,725         |                    | 843       |  |
| Gain on disposals of property and equipment         (48)         —           Defered income taxes         (3,839)         564           Changes in operating assets and liabilities, net of impact of acquisitions:         508         (6,396)           Billed receivables         508         (6,396)           Unbilled receivables         (4,490)         (3,759)           Prepaid expenses and other assets         (2,053)         (679)           Accounts payable         (2,053)         (6,791)           Accrued liabilities         (9,170)         (3,259)           Income taxes payable         (2,789)         (6,713)           Billings in excess of costs and estimated earnings on uncompleted contracts         (5,972)         485           Deposits         68         —           Net eash provided by operating activities         21,557         17,570           Cash Flows From Investing Activities:         21,557         17,570           Cash paid for acquisitions (net of cash received from acquisitions)         (29,365)         (28,460)           Purchase of property and equipment         (1,810)         (1,582)           Net cash used in investing activities         (31,175)         30,042           Cash Flows From Financing Activities:           Proceeds from secondar   | Stock based compensation   |                   | 6,989         |                    | 4,541     |  |
| Deferred income taxes   | Change in fair value of contingent consideration   |                   | 49            |                    | 267       |  |
| Changes in operating assets and liabilities, net of impact of acquisitions:         508         (6,396           Billed receivables         (4,490)         (3,759)         819           Prepaid expenses and other assets         (5,279)         819           Accounts payable         (2,053)         (679           Accrued liabilities         (9,170)         (3,259)           Income taxes payable         (2,789)         (6,713)           Billings in excess of costs and estimated earnings on uncompleted contracts         (5,972)         485           Deposits         68         —           Net cash provided by operating activities         21,557         17,570           Cash Flows From Investing Activities:         22,557         17,570           Cash paid for acquisitions (net of eash received from acquisitions)         (29,365)         (28,460)           Purchase of property and equipment         (1,810)         (1,582)           Net cash used in investing activities         (31,175)         (30,042)           Cash Flows From Financing Activities:         —         93,469           Borrowings from Senior Credit Facility         10,000         —           Payments on notes payable         (8,483)         (7,410           Payments of contingent consideration         (1,213)  | Gain on disposals of property and equipment  |                   | (48)          |                    | _         |  |
| Billed receivables         508         (6,396           Unbilled receivables         (4,490)         (3,759           Prepaid expenses and other assets         (5,279)         819           Accounts payable         (2,053)         (679           Accured liabilities         (9,170)         (3,259           Income taxes payable         (2,789)         (6,713           Billings in excess of costs and estimated earnings on uncompleted contracts         (5,972)         485           Deposits         68         —           Net eash provided by operating activities         21,557         17,570           Cash Flows From Investing Activities:           Cash paid or acquisitions (net of cash received from acquisitions)         (29,365)         (28,400           Purchase of property and equipment         (1,810)         (1,582           Net cash used in investing activities         (31,175)         (30,042           Cash Flows From Financing Activities:         —         93,469           Deroceeds from secondary offering         —         93,469           Borrowings from Senior Credit Facility         10,000         —           Payments on notes payable         (8,483)         (7,410           Payments of contingent consideration         (1,213)   | Deferred income taxes  |                   | (3,839)       |                    | 564       |  |
| Unbilled receivables         (4,490)         (3,759)           Prepaid expenses and other assets         (5,279)         819           Accounts payable         (2,053)         (679           Accrued liabilities         (9,170)         (3,259           Income taxes payable         (2,789)         (6,713           Billings in excess of costs and estimated earnings on uncompleted contracts         (5,972)         485           Deposits         68         —           Net cash provided by operating activities         21,557         17,570           Cash Flows From Investing Activities:         29,365)         (28,460           Purchase of property and equipment         (1,810)         (1,582           Net cash used in investing activities         (31,175)         (30,042           Cash Flows From Financing Activities:         —         93,469           Porceeds from secondary offering         —         93,469           Borrowings from Senior Credit Facility         10,000         —           Payments on notes payable         (8,483)         (7,410           Payments of contingent consideration         (1,213)         (728           Proceeds from exercise of unit warrant         —         1,092           Payments of borrowings from Senior Credit Facility  | Changes in operating assets and liabilities, net of impact of acquisitions:  |                   |               |                    |           |  |
| Prepaid expenses and other assets         (5,279)         819           Accounts payable         (2,053)         (679           Accrued liabilities         (9,170)         (3,259           Income taxes payable         (2,789)         (6,713           Billings in excess of costs and estimated earnings on uncompleted contracts         (5,972)         485           Deposits         68         —           Net cash provided by operating activities         21,557         17,570           Cash Flows From Investing Activities:           Cash paid for acquisitions (red of cash received from acquisitions)         (29,365)         (28,460)           Purchase of property and equipment         (1,810)         (1,582)           Net cash used in investing activities         (31,175)         (30,042)           Cash Flows From Financing Activities:           Proceeds from Secondary offering         —         93,469           Borrowings from Senior Credit Facility         10,000         —           Payments of contingent consideration         (1,213)         (7,210           Payments of contingent consideration         (1,213)         (7,28           Proceeds from exercise of unit warrant         —         1,092           Payments of borrowings from Senior Cred   | Billed receivables   |                   | 508           |                    | (6,396)   |  |
| Accounts payable         (2,053)         (679           Accrued liabilities         (9,170)         (3,259           Income taxes payable         (2,789)         (6,713           Billings in excess of costs and estimated earnings on uncompleted contracts         (5,972)         485           Deposits         68         —           Net cash provided by operating activities         21,557         17,570           Cash Flows From Investing Activities:           Cash paid for acquisitions (net of cash received from acquisitions)         (29,365)         (28,460           Purchase of property and equipment         (1,810)         (1,582           Net cash used in investing activities         31,175         (30,042           Cash Flows From Financing Activities:         —         93,469           Borrowings from secondary offering         —         93,469           Borrowings from Senior Credit Facility         10,000         —           Payments of contingent consideration         (1,213)         (7,28           Proceeds from exercise of unit warrant         —         1,092           Payments of borrowings from Senior Credit Facility         —         36,500           Net cash provided by financing activities         304         49,923           Net (decrease) incre   | Unbilled receivables   |                   | (4,490)       |                    | (3,759)   |  |
| Accrued liabilities         (9,170)         (3,259)           Income taxes payable         (2,789)         (6,713)           Billings in excess of costs and estimated earnings on uncompleted contracts         (5,972)         485           Deposits         68         —           Net cash provided by operating activities         21,557         17,570           Cash Flows From Investing Activities:           Cash paid for acquisitions (net of cash received from acquisitions)         (29,365)         (28,460)           Purchase of property and equipment         (1,810)         (1,582)           Net cash used in investing activities         (31,175)         (30,042)           Cash Flows From Financing Activities:           Proceeds from secondary offering         —         93,469           Borrowings from Senior Credit Facility         10,000         —           Payments on notes payable         (8,483)         (7,410           Payments of contingent consideration         (1,213)         (728           Proceeds from exercise of unit warrant         —         (36,500)           Payments of borrowings from Senior Credit Facility         —         (36,500)           Net cash provided by financing activities         304         49,923           Net (decrease) increase in Cash   | Prepaid expenses and other assets  |                   | (5,279)       |                    | 819       |  |
| Income taxes payable  | Accounts payable   |                   | (2,053)       |                    | (679)     |  |
| Billings in excess of costs and estimated earnings on uncompleted contracts   6.8   —     Deposits   6.8   —     Net cash provided by operating activities   21,557   17,570     Cash Flows From Investing Activities:           Cash paid for acquisitions (net of eash received from acquisitions)   (29,365)   (28,460     Purchase of property and equipment   (1,810)   (1,582     Net cash used in investing activities   (31,175)   (30,042     Cash Flows From Financing Activities:         Proceeds from secondary offering   —   93,469     Borrowings from Senior Credit Facility   10,000   —     Payments on notes payable   (8,483)   (7,410     Payments of contingent consideration   (1,213)   (728     Proceeds from exercise of unit warrant   —   1,092     Payments of borrowings from Senior Credit Facility   —   (36,500     Net cash provided by financing activities   304   49,923     Net (decrease) increase in Cash and Cash Equivalents   (9,314)   37,451     Cash and cash equivalents – beginning of period   40,739   18,751  | Accrued liabilities  |                   | (9,170)       |                    | (3,259)   |  |
| Deposits         68         —           Net cash provided by operating activities         21,557         17,570           Cash Flows From Investing Activities:           Cash paid for acquisitions (net of cash received from acquisitions)         (29,365)         (28,460           Purchase of property and equipment         (1,810)         (1,582           Net cash used in investing activities         (31,175)         (30,042           Cash Flows From Financing Activities:           Proceeds from secondary offering         —         93,469           Borrowings from Senior Credit Facility         10,000         —           Payments on notes payable         (8,483)         (7,410           Payments of contingent consideration         (1,213)         (728           Proceeds from exercise of unit warrant         —         1,092           Payments of borrowings from Senior Credit Facility         —         (36,500           Net cash provided by financing activities         304         49,923           Net (decrease) increase in Cash and Cash Equivalents         (9,314)         37,451           Cash and cash equivalents – beginning of period         40,739         18,751   | Income taxes payable   |                   | (2,789)       |                    | (6,713)   |  |
| Net cash provided by operating activities       21,557       17,570         Cash Flows From Investing Activities:       29,365       (28,460         Purchase of property and equipment       (1,810)       (1,582         Net cash used in investing activities       (31,175)       (30,042         Cash Flows From Financing Activities:         Proceeds from secondary offering       —       93,469         Borrowings from Senior Credit Facility       10,000       —         Payments on notes payable       (8,483)       (7,410         Payments of contingent consideration       (1,213)       (728         Proceeds from exercise of unit warrant       —       1,092         Payments of borrowings from Senior Credit Facility       —       (36,500         Net cash provided by financing activities       304       49,923         Net (decrease) increase in Cash and Cash Equivalents       (9,314)       37,451         Cash and cash equivalents – beginning of period       40,739       18,751   | Billings in excess of costs and estimated earnings on uncompleted contracts  |                   | (5,972)       |                    | 485       |  |
| Cash Flows From Investing Activities:         Cash paid for acquisitions (net of cash received from acquisitions)       (29,365)       (28,460         Purchase of property and equipment       (1,810)       (1,582         Net cash used in investing activities       (31,175)       (30,042         Cash Flows From Financing Activities:         Proceeds from secondary offering       —       93,469         Borrowings from Senior Credit Facility       10,000       —         Payments on notes payable       (8,483)       (7,410         Payments of contingent consideration       (1,213)       (728         Proceeds from exercise of unit warrant       —       1,092         Payments of borrowings from Senior Credit Facility       —       (36,500         Net cash provided by financing activities       304       49,923         Net (decrease) increase in Cash and Cash Equivalents       (9,314)       37,451         Cash and cash equivalents – beginning of period       40,739       18,751   | Deposits   |                   | 68            |                    | _         |  |
| Cash paid for acquisitions (net of cash received from acquisitions)       (29,365)       (28,460         Purchase of property and equipment       (1,810)       (1,582         Net cash used in investing activities       (31,175)       (30,042         Cash Flows From Financing Activities:         Proceeds from secondary offering       —       93,469         Borrowings from Senior Credit Facility       10,000       —         Payments on notes payable       (8,483)       (7,410         Payments of contingent consideration       (1,213)       (728         Proceeds from exercise of unit warrant       —       1,092         Payments of borrowings from Senior Credit Facility       —       (36,500         Net cash provided by financing activities       304       49,923         Net (decrease) increase in Cash and Cash Equivalents       (9,314)       37,451         Cash and cash equivalents – beginning of period       40,739       18,751   | Net cash provided by operating activities  |                   | 21,557        |                    | 17,570    |  |
| Cash paid for acquisitions (net of cash received from acquisitions)       (29,365)       (28,460         Purchase of property and equipment       (1,810)       (1,582         Net cash used in investing activities       (31,175)       (30,042         Cash Flows From Financing Activities:         Proceeds from secondary offering       —       93,469         Borrowings from Senior Credit Facility       10,000       —         Payments on notes payable       (8,483)       (7,410         Payments of contingent consideration       (1,213)       (728         Proceeds from exercise of unit warrant       —       1,092         Payments of borrowings from Senior Credit Facility       —       (36,500         Net cash provided by financing activities       304       49,923         Net (decrease) increase in Cash and Cash Equivalents       (9,314)       37,451         Cash and cash equivalents – beginning of period       40,739       18,751   | Cash Flows From Investing Activities:  |                   |               |                    |           |  |
| Purchase of property and equipment         (1,810)         (1,582)           Net cash used in investing activities         (31,175)         (30,042)           Cash Flows From Financing Activities:         Secondary offering         93,469           Proceeds from secondary offering         -         93,469           Borrowings from Senior Credit Facility         10,000         -           Payments on notes payable         (8,483)         (7,410)           Payments of contingent consideration         (1,213)         (728)           Proceeds from exercise of unit warrant         -         1,092           Payments of borrowings from Senior Credit Facility         -         36,500           Net cash provided by financing activities         304         49,923           Net (decrease) increase in Cash and Cash Equivalents         (9,314)         37,451           Cash and cash equivalents – beginning of period         40,739         18,751   |  |                   | (29,365)      |                    | (28,460)  |  |
| Net cash used in investing activities         (31,175)         (30,042)           Cash Flows From Financing Activities:           Proceeds from secondary offering         —         93,469           Borrowings from Senior Credit Facility         10,000         —           Payments on notes payable         (8,483)         (7,410           Payments of contingent consideration         (1,213)         (728           Proceeds from exercise of unit warrant         —         1,092           Payments of borrowings from Senior Credit Facility         —         (36,500)           Net cash provided by financing activities         304         49,923           Net (decrease) increase in Cash and Cash Equivalents         (9,314)         37,451           Cash and cash equivalents – beginning of period         40,739         18,751  |  |                   |               |                    | (1,582)   |  |
| Proceeds from secondary offering — 93,469 Borrowings from Senior Credit Facility 10,000 — Payments on notes payable (8,483) (7,410 Payments of contingent consideration (1,213) (728 Proceeds from exercise of unit warrant — 1,092 Payments of borrowings from Senior Credit Facility — (36,500 Net cash provided by financing activities 304 49,923  Net (decrease) increase in Cash and Cash Equivalents (9,314) 37,451 Cash and cash equivalents — beginning of period 40,739 18,751  | Net cash used in investing activities  |                   |               |                    | (30,042)  |  |
| Proceeds from secondary offering — 93,469 Borrowings from Senior Credit Facility 10,000 — Payments on notes payable (8,483) (7,410 Payments of contingent consideration (1,213) (728 Proceeds from exercise of unit warrant — 1,092 Payments of borrowings from Senior Credit Facility — (36,500 Net cash provided by financing activities 304 49,923  Net (decrease) increase in Cash and Cash Equivalents (9,314) 37,451 Cash and cash equivalents — beginning of period 40,739 18,751  | Cash Flows From Financing Activities:  |                   |               |                    |           |  |
| Borrowings from Senior Credit Facility  Payments on notes payable  Reyments of contingent consideration  Payments of contingent consideration  Proceeds from exercise of unit warrant  Payments of borrowings from Senior Credit Facility  Payments of contingent consideration  (1,213)  (36,500)  Net cash provided by financing activities  304  49,923  Poet (decrease) increase in Cash and Cash Equivalents  (9,314)  37,451  Cash and cash equivalents – beginning of period  40,739  18,751   | _  |                   | <u> </u>      |                    | 93,469    |  |
| Payments on notes payable (8,483) (7,410 Payments of contingent consideration (1,213) (728 Proceeds from exercise of unit warrant — 1,092 Payments of borrowings from Senior Credit Facility — (36,500 Net cash provided by financing activities 304 49,923  Net (decrease) increase in Cash and Cash Equivalents (9,314) 37,451 Cash and cash equivalents – beginning of period 40,739 18,751  |  |                   | 10.000        |                    | _         |  |
| Payments of contingent consideration (1,213) (728) Proceeds from exercise of unit warrant — 1,092 Payments of borrowings from Senior Credit Facility — (36,500) Net cash provided by financing activities 304 49,923  Net (decrease) increase in Cash and Cash Equivalents (9,314) 37,451 Cash and cash equivalents – beginning of period 40,739 18,751   | -  |                   | (8.483)       |                    | (7.410)   |  |
| Proceeds from exercise of unit warrant — 1,092 Payments of borrowings from Senior Credit Facility — (36,500 Net cash provided by financing activities 304 49,923  Net (decrease) increase in Cash and Cash Equivalents (9,314) 37,451 Cash and cash equivalents – beginning of period 40,739 18,751   |  |                   |               |                    | (728)     |  |
| Payments of borrowings from Senior Credit Facility  Net cash provided by financing activities  Set (decrease) increase in Cash and Cash Equivalents Cash and cash equivalents – beginning of period  (36,500  49,923  (9,314)  37,451  18,751   | -  |                   | _             |                    |           |  |
| Net cash provided by financing activities 304 49,923  Net (decrease) increase in Cash and Cash Equivalents (9,314) 37,451  Cash and cash equivalents – beginning of period 40,739 18,751  |  |                   | _             |                    |           |  |
| Net (decrease) increase in Cash and Cash Equivalents  Cash and cash equivalents – beginning of period  (9,314)  37,451  40,739  18,751  |  |                   | 304           |                    |           |  |
| Cash and cash equivalents – beginning of period 40,739 18,751   | The same provided by the same and the same a |                   |               |                    | 12 32 = 0 |  |
| Cash and cash equivalents – beginning of period 40,739 18,751   | Net (decrease) increase in Cash and Cash Equivalents   |                   | (9.314)       |                    | 37,451    |  |
|   | Cash and cash equivalents – beginning of period  |                   |               |                    | 18,751    |  |
|   |  | \$                |               | \$                 |           |  |

## NV5 Global, Inc. and Subsidiaries CONSOLIDATED STATEMENTS OF CASH FLOWS (UNAUDITED) (in thousands)

|   |    | Nine Months Ended |                    |       |  |  |  |  |
|---|----|-------------------|--------------------|-------|--|--|--|--|
|   | Se | eptember 28, 2019 | September 29, 2018 |       |  |  |  |  |
| Non-cash investing and financing activities:                |    |                   |                    |       |  |  |  |  |
| Contingent consideration (earn-out)                         | \$ | 2,570             | \$                 | 1,565 |  |  |  |  |
| Notes payable and other obligations issued for acquisitions | \$ | 10,044            | \$                 | 8,356 |  |  |  |  |
| Stock issuance for acquisitions                             | \$ | 2,632             | \$                 | 4,950 |  |  |  |  |
| Capital leases  | \$ | 769               | \$                 | 2,878 |  |  |  |  |

(in thousands, except share data)

## Note 1 - Organization and Nature of Business Operations

#### Business

NV5 Global, Inc. and its subsidiaries (collectively, the "Company," "NV5 Global," "our," "we") is a provider of professional and technical engineering and consulting solutions to public and private sector clients in the infrastructure, energy, construction, real estate and environmental markets, operating nationwide and abroad. The Company's clients include the U.S. federal, state and local governments, and the private sector. NV5 Global provides a wide range of services, including, but not limited to:

- Infrastructure, engineering and support
- Construction quality assurance, testing and inspection
- Program management
- Energy
- Environmental
- Planning
- Design
- Consulting

- Management oversight
- Permitting
- Inspection and field supervision
- Testing inspection and certification
- Forensic engineering
- Litigation support
- Condition assessment
- Compliance certification

#### Note 2 - Summary of Significant Accounting Policies

Basis of Presentation and Principles of Consolidation

The consolidated financial statements of the Company are presented in U.S. dollars in conformity with accounting principles generally accepted in the United States ("U.S. GAAP") and have been prepared pursuant to the rules and regulations of the Securities and Exchange Commission (the "SEC") for reporting of interim financial information. Pursuant to such rules and regulations, certain information and footnote disclosures normally included in financial statements prepared in accordance with U.S. GAAP have been condensed or omitted. The consolidated financial statements include the accounts of the Company and its subsidiaries. All intercompany transactions and balances have been eliminated in consolidation. Certain amounts in the consolidated financial statements and associated notes may not add due to rounding. All percentages have been calculated using unrounded amounts.

In the opinion of management, the accompanying unaudited interim consolidated financial statements of the Company contain all adjustments necessary to present fairly the financial position and results of operations of the Company as of the dates and for the periods presented. Accordingly, these statements should be read in conjunction with the consolidated financial statements and notes contained in the Company's Annual Report on Form 10-K for the year ended December 29, 2018 (the "2018 Form 10-K"). The results of operations and cash flows for the interim periods presented are not necessarily indicative of the results to be expected for any future interim period or for the full 2019 fiscal year.

## Goodwill and Intangible Assets

Goodwill is the excess of consideration paid for an acquired entity over the amounts assigned to assets acquired, including other identifiable intangible assets and liabilities assumed in a business combination. To determine the amount of goodwill resulting from a business combination, the Company performs an assessment to determine the acquisition date fair value of the acquired company's tangible and identifiable intangible assets and liabilities.

Goodwill is required to be evaluated for impairment on an annual basis or whenever events or changes in circumstances indicate the asset may be impaired. An entity has the option to first assess qualitative factors to determine whether the existence of events or circumstances leads to a determination that it is more likely than not that the fair value of a reporting unit is less than its carrying amount. These qualitative factors include: macroeconomic and industry conditions, cost factors, overall financial performance and other relevant entity-specific events. If the entity determines that this threshold is met, then performing the two-step quantitative impairment test is unnecessary. The two-step impairment test requires a comparison of the carrying value of the assets and liabilities associated with a reporting unit, including goodwill, with the fair value of the reporting unit. The Company

(in thousands, except share data)

determines fair value through multiple valuation techniques, and weights the results accordingly. NV5 Global is required to make certain subjective and complex judgments in assessing whether an event of impairment of goodwill has occurred, including assumptions and estimates used to determine the fair value of its reporting units. If the carrying value of a reporting unit exceeds the fair value of the reporting unit, the Company would calculate the implied fair value of its reporting unit goodwill as compared to the carrying value of its reporting unit goodwill to determine the appropriate impairment charge, if any. The Company has elected to perform its annual goodwill impairment review on August 1 of each year. The Company conducts its annual impairment tests on the goodwill using the quantitative method of evaluating goodwill.

Identifiable intangible assets primarily include customer backlog, customer relationships, trade names and non-compete agreements. Amortizable intangible assets are amortized on a straight-line basis over their estimated useful lives and reviewed for impairment whenever events or changes in circumstances indicate that the assets may be impaired. If an indicator of impairment exists, the Company compares the estimated future cash flows of the asset, on an undiscounted basis, to the carrying value of the asset. If the undiscounted cash flows exceed the carrying value, no impairment is indicated. If the undiscounted cash flows do not exceed the carrying value, then impairment, if any, is measured as the difference between fair value and carrying value, with fair value typically based on a discounted cash flow model.

On August 1, 2019, the Company conducted its annual impairment tests using the quantitative method of evaluating goodwill. Based on the quantitative analyses the Company determined the fair value of each of the reporting units exceeded its carrying value. Therefore, the goodwill was not impaired and the Company did not recognize an impairment charge relating to goodwill as of August 1, 2019. Furthermore, there were no indicators, events or changes in circumstances that would indicate goodwill was impaired during the period from August 2, 2019 through September 28, 2019.

See Note 8 for further information on goodwill and identified intangibles.

There have been no significant changes, other than those related to the adopted new accounting standards below, in the Company's accounting policies from those disclosed in our 2018 Form 10-K.

Adoption of New Accounting Standards

#### Leases

We adopted ASU No. 2016-2, *Leases* ("Topic 842"), as of the first day of the fiscal year 2019 using the modified retrospective approach and elected not to adjust comparative periods. In addition, we elected the package of practical expedients permitted under the transition guidance within the new standard, which permits us not to reassess under the new standard our prior conclusions about lease identification, lease classification and the initial direct costs. We elected the practical expedient to keep leases with an initial term of 12 months or less off the balance sheet and the practical expedient to account for non-lease components in a contract as part of a single lease component. Lease payments are recognized in the Consolidated Statements of Operations on a straight-line basis over the lease term. Adoption of the new standard resulted in the recording of additional right-of-use lease assets and lease liabilities of \$34,186 and \$34,965, respectively, as of the first day of the fiscal year 2019. The standard did not materially impact our consolidated net earnings and had no impact on cash flows. Additionally, there was no cumulative effect of adoption on retained earnings in the Statement of Changes in Stockholders' Equity.

#### **Revenue Recognition**

On the first day of fiscal year 2018, we adopted ASC Topic 606, *Revenue from Contracts with Customers* ("Topic 606"), using the modified retrospective approach to all contracts that were not completed as of the beginning of fiscal year 2018. We utilize the portfolio method practical expedient, which allows companies to account for multiple contracts as a portfolio, instead of accounting for them on a contract by contract basis (commonly known as the contract method). For our time and materials contracts, we apply the as-invoiced practical expedient, which permits us to recognize revenue as the right to invoice for services performed. The new standard did not materially affect our consolidated net income, financial position, or cash flows.

#### Performance Obligations

Some of our contracts have a single performance obligation as the promise to transfer the individual goods or services is not separately identifiable from other promises in the contracts and therefore, is not distinct. However, in some instances, we may also promise to provide distinct goods or services within a contract, resulting in multiple performance obligations. For contracts

(in thousands, except share data)

with multiple performance obligations, we allocate the contract transaction price to each performance obligation using the best estimate of the standalone selling price of each distinct good or service in the contract. Typically, we sell a customer a specific service and use the expected cost plus a margin approach to estimate the standalone selling price of each performance obligation.

The Company's performance obligations are satisfied as work progresses or at a point in time. Revenue on our cost-reimbursable contracts is recognized over time using direct costs incurred or direct costs incurred to date as compared to the estimated total direct costs for performance obligations because it best depicts the transfer of control to the customer. Contract costs include labor, subcontractors' costs and other direct costs.

Gross revenue from services transferred to customers at a point in time is recognized when the customer obtains control of the asset, which is generally upon delivery and acceptance by the customer of the reports and/or analysis performed.

As of September 28, 2019, we had \$504,349 of remaining performance obligations, of which \$390,454 is expected to be recognized over the next 12 months and the majority of the balance over the next 24 months. Contracts for which work authorizations have been received are included in performance obligations. Most of our government contracts are multi-year contracts for which funding is appropriated on an annual basis, therefore performance obligations includes only those amounts that have been funded and authorized and does not reflect the full amounts we may receive over the term of such contracts. In the case of non-government contracts and project awards, performance obligations includes future revenue at contract or customary rates, excluding contract renewals or extensions that are at the discretion of the client. For contracts with a not-to-exceed maximum amount, we include revenue from such contracts in performance obligations to the extent of the remaining estimated amount.

#### Contract Assets and Liabilities

The timing of revenue recognition, billings and cash collections results in billed receivables, unbilled receivables (contract assets), and billings in excess of costs and estimated earnings on uncompleted contracts (contract liabilities) on the Consolidated Balance Sheet. The liability "Billings in excess of costs and estimated earnings on uncompleted contracts" represents billings in excess of revenues recognized on these contracts as of the reporting date. This liability is generally classified as current. Revenue recognized that was included in the contract liability balance at the beginning of the fiscal year was \$2 and \$6,437 for the three and nine months ended September 28, 2019, respectively and \$222 and \$434 for the three and nine months ended September 29, 2018, respectively.

## **Note 3 – Recent Accounting Pronouncements**

In January 2017, the FASB issued ASU 2017-04, *Intangibles-Goodwill and Other* ("Topic 350") simplifying the test for goodwill impairment. This ASU eliminates Step 2 of the goodwill impairment test and simplifies how the amount of an impairment loss is determined. The update is effective for public companies in the beginning of fiscal year 2020 and will be applied on a prospective basis. We will adopt this ASU at the beginning of fiscal year 2020. We do not expect the impact of this ASU to be material to our consolidated financial statements.

## Note 4 - Earnings per Share

Basic earnings per share is calculated by dividing net income by the weighted average number of common shares outstanding during the period. Diluted earnings per share reflects the potential dilution that could occur if securities or other contracts to issue common stock were exercised or converted into common stock or resulted in the issuance of common stock that then shared in the earnings of the Company. The effect of potentially dilutive securities is not considered during periods of loss or if the effect is anti-dilutive.

The weighted average number of shares outstanding in calculating basic earnings per share for the nine months ended September 28, 2019 and September 29, 2018 exclude 625,687 and 588,430 non-vested restricted shares, respectively. There were no potentially anti-dilutive securities during the three and nine months ended September 28, 2019 and September 29, 2018.

The following table represents a reconciliation of the net income and weighted average shares outstanding for the calculation of basic and diluted earnings per share:

(in thousands, except share data)

|   |    | Three Mo            | Ended                 | Nine Months Ended |     |                       |    |                     |
|---|----|---------------------|-----------------------|-------------------|-----|-----------------------|----|---------------------|
|   | Se | ptember 28,<br>2019 | September 29,<br>2018 |                   | Sep | September 28,<br>2019 |    | otember 29,<br>2018 |
| Numerator:  |    |                     |                       |                   |     |                       |    |                     |
| Net income – basic and diluted                            | \$ | 5,843               | \$                    | 7,285             | \$  | 20,180                | \$ | 19,197              |
|   |    |                     |                       |                   |     |                       |    |                     |
| Denominator:  |    |                     |                       |                   |     |                       |    |                     |
| Basic weighted average shares outstanding                 |    | 12,191,405          |                       | 11,256,946        |     | 12,086,588            |    | 10,686,040          |
| Effect of dilutive non-vested restricted shares and units |    | 308,240             |                       | 381,250           |     | 325,219               |    | 396,577             |
| Effect of issuable shares related to acquisitions         |    | 67,321              |                       | 63,198            |     | 73,242                |    | 88,594              |
| Effect of warrants  |    | <u> </u>            |                       | _                 |     | _                     |    | 34,537              |
| Diluted weighted average shares outstanding               |    | 12,566,966          |                       | 11,701,394        |     | 12,485,049            |    | 11,205,748          |

#### Warrant exercise

In conjunction with our initial public offering on March 26, 2013, the underwriter received a warrant to acquire up to 140,000 units ("Unit Warrant"). On March 23, 2016, the underwriter paid us \$1,008 to exercise the Unit Warrant. Each of the units delivered upon exercise consisted of one share of our common stock and one warrant to purchase one share of our common stock at an exercise price of \$7.80 per share ("Warrant"), which warrant expired on March 27, 2018. On March 19, 2018, the underwriter paid us \$1,092 to exercise the Warrant. On March 21, 2018, we delivered 140,000 shares of common stock to the underwriter.

## Note 5 – Business Acquisitions

## 2019 Acquisitions

On July 2, 2019, the Company acquired WHPacific, Inc. ("WHPacific"), a leading provider of design engineering and surveying services serving Washington, Oregon, Idaho, New Mexico, Arizona and California for a cash purchase price of \$9,000. In order to determine the fair values of tangible and intangible assets acquired and liabilities assumed for WHPacific, the Company engaged a third-party independent valuation specialist to assist in the determination of fair values. The Company expects to finalize the purchase price allocation with respect to this transaction by the end of the fourth quarter 2019.

On July 1, 2019, the Company acquired GeoDesign, Inc. ("GeoDesign"), a geotechnical, environmental, geological, mining and pavement engineering company with serving Washington, Oregon and California. The aggregate purchase price is up to \$12,800, including \$8,500 of cash, \$2,000 in promissory note (bearing interest at 4%), payable in four equal installments of \$500 due on the first, second, third and fourth anniversaries of July 1, 2019 and \$375 of the Company's common stock (4,731 shares) issued at the closing date. The purchase price also includes \$425 of the Company's common stock payable on the first and second anniversary of July 1, 2019. Further, the purchase price includes a \$1,500 earn-out of cash, which was recorded at an estimated fair value of \$1,456. In order to determine the fair values of tangible and intangible assets acquired and liabilities assumed for GeoDesign, the Company engaged a third-party independent valuation specialist to assist in the determination of fair values. The Company expects to finalize the purchase price allocation with respect to this transaction by the end of the fourth quarter 2019.

On June 3, 2019, the Company acquired Alta Environmental, L.P. ("Alta"), a consulting firm specializing in air quality, environmental building sciences, water resources, site assessment and remediation as well as environmental health and safety compliance services. The aggregate purchase price is up to \$6,500, including \$4,000 of cash and \$2,000 in promissory note (bearing interest at 4%), payable in four equal installments of \$500 due on the first, second, third and fourth anniversaries of June 3, 2019. Further, the purchase price includes a \$500 earn-out of cash, which was recorded at an estimated fair value of \$485. In order to determine the fair values of tangible and intangible assets acquired and liabilities assumed for Alta, the Company engaged a third-party independent valuation specialist to assist in the determination of fair values. The Company expects to finalize the purchase price allocation with respect to this transaction by the end of the fourth quarter 2019.

(in thousands, except share data)

On June 3, 2019, the Company acquired Page One Consultants ("Page One"), a program management and construction quality assurance firm based in Orlando, Florida. The aggregate purchase price is up to \$3,900, including \$2,000 of cash, \$1,000 in promissory note (bearing interest at 3%), payable in three equal installments of \$333 due on the first, second and third anniversaries of June 3, 2019 and \$200 of the Company's common stock (2,647 shares) issued at the closing date. The purchase price also includes \$200 of the Company's common stock payable on the first anniversary of June 3, 2019. Further, the purchase price includes a \$500 earn-out of cash and stock, which was recorded at an estimated fair value of \$448. In order to determine the fair values of tangible and intangible assets acquired and liabilities assumed for Page One, the Company engaged a third-party independent valuation specialist to assist in the determination of fair values. The Company expects to finalize the purchase price allocation with respect to this transaction by the end of the fourth quarter 2019.

On March 22, 2019, the Company acquired The Sextant Group, Inc. ("The Sextant Group"), a national leading provider of audiovisual, information and communications technology, acoustics consulting, and design services headquartered in Pittsburgh, PA. The Sextant Group provides services throughout the U.S. and is well-known for creating integrated technology solutions for a wide range of public and private sector clients. The aggregate purchase price is up to \$11,000, including \$7,000 of cash and \$4,000 in promissory note (bearing interest at 4%), payable in four equal installments of \$1,000 due on the first, second, third and fourth anniversaries of March 22, 2019. In order to determine the fair values of tangible and intangible assets acquired and liabilities assumed for The Sextant Group, the Company engaged a third-party independent valuation specialist to assist in the determination of fair values. The Company expects to finalize the purchase price allocation with respect to this transaction by the end of the fourth quarter 2019.

On December 31, 2018, the Company acquired certain assets of Celtic Energy, Inc. ("Celtic"), a nationally recognized energy consulting firm that specializes in energy project management and oversight. The aggregate purchase price is up to \$1,900, including \$1,000 in cash, \$300 in promissory note (bearing interest at 3%), payable in three equal installments of \$100 on the first, second and third anniversaries of December 31, 2018 and \$200 of the Company's common stock (3,227 shares) issued at the closing date. The purchase price also includes \$200 of the Company's common stock payable on the first anniversary of December 31, 2018. Further, the purchase price includes a \$200 earn-out of cash, which was recorded at an estimated fair value of \$181. In order to determine the fair values of tangible and intangible assets acquired and liabilities assumed for Celtic, the Company performed a purchase price allocation.

#### 2018 Acquisitions

On November 2, 2018 the Company acquired CHI Engineering, Inc. ("CHI"), an infrastructure engineering firm based in Portsmouth, New Hampshire. CHI is a leading provider of engineering, procurement, and construction management services to the liquefied natural gas ("LNG"), petroleum gas ("LPG") and Natural Gas industries. CHI's client base includes the majority of LNG facility owner/operators in the U.S. The aggregate purchase price of this acquisition is up to \$53,000, including \$30,000 in cash, \$15,000 in promissory notes (bearing interest at 3%), payable in four equal installments of \$3,750 on the first, second, third and fourth anniversaries of November 2, 2018 and \$3,000 of the Company's common stock (36,729 shares) issued at the closing date. In July 2019, the Company received \$2,360 from the sellers of CHI, as a working capital adjustment which was recorded a reduction of the purchase price paid for the acquisition of CHI. The purchase price also includes \$3,000 of the Company's common stock payable in three installments of \$1,000, due on the first, second and third anniversaries of November 2, 2018. The purchase price also includes a \$2,000 earn-out of cash (at a 3% interest rate which begins to accrue on January 1, 2020), which was recorded at its estimated fair value of \$1,547, based on a probability-weighted approach valuation technique used to determine the fair value of the contingent consideration on the acquisition date. The note and the earn-out are due to related party individuals who became employees of the Company upon the acquisition. In order to determine the fair values of tangible and intangible assets acquired and liabilities assumed for CHI, the Company engaged a third-party independent valuation specialist to assist in the determination of fair values.

On August 24, 2018, the Company acquired all of the outstanding equity interests in CALYX Engineers and Consultants, Inc. ("CALYX"), an infrastructure and transportation firm based in Cary, North Carolina. CALYX provides roadway and structure design, transportation planning, water resources, construction services, utility services, building structure design, land development, traffic services, cultural resources, surveying, and environmental services. CALYX serves both public and private clients, including state departments of transportation, municipalities, developers, higher education, and healthcare systems. The acquisition of CALYX will expand our infrastructure engineering service in the southeast United States. The purchase price of this acquisition is \$34,000, subject to customary closing working capital adjustments, including \$25,000 in cash, \$4,000 in promissory notes (bearing interest at 3.75%), payable in four installments of \$1,000, due on the first, second, third and fourth

(in thousands, except share data)

anniversaries of August 24, 2018 (see Note 10), \$3,000 of the Company's common stock (36,379 shares) as of the closing date of the acquisition, and \$2,000 in cash payable within 120 days of the closing date. The note is due to related party individuals who became employees of the Company. In order to determine the fair values of tangible and intangible assets acquired and liabilities assumed for CALYX, the Company engaged a third-party independent valuation specialist to assist in the determination of fair values.

On February 2, 2018, the Company acquired CSA (M&E) Ltd. ("CSA"), a leading provider of Mechanical, Electrical, and Plumbing (MEP) engineering and sustainability consulting services. CSA provides MEP and sustainability services for the retail, education, healthcare, industrial, corporate, hospitality and infrastructure market sectors with offices in Hong Kong, Macau and the UAE. CSA serves private and public sector clients throughout Asia and the Middle East. The purchase price of this acquisition was up to \$4,200, including \$2,000 in cash; \$600 in promissory notes (bearing interest at 3%), payable in four installments of \$150, due on the first, second, third and fourth anniversaries of February 2, 2018, the effective date of the acquisition; and \$150 of the Company's common stock (2,993 shares) issued as of the closing date. The purchase price also includes \$250 of the Company's common stock payable in two installments of \$125, due on the first and second anniversaries of the acquisition. The purchase price also included a non-interest bearing earn-out of up to \$1,200 payable in cash and stock, subject to the achievement of certain agreed upon financial metrics for fiscal year 2018. The earn-out of \$1,200 is non-interest bearing and was recorded at its estimated fair value of \$899, based on a probability-weighted approach valuation technique used to determine the fair value of the contingent consideration on the acquisition date. The note and the earn-out are due to a related party individual who became an employee of the Company upon the acquisition. In order to determine the fair values of tangible and intangible assets acquired and liabilities assumed for CSA, the Company engaged a third-party independent valuation specialist to assist in the determination of fair values.

On January 12, 2018, the Company acquired all of the outstanding equity interest in Butsko Utility Design, Inc. ("Butsko"). Butsko is leading provider of utility planning and design services serving both public and private sector clients through its offices in Southern California and Washington. The purchase price of this acquisition was up to \$4,250, including \$1,500 in cash; \$1,000 in promissory notes (bearing interest at 3%), payable in four installments of \$250, due on the first, second, third and fourth anniversaries of January 12, 2018, the effective date of the acquisition; and \$300 of the Company's common stock (5,630 shares) issued as of the closing date. The purchase price also includes \$600 of the Company's common stock payable in two installments of \$300, due on the first and second anniversaries of the acquisition. The purchase price also included a non-interest bearing earn-out of up to \$850 payable in cash and stock, subject to the achievement of certain agreed upon financial metrics for fiscal year 2018. The earn-out of \$850 is non-interest bearing and was recorded at its estimated fair value of \$666, based on a probability-weighted approach valuation technique used to determine the fair value of the contingent consideration on the acquisition date. The note and the earn-out are due to a related party individual who became an employee of the Company upon the acquisition. In order to determine the fair values of tangible and intangible assets acquired and liabilities assumed for Butsko, the Company engaged a third-party independent valuation specialist to assist in the determination of fair values.

The following table summarizes the fair values of the assets acquired and liabilities assumed as of the acquisition date for the acquisitions closed during 2019 and 2018:

(in thousands, except share data)

|  | 2019         | 2018         |
|--|--------------|--------------|
| Cash   | \$<br>75     | \$<br>345    |
| Billed and unbilled receivables, net   | 19,417       | 20,999       |
| Property and equipment   | 1,580        | 3,122        |
| Prepaid expenses   | 2,056        | 589          |
| Other assets   | 164          | 83           |
| Intangible assets:   |              |              |
| Customer relationships   | 12,839       | 32,267       |
| Trade name   | 910          | 2,479        |
| Customer backlog   | 852          | 8,007        |
| Non-compete  | <br>1,647    | 4,306        |
| Total Assets   | 39,540       | 72,197       |
| Liabilities  | (9,263)      | (11,589)     |
| Deferred tax liabilities   | <br>(4,495)  | (8,903)      |
| Net assets acquired  | \$<br>25,782 | \$<br>51,705 |
|  |              |              |
| Consideration paid (Cash, Notes and/or stock)  | \$<br>42,898 | \$<br>90,516 |
| Contingent earn-out liability (Cash and stock)                                       | 2,570        | 3,112        |
| Total Consideration  | \$<br>45,468 | \$<br>93,628 |
| Excess consideration over the amounts assigned to the net assets acquired (Goodwill) | \$<br>19,686 | \$<br>41,923 |

Goodwill was recorded based on the amount by which the purchase price exceeded the fair value of the net assets acquired and the amount is attributable to the reputation of the business acquired, the workforce in place and the synergies to be achieved from these acquisitions. See Note 8 for further information on goodwill and identified intangibles.

The consolidated financial statements of the Company for the three and nine months ended September 28, 2019 and September 29, 2018 include the results of operations from any business acquired from their respective dates of acquisition during each of the respective period as follows:

|                            |       | Three Mo      | Ended | Nine Months Ended |       |                |      |                |  |
|----------------------------|-------|---------------|-------|-------------------|-------|----------------|------|----------------|--|
|                            | Septe | mber 28, 2019 | Sept  | tember 29, 2018   | Septe | ember 28, 2019 | Sept | ember 29, 2018 |  |
| Gross Revenues             | \$    | 16,537        | \$    | 4,984             | \$    | 22,229         | \$   | 11,374         |  |
| Income before income taxes | \$    | 1,272         | \$    | 1,356             | \$    | 2,211          | \$   | 2,008          |  |

The following table presents the unaudited, pro forma consolidated results of operations (in thousands, except per share amounts) for the three and nine months ended September 28, 2019 and September 29, 2018 as if the acquisitions of CHI, CALYX, The Sextant Group, Page One, Alta, WHPacific and GeoDesign had occurred as of January 1, 2018. The pro forma information provided below is compiled from the pre-acquisition financial information of CHI, CALYX, The Sextant Group, Page One, Alta, WHPacific and GeoDesign which includes pro forma adjustments for amortization expense, adjustments to certain expenses, and the income tax impact of these adjustments. The pro forma results are not necessarily indicative of (i) the results of operations that would have occurred had the operations of these acquisitions actually been acquired on January 1, 2018 or (ii) future results of operations:

(in thousands, except share data)

|                            |                | Three Mo          | Ended | <b>Nine Months Ended</b> |     |                  |     |                  |
|----------------------------|----------------|-------------------|-------|--------------------------|-----|------------------|-----|------------------|
|                            | $\overline{s}$ | eptember 28, 2019 | Se    | ptember 29, 2018         | Sej | ptember 28, 2019 | Sep | ptember 29, 2018 |
| Gross revenues             | \$             | 131,032           | \$    | 141,249                  | \$  | 403,162          | \$  | 417,672          |
| Net income                 | \$             | 5,843             | \$    | 9,526                    | \$  | 19,407           | \$  | 24,550           |
| Basic earnings per share   | \$             | 0.48              | \$    | 0.84                     | \$  | 1.61             | \$  | 2.28             |
| Diluted earnings per share | \$             | 0.46              | \$    | 0.81                     | \$  | 1.55             | \$  | 2.18             |

The Company has determined the supplemental disclosures pursuant to ASC 805-10-50-2h, for the Celtic, CSA and Butsko acquisitions were not material to the Company's consolidated financial statements both individually and in the aggregate.

## Note 6 - Billed and Unbilled Receivables

Billed and Unbilled Receivables consists of the following:

|                                       | Septe   | ember 28, 2019 | Decembe | er 29, 2018 |
|---------------------------------------|---------|----------------|---------|-------------|
| Billed receivables                    | \$      | 114,148        | \$      | 101,482     |
| Less: allowance for doubtful accounts |         | (4,557)        |         | (3,158)     |
| Billed receivables, net               | \$      | 109,590        | \$      | 98,324      |
|                                       | <u></u> |                |         |             |
| Unbilled receivables                  | \$      | 55,206         | \$      | 44,799      |
| Less: allowance for doubtful accounts |         | (1,388)        |         | (1,388)     |
| Unbilled receivables, net             | \$      | 53,818         | \$      | 43,411      |

## Note 7 - Property and Equipment, net

Property and equipment, net, consists of the following:

|                                | Septer | December 29, 2018 |    |          |  |
|--------------------------------|--------|-------------------|----|----------|--|
| Office furniture and equipment | \$     | 2,877             | \$ | 2,328    |  |
| Computer equipment             |        | 12,382            |    | 11,640   |  |
| Survey and field equipment     |        | 6,388             |    | 5,526    |  |
| Leasehold improvements         |        | 3,981             |    | 2,541    |  |
|                                |        | 25,628            |    | 22,035   |  |
| Accumulated depreciation       |        | (13,279)          |    | (10,358) |  |
|                                | \$     | 12,349            | \$ | 11,677   |  |

Depreciation expense was \$1,317 and \$3,591 for the three and nine months ended September 28, 2019, respectively and \$1,077 and \$3,111 for the three and nine months ended September 29, 2018, respectively.

## Note 8 - Goodwill and Intangible Assets

Goodwill

The changes in the carrying value by reportable segment for the nine months ended September 28, 2019 were as follows:

(in thousands, except share data)

## **Nine Months Ended**

|       | December | 2019    | Acquisitions | Dispo  | sed/Adjustments | September 28, 2019 |    |         |  |
|-------|----------|---------|--------------|--------|-----------------|--------------------|----|---------|--|
| INF   | \$       | 69,255  | \$           | 13,797 | \$              | (2,193)            | \$ | 80,859  |  |
| BTS   |          | 71,675  |              | 5,889  |                 | _                  |    | 77,564  |  |
| Total | \$       | 140,930 | \$           | 19,686 | \$              | (2,193)            | \$ | 158,423 |  |

Goodwill of approximately \$5,712 and \$13,549 from acquisitions during the nine months ended September 28, 2019 and September 29, 2018, respectively, is expected to be deductible for income tax purposes. In July 2019, the Company received \$2,360 from the sellers of CHI, as a working capital adjustment which was recorded a reduction of goodwill and the purchase price paid for the acquisition of CHI.

Intangible Assets

Intangible assets, net, as of September 28, 2019 and December 29, 2018 consist of the following:

|                            |                             | ember 28, 2019 |                           | December 29, 2018 |               |    |                             |    |                             |    |               |
|----------------------------|-----------------------------|----------------|---------------------------|-------------------|---------------|----|-----------------------------|----|-----------------------------|----|---------------|
|                            | Gross<br>Carrying<br>Amount |                | ccumulated<br>mortization |                   | Net<br>Amount |    | Gross<br>Carrying<br>Amount |    | Accumulated<br>Amortization |    | Net<br>Amount |
| Customer relationships (1) | \$<br>113,794               | \$             | (26,434)                  | \$                | 87,360        | \$ | 100,956                     | \$ | (18,724)                    | \$ | 82,232        |
| Trade name (2)             | 9,798                       |                | (8,084)                   |                   | 1,714         |    | 8,888                       |    | (6,469)                     |    | 2,419         |
| Customer backlog (1)       | 16,853                      |                | (10,966)                  |                   | 5,887         |    | 16,000                      |    | (6,730)                     |    | 9,270         |
| Favorable lease (3)        | 553                         |                | (233)                     |                   | 320           |    | 552                         |    | (197)                       |    | 355           |
| Non-compete (4)            | 10,201                      |                | (4,794)                   |                   | 5,407         |    | 8,554                       |    | (3,074)                     |    | 5,480         |
| Total                      | \$<br>151,199               | \$             | (50,511)                  | \$                | 100,688       | \$ | 134,950                     | \$ | (35,194)                    | \$ | 99,756        |

- (1) Amortized on a straight-line basis over estimated lives (1 to 10 years)
- (2) Amortized on a straight-line basis over their estimated lives (1 to 3 years)
- (3) Amortized on a straight-line basis over the remaining lease term of 9 years
- (4) Amortized on a straight-line basis over their contractual lives (4 to 5 years)

Amortization expense was \$5,234 and \$15,317 for the three and nine months ended September 28, 2019, respectively and \$2,980 and \$8,549 for the three and nine months ended September 29, 2018, respectively.

## Note 9 - Accrued Liabilities

Accrued liabilities consist of the following:

|                                | September 28, 2019 | December 29, 2018 |
|--------------------------------|--------------------|-------------------|
| Accrued lease liability        | \$<br>10,768       | \$<br>_           |
| Accrued vacation               | 9,958              | 7,994             |
| Payroll and related taxes      | 5,383              | 8,136             |
| Benefits                       | 1,708              | 1,598             |
| Unrecognized tax benefits      | 878                | 548               |
| Professional liability reserve | 1,168              | 157               |
| Deferred rent                  | _                  | 779               |
| Other                          | 3,791              | 1,641             |
| Total                          | \$<br>33,654       | \$<br>20,853      |

## Note 10 - Notes Payable and Other Obligations

(in thousands, except share data)

Notes payable and other obligations consists of the following:

|   | September | December 29, 2018 |             |
|---|-----------|-------------------|-------------|
| Other obligations   | \$        | 3,774             | \$<br>4,893 |
| Uncollateralized promissory notes                         |           | 41,765            | 40,001      |
| Senior credit facility                                    |           | 10,000            | _           |
| Capital leases  |           | 2,676             | 2,092       |
| Total notes payable and other obligations                 |           | 58,215            | 46,986      |
| Current portion of notes payable and other obligations    |           | (17,578)          | (17,139)    |
| Notes payable and other obligations, less current portion |           | 40,638            | <br>29,847  |

As of September 28, 2019 and December 29, 2018, the carrying amount of debt obligations approximates their fair values based on Level 2 inputs as the terms are comparable to terms currently offered by local lending institutions for arrangements with similar terms to industry peers with comparable credit characteristics.

## Senior Credit Facility

On December 20, 2018, we entered into an amendment to a Credit Agreement (the "Credit Agreement") dated December 7, 2016 with Bank of America, N.A. ("Bank of America") and Merrill Lynch, Pierce, Fenner & Smith Incorporated ("MLPFS"). Pursuant to the amended Credit Agreement, Bank of America agreed to be the sole administrative agent for a five-year \$125,000 Senior Secured Revolving Credit Facility ("Senior Credit Facility") to the Company and, together with PNC Bank, National Association and Regions Bank as the other lenders under the Senior Credit Facility, has committed to lend to the Company all of the Senior Credit Facility, subject to certain terms and conditions. The Senior Credit Facility is secured by a first priority lien on substantially all of the assets of the Company. MLPFS has undertaken to act as sole lead arranger and sole book manager for the Senior Credit Facility. In addition, the Senior Credit Facility includes an accordion feature permitting the Company to request an increase in the Senior Credit Facility by an additional amount of up to \$100,000. The Senior Credit Facility includes a \$20,000 sublimit for the issuance of standby letters of credit and a \$15,000 sublimit for swingline loans. The proceeds of the Senior Credit Facility are intended to be used (i) to finance permitted acquisitions, (ii) for capital expenditures, and (iii) for general corporate purposes.

Borrowings under the Credit Agreement are at variable rates which are, at our option, tied to a Eurocurrency rate equal to LIBOR (London Interbank Offered Rate) plus an applicable rate or a base rate denominated in U.S. dollars. Interest rates are subject to change based on our Consolidated Senior Leverage Ratio (as defined in the Credit Agreement).

The Senior Credit Facility contains certain financial covenants, including a maximum leverage ratio of 4.0:1 and a minimum fixed charge coverage ratio of 1.20:1. Furthermore, the Senior Credit Facility also contains financial reporting covenant provisions and other covenants, representations, warranties, indemnities, and events of default that are customary for facilities of this type. As of September 28, 2019 and December 29, 2018, the Company is in compliance with the financial covenants. As of September 28, 2019 there was \$10,000 outstanding on the Senior Credit Facility. As of December 29, 2018, we had no outstanding balance on the Senior Credit Facility.

## Other Obligations

On July 1, 2019, the Company acquired GeoDesign. The purchase price allowed for the payment of \$425 in shares of the Company's stock or a combination of cash and shares of the Company's stock, at our discretion, payable on the first and second anniversary of July 1, 2019. At September 28, 2019, the outstanding balance of this obligation was \$382.

On June 3, 2019, the Company acquired Page One. The purchase price allowed for the payment of \$200 in shares of the Company's stock or a combination of cash and shares of the Company's stock, at our discretion, payable on the first anniversary of June 3, 2019. At September 28, 2019, the outstanding balance of this obligation was \$181.

(in thousands, except share data)

On December 31, 2018, the Company acquired certain assets of Celtic. The purchase price allowed for the payment of \$200 in shares of the Company's stock or a combination of cash and shares of the Company's stock, at our discretion, payable on the first anniversary of December 31, 2018. At September 28, 2019 the outstanding balance of this obligation was \$181.

On November 2, 2018, the Company acquired CHI. The purchase price allowed for the payment of \$3,000 in shares of the Company's stock or a combination of cash and shares of the Company's stock, at our discretion, payable in three equal annual installments. At September 28, 2019 and December 29, 2018, the outstanding balance of this obligation was \$2,631.

On February 2, 2018, the Company acquired CSA. The purchase price allowed for the payment of \$250 in shares of the Company's stock or a combination of cash and shares of the Company's stock, at our discretion, payable in two equal annual installments. At September 28, 2019 and December 29, 2018, the outstanding balance of this obligation was \$111 and \$222, respectively.

On January 12, 2018, the Company acquired all of the outstanding equity interest in Butsko. The purchase price allowed for the payment of \$600 in shares of the Company's stock or a combination of cash and shares of the Company's stock, at our discretion, payable in two equal annual installments. At September 28, 2019 and December 29, 2018, the outstanding balance of this obligation was \$267 and \$534, respectively.

On September 6, 2017, the Company acquired all of the outstanding equity interest in Marron. The purchase price allowed for the payment of \$133 in shares of the Company's stock or a combination of cash and shares of the Company's stock, at our discretion, payable in two equal annual installments. There was no outstanding balance on this obligation as of September 28, 2019. At December 29, 2018, the outstanding balance of this obligation was \$55.

On June 6, 2017, the Company acquired all of the outstanding equity interest in RDK. The purchase price allowed for the payment of \$1,333 in shares of the Company's stock or a combination of cash and shares of the Company's stock, at our discretion, payable in two equal annual installments. There was no outstanding balance on this obligation as of September 28, 2019. At December 29, 2018, the outstanding balance of this obligation \$504.

On May 20, 2016, the Company acquired all of the outstanding equity interests of Dade Moeller. The purchase price allowed for the payment of \$3,000 in shares of the Company's stock or a combination of cash and shares of the Company's stock, at our discretion, payable in three equal annual installments of \$1,000. There was no outstanding balance on this obligation as of September 28, 2019. At December 29, 2018, the outstanding balance of this obligation was \$936.

#### Uncollateralized Promissory Notes

On July 1, 2019, the Company acquired GeoDesign. The purchase price included an uncollateralized \$2,000 promissory note bearing interest at 4.0% ("GeoDesign Note") and payable in four equal annual installments. The outstanding balance of the GeoDesign Note was \$2,000 as of September 28, 2019.

On June 3, 2019, the Company acquired Alta. The purchase price included an uncollateralized \$2,000 promissory note bearing interest at 4.0% ("Alta Note") and payable in four equal annual installments. The outstanding balance of the Alta Note was \$2,000 as of September 28, 2019.

On June 3, 2019, the Company acquired Page One. The purchase price included an uncollateralized \$1,000 promissory note bearing interest at 3.0% ("Page One Note") and payable in three equal annual installments. The outstanding balance of the Page One Note was \$1,000 as of September 28, 2019.

On March 22, 2019, the Company acquired The Sextant Group. The purchase price included an uncollateralized \$4,000 promissory note bearing interest at 4.0% ("The Sextant Group Note") and payable in four equal annual installments. The outstanding balance of The Sextant Group Note was \$4,000 as of September 28, 2019.

On December 31, 2018, the Company acquired certain assets of Celtic. The purchase price included an uncollateralized \$300 promissory note bearing interest at 3.0% (the "Celtic Note") payable in three equal annual installments. The outstanding balance of the Celtic note was \$300 as of September 28, 2019.

(in thousands, except share data)

On November 2, 2018, the Company acquired CHI. The purchase price included an uncollateralized \$15,000 promissory note bearing interest at 3.0% (the "CHI Note") payable in four equal annual installments. The outstanding balance of the CHI Note was \$15,000 as of September 28, 2019 and December 29, 2018.

On August 24, 2018, the Company acquired CALYX. The purchase price included an uncollateralized \$4,000 promissory note bearing interest at 3.75% (the "CALYX Note") payable in four equal annual installments. The outstanding balance of the CALYX Note was \$3,000 and \$4,000 as of September 28, 2019 and December 29, 2018, respectively.

On February 2, 2018, the Company acquired CSA. The purchase price included an uncollateralized \$600 promissory note bearing interest at 3.0% (the "CSA Note") payable in four equal annual installments. The outstanding balance of the CSA Note was \$450 and \$600 as of September 28, 2019 and December 29, 2018, respectively.

On January 12, 2018, the Company acquired all of the outstanding equity interest in Butsko. The purchase price included an uncollateralized \$1,000 promissory note bearing interest at 3.0% (the "Butsko Note") payable in four equal annual installments. The outstanding balance of the Butsko Note was \$750 and \$1,000 as of September 28, 2019 and December 29, 2018, respectively.

On September 6, 2017, the Company acquired all of the outstanding interests in Marron. The purchase price included an uncollateralized \$300 promissory note bearing interest at 3.0% (the "Marron Note") payable in three equal annual installments. The outstanding balance of the Marron Note was \$100 and \$200 as of September 28, 2019 and December 29, 2018, respectively.

On June 6, 2017, the Company acquired all of the outstanding equity interest in RDK. The purchase price included an uncollateralized \$5,500 promissory note bearing interest at 3.0% (the "RDK Note") payable in four equal annual installments. The outstanding balance of the RDK Note was \$2,750 and \$4,125 as of September 28, 2019 and December 29, 2018, respectively.

On May 4, 2017, the Company acquired all of the outstanding equity interest in H&K. The purchase price included an uncollateralized \$600 promissory note bearing interest at 3.0% (the "H&K Note") payable in four equal annual installments. The outstanding balance of the H&K Note was \$300 and \$450 as of September 28, 2019 and December 29, 2018, respectively.

On May 1, 2017, the Company acquired all of the outstanding equity interest in Lochrane. The purchase price included an uncollateralized \$1,650 promissory note bearing interest at 3.0% (the "Lochrane Note") payable in four equal annual installments. The outstanding balance of the Lochrane Note was \$825 and \$1,238 as of September 28, 2019 and December 29, 2018, respectively.

On December 6, 2016, the Company acquired all of the outstanding interests of CivilSource. The purchase price included an uncollateralized \$3,500 promissory note bearing interest at 3.0% (the "CivilSource Note") payable in four equal annual installments. The outstanding balance of the CivilSource Note was \$1,606 and \$2,625 as of September 28, 2019 and December 29, 2018, respectively.

On November 30, 2016, the Company acquired all of the outstanding interests of Hanna. The purchase price included an uncollateralized \$2,700 promissory note bearing interest at 3.0% (the "Hanna Note") payable in four equal annual installments. The outstanding balance of the Hanna Note was \$1,350 as of September 28, 2019 and December 29, 2018.

On October 26, 2016, the Company acquired all of the outstanding interests of JBA. The purchase price included an uncollateralized \$7,000 promissory note bearing interest at 3.0% (the "JBA Note") payable in five equal annual installments. The outstanding balance of the JBA Note was \$4,200 as of September 28, 2019 and December 29, 2018.

On September 12, 2016, the Company acquired certain assets of Weir. The purchase price included an uncollateralized \$500 promissory note bearing interest at 3.0% (the "Weir Note") payable in four equal annual installments. The outstanding balance of the Weir Note was \$125 and \$250 as of September 28, 2019 and December 29, 2018, respectively.

On May 20, 2016, the Company acquired all of the outstanding equity interests of Dade Moeller. The purchase price included an aggregate of \$6,000 of uncollateralized promissory notes bearing interest at 3.0% (the "Dade Moeller Notes") payable in four equal annual installments. The outstanding balance of the Dade Moeller Notes was \$1,497 and \$3,036 as of September 28, 2019 and December 29, 2018, respectively.

(in thousands, except share data)

On July 1, 2015, the Company acquired all of the outstanding equity interests of RBA. The purchase price included an uncollateralized \$4,000 promissory notes bearing interest at 3.0% (the "RBA Note") payable in four equal annual installments. There was no outstanding balance on the RBA Note as of September 28, 2019. The outstanding balance of the RBA Note was \$1,000 as of December 29, 2018.

On January 30, 2015, the Company acquired all of the outstanding equity interests of JLA. The purchase price included an uncollateralized \$1,250 promissory note bearing interest at 3.5% (the "JLA Note") payable in four equal annual installments. There was no outstanding balance on the JLA Note as of September 28, 2019. As of December 29, 2018, the outstanding balance of the JLA note was \$313.

#### **Note 11 – Contingent Consideration**

The following table summarizes the changes in the carrying value of estimated contingent consideration:

|   | <b>September 28, 2019</b> | December 29, 2018 |
|---|---------------------------|-------------------|
| Contingent consideration, beginning of the year               | \$ 4,698                  | \$ 1,890          |
| Additions for acquisitions                                    | 2,737                     | 3,112             |
| Reduction of liability for payments made                      | (1,938)                   | (728)             |
| Increase of liability related to re-measurement of fair value | 49                        | 424               |
| Total contingent consideration, end of the period             | 5,546                     | 4,698             |
| Current portion of contingent consideration                   | (3,351)                   | (1,845)           |
| Contingent consideration, less current portion                | \$ 2,195                  | \$ 2,853          |

#### Note 12 - Commitments and Contingencies

Litigation, Claims and Assessments

We are subject to certain claims and lawsuits typically filed against the engineering, consulting and construction profession, alleging primarily professional errors or omissions. The Company carries professional liability insurance, subject to certain deductibles and policy limits, against such claims. However, in some actions, parties are seeking damages that exceed our insurance coverage or for which we are not insured. While management does not believe that the resolution of these claims will have a material adverse effect, individually or in aggregate, on its financial position, results of operations or cash flows, management acknowledges the uncertainty surrounding the ultimate resolution of these matters.

## Note 13 - Stock-Based Compensation

In October 2011, our stockholders approved the 2011 Equity Incentive Plan, which was subsequently amended and restated in March 2013 (as amended, the "2011 Equity Plan"). The 2011 Equity Plan provides directors, executive officers, and other employees of the Company with additional incentives by allowing them to acquire ownership interest in the business and, as a result, encouraging them to contribute to the Company's success. We may provide these incentives through the grant of stock options, stock appreciation rights, restricted stock, restricted stock units, performance shares and units, and other cash-based or stock-based awards. As of September 28, 2019, 1,256,403 shares of common stock are authorized and reserved for issuance under the 2011 Equity Plan. This reserve automatically increases on each January 1 from 2014 through 2023, by an amount equal to the smaller of (i) 3.5% of the number of shares issued and outstanding on the immediately preceding December 31, or (ii) an amount determined by our Board of Directors. The restricted shares of common stock granted generally provide for service-based vesting after two to four years following the grant date.

(in thousands, except share data)

The following summarizes the activity of restricted stock awards during the nine months ended September 28, 2019:

|                    | Number of      |          |         |
|--------------------|----------------|----------|---------|
|                    | Unvested       |          |         |
|                    | Restricted     |          |         |
|                    | Shares of      | Weigh    | ited    |
|                    | Common Stock   | Avera    | age     |
|                    | and Restricted | Grant Da | te Fair |
|                    | Stock Units    | Valu     | ıe      |
| December 29, 2018  | 626,911        | \$       | 39.81   |
| Granted            | 248,694        | \$       | 72.98   |
| Vested             | (204,655)      | \$       | 20.24   |
| Forfeited          | (35,263)       | \$       | 51.84   |
| September 28, 2019 | 635,687        | \$       | 58.42   |

Share-based compensation expense relating to restricted stock awards during the three and nine months ended September 28, 2019 was \$2,822 and \$6,989, respectively and \$1,902 and \$4,541, respectively, for the three and nine months ended September 29, 2018. Approximately \$19,867 of deferred compensation, which is expected to be recognized over the remaining weighted average vesting period of 2.33 years, is unrecognized at September 28, 2019. The total fair value of restricted shares vested during the nine months ended September 28, 2019 and September 29, 2018 was \$14,514 and \$7,422, respectively.

#### Note 14 - Income Taxes

As of September 28, 2019 and December 29, 2018, we had net deferred income tax liabilities of \$16,881 and \$16,224, respectively. No valuation allowance against our deferred income tax assets is needed as of September 28, 2019 and December 29, 2018 as it is more-likely-than-not that the positions will be realized upon settlement. Deferred income tax liabilities primarily relate to intangible assets and accounting basis adjustments where we have a future obligation for tax purposes.

Our consolidated effective income tax rate was 22.9% and 24.1%, respectively, for the three and nine months ended September 28, 2019 and 24.5% and 24.8%, respectively, for the three and nine months ended September 29, 2018. Our tax provision includes an income tax benefit related to the vesting of restricted stock totaling \$121 and \$2,593, respectively, for the three and nine months ended September 28, 2019 and \$95 and \$1,210, respectively, for the three and nine months ended September 29, 2018.

We evaluate tax positions for recognition using a more-likely-than-not recognition threshold, and those tax positions eligible for recognition are measured as the largest amount of tax benefit that is greater than 50% likely of being realized upon the effective settlement with a taxing authority that has full knowledge of all relevant information. The California Franchise Tax Board ("CFTB") challenged research and development tax credits generated for the years 2012 to 2014. Fiscal years 2012 through 2018 are considered open tax years in the State of California and 2015 through 2018 in the U.S. federal jurisdiction and other state jurisdictions. The evaluation by the CFTB is ongoing and at September 28, 2019 and December 29, 2018, we had \$878 and \$548, respectively, of unrecognized tax benefits, which if recognized would affect our effective tax rate. It is not expected that there will be a significant change in the unrecognized tax benefits in the next 12 months.

#### Note 15 – Reportable Segments

We report segment information in accordance with ASC Topic No. 280 "Segment Reporting" ("Topic No. 280"). Our Chief Executive Officer is the chief operating decision maker and organized the Company into two operating and reportable segments: Infrastructure (INF), which includes our engineering, civil program management, and construction quality assurance practices; and Building, Technology & Sciences (BTS), which includes our energy, environmental practices and buildings program management practices.

We evaluate the performance of these reportable segments based on their respective operating income before the effect of amortization expense related to acquisitions and other unallocated corporate expenses. We account for inter-segment revenues and transfers as if the sales and transfers were to third parties. All intercompany balances and transactions are eliminated in consolidation.

(in thousands, except share data)

The following tables set forth summarized financial information concerning our reportable segments:

|  |                       | Three Mon | nths l | Ended           | Nine Months Ended |                |     |                 |  |  |  |
|--|-----------------------|-----------|--------|-----------------|-------------------|----------------|-----|-----------------|--|--|--|
|  | September 28, 2019 Se |           | Sep    | tember 29, 2018 | Septe             | ember 28, 2019 | Sep | tember 29, 2018 |  |  |  |
| Gross revenues                         |                       |           |        | _               |                   | _              |     |                 |  |  |  |
| INF                                    | \$                    | 89,035    | \$     | 64,053          | \$                | 250,354        | \$  | 180,611         |  |  |  |
| BTS                                    |                       | 43,060    |        | 41,441          |                   | 128,706        |     | 125,344         |  |  |  |
| Elimination of inter- segment revenues |                       | (1,063)   |        | (1,309)         |                   | (2,720)        |     | (3,218)         |  |  |  |
| Total gross revenues                   | \$                    | 131,032   | \$     | 104,185         | \$                | 376,340        | \$  | 302,737         |  |  |  |
|  |                       |           |        |                 |                   |                |     |                 |  |  |  |
|  |                       |           |        |                 |                   |                |     |                 |  |  |  |
| Segment income before taxes            |                       |           |        |                 |                   |                |     |                 |  |  |  |
| INF                                    | \$                    | 14,008    | \$     | 11,108          | \$                | 41,273         | \$  | 28,951          |  |  |  |
| BTS                                    |                       | 6,871     |        | 7,236           |                   | 20,287         |     | 21,069          |  |  |  |
| Total Segment income before taxes      |                       | 20,879    |        | 18,344          |                   | 61,560         |     | 50,020          |  |  |  |
| Corporate (1)                          |                       | (13,476)  |        | (8,821)         |                   | (37,958)       |     | (26,107)        |  |  |  |
| Total income before taxes              | \$                    | 7,403     | \$     | 9,523           | \$                | 23,602         | \$  | 23,913          |  |  |  |
|  |                       |           |        |                 |                   |                |     |                 |  |  |  |

<sup>(1)</sup> Includes amortization of intangibles of \$5,234 and 15,317 for the three and nine months ended September 28, 2019, respectively and \$2,980 and \$8,549 for the three and nine months ended September 29, 2018, respectively.

Upon adoption of Topic 606, we disaggregate our gross revenues from contracts with customers by geographic location, customer-type and contract-type for each of our reportable segments. Disaggregated revenues include the elimination of inter-segment revenues which has been allocated to each segment. We believe this best depicts how the nature, amount, timing and uncertainty of our revenues and cash flows are affected by economic factors.

| Three Months Ended September 28, 2019 |  |              |                             |  | Nine Months Ended September 28, 2019 |                  |                         |  |                               |                               |  |
|---------------------------------------|--|--------------|-----------------------------|--|--------------------------------------|------------------|-------------------------|--|-------------------------------|-------------------------------|--|
| <br>INF                               |  | BTS          |                             | Total  |                                      | INF              |                         | BTS  |                               | Total                         |  |
|                                       |  |              |                             |  |                                      |                  |                         |  |                               |                               |  |
| \$<br>87,572                          | \$   | 41,872       | \$                          | 129,444  | \$                                   | 247,634          | \$                      | 121,829  | \$                            | 369,463                       |  |
|                                       |  | 1,588        |                             | 1,588  |                                      | _                |                         | 6,877  |                               | 6,877                         |  |
| \$<br>87,572                          | \$   | 43,460       | \$                          | 131,032  | \$                                   | 247,634          | \$                      | 128,706  | \$                            | 376,340                       |  |
| Three Months Ended September 29, 201  |  |              | 9, 2018                     | Nine Months Ended Septemb  |                                      |                  |                         |  | mber 29, 2018                 |                               |  |
| INF                                   |  | BTS          |                             | Total  |                                      | INF              |                         | BTS  |                               | Total                         |  |
|                                       |  |              |                             |  |                                      |                  |                         |  |                               |                               |  |
| \$<br>63,514                          | \$   | 38,678       | \$                          | 102,192  | \$                                   | 178,531          | \$                      | 115,371  | \$                            | 293,902                       |  |
|                                       |  | 1,993        |                             | 1,993  |                                      |                  |                         | 8,835  |                               | 8,835                         |  |
| \$<br>63,514                          | \$   | 40,671       | \$                          | 104 185  | \$                                   | 178 531          | \$                      | 124 206  | S                             | 302,737                       |  |
| \$                                    | \$ 87,572  \$ 87,572  Three Mor INF  \$ 63,514 | \$ 87,572 \$ | Section 2016   Section 2016 | Section   Sect | INF   BTS   Total                    | NF   BTS   Total | INF   BTS   Total   INF | INF         BTS         Total         INF           \$ 87,572         \$ 41,872         \$ 129,444         \$ 247,634         \$ | INF   BTS   Total   INF   BTS | INF   BTS   Total   INF   BTS |  |

(in thousands, except share data)

|  | Three Months Ended September 28, 2019 |        |             |       |         |    | Nine Mon | ths E | nded Septem | ber 2 | 8, 2019 |
|--|---------------------------------------|--------|-------------|-------|---------|----|----------|-------|-------------|-------|---------|
|  | INF                                   |        | BTS         |       | Total   |    | INF      |       | BTS         |       | Total   |
| <b>Gross revenues by Customer</b>      |                                       | _      |             |       |         |    |          |       |             |       |         |
| Public and quasi-public sector         | \$<br>69,297                          | \$     | 17,899      | \$    | 87,196  | \$ | 209,837  | \$    | 50,471      | \$    | 260,308 |
| Private sector                         | 18,275                                |        | 25,561      |       | 43,836  |    | 37,797   |       | 78,235      |       | 116,032 |
| Total gross revenues                   | \$<br>87,572                          | \$     | 43,460      | \$    | 131,032 | \$ | 247,634  | \$    | 128,706     | \$    | 376,340 |
|  | Three Mor                             | nths E | nded Septem | ber 2 | 9, 2018 |    | Nine Mon | ths E | nded Septem | ber 2 | 9, 2018 |
|  | INF                                   |        | BTS         |       | Total   |    | INF      |       | BTS         |       | Total   |
| <b>Gross revenues by Customer</b>      |                                       |        |             |       |         |    |          |       |             |       |         |
| Public and quasi-public sector         | \$<br>58,157                          | \$     | 15,469      | \$    | 73,626  | \$ | 162,193  | \$    | 49,519      | \$    | 211,712 |
| Private sector                         | 5,357                                 |        | 25,202      |       | 30,559  |    | 16,338   |       | 74,687      |       | 91,025  |
| Total gross revenues                   | \$<br>63,514                          | \$     | 40,671      | \$    | 104,185 | \$ | 178,531  | \$    | 124,206     | \$    | 302,737 |
|  | Three Mor                             | iths E | nded Septem | ber 2 | 8. 2019 |    | Nine Mon | ths E | nded Septem | ber 2 | 8. 2019 |
|  | <br>INF                               |        | BTS         |       | Total   |    | INF      |       | BTS         |       | Total   |
| <b>Gross revenues by Contract Type</b> |                                       |        |             |       |         | -  |          |       |             |       |         |
| Cost-reimbursable contracts            | \$<br>84,426                          | \$     | 31,797      | \$    | 116,223 | \$ | 240,166  | \$    | 99,935      | \$    | 340,101 |
| Fixed-unit price contracts             | 3,146                                 |        | 11,663      |       | 14,809  |    | 7,468    |       | 28,771      |       | 36,239  |
| Total gross revenues                   | \$<br>87,572                          | \$     | 43,460      | \$    | 131,032 | \$ | 247,634  | \$    | 128,706     | \$    | 376,340 |
|  | Three Moi                             | nths E | nded Septem | ber 2 | 9, 2018 |    | Nine Mon | ths E | nded Septem | ber 2 | 9, 2018 |
|  | <br>INF                               |        | BTS         |       | Total   |    | INF      |       | BTS         |       | Total   |
| <b>Gross revenues by Contract Type</b> |                                       | -      |             |       |         |    |          |       |             |       |         |
| Cost-reimbursable contracts            | \$<br>63,507                          | \$     | 32,196      | \$    | 95,703  | \$ | 178,255  | \$    | 98,756      | \$    | 277,011 |
| Fixed-unit price contracts             | 7                                     |        | 8,475       |       | 8,482   |    | 276      |       | 25,450      |       | 25,726  |
| Total gross revenues                   | \$<br>63,514                          | \$     | 40,671      | \$    | 104,185 | \$ | 178,531  | \$    | 124,206     | \$    | 302,737 |

## Note 16 - Leases

Our operating leases consist of various office facilities, which we lease from unrelated parties. We use a portfolio approach to account for such leases due to the similarities in characteristics and apply an incremental borrowing rate equal to the interest rate of our existing secured line of credit. Our office leases with an initial term of 12 months or less are not recorded on the balance sheet. We account for lease components (e.g. fixed payments including rent, real estate taxes and common area maintenance costs) as a single lease component. Some of our leases include one or more options to renew the lease term at our sole discretion; however, these are not included in the calculation of our lease liability or ROU lease asset because they are not reasonably certain of exercise.

We also lease vehicles through a fleet leasing program. The payments for the vehicles are based on the terms selected. We have determined that it is reasonably certain that the leased vehicles will be held beyond the period in which the entire capitalized value of the vehicle has been paid to the lessor. As such, the capitalized value is the delivered price of the vehicle. Our vehicle leases are classified as financing leases.

(in thousands, except share data)

| Leases                  | Classification  | Septer | <b>September 28, 2019</b> |  |  |  |  |
|-------------------------|---|--------|---------------------------|--|--|--|--|
| Assets                  |   |        |                           |  |  |  |  |
| Operating lease assets  | Right-of-use lease asset, net (1)                         | \$     | 42,366                    |  |  |  |  |
| Finance lease assets    | Property and equipment, net (1)                           |        | 2,311                     |  |  |  |  |
| Total leased assets     |   | \$     | 44,677                    |  |  |  |  |
| Liabilities             |   |        |                           |  |  |  |  |
| Current                 |   |        |                           |  |  |  |  |
| Operating               | Accrued liabilities                                       | \$     | 10,768                    |  |  |  |  |
|                         |   |        |                           |  |  |  |  |
| Finance                 | Current portion of notes payable and other obligations    |        | 896                       |  |  |  |  |
| Noncurrent              |   |        |                           |  |  |  |  |
| Operating               | Long-term lease liability                                 |        | 32,781                    |  |  |  |  |
|                         |   |        |                           |  |  |  |  |
| Finance                 | Notes payable and other obligations, less current portion |        | 1,780                     |  |  |  |  |
| Total lease liabilities |   | \$     | 46,225                    |  |  |  |  |

(1): At September 28, 2019, operating right of-use lease assets and finance lease assets are recorded net of accumulated amortization of \$6,770 and \$1,365, respectively.

|  |                                   | Thr | ee Months Ended  | <b>Nine Months Ended</b> |       |  |  |
|--|-----------------------------------|-----|------------------|--------------------------|-------|--|--|
| Lease Cost                             | Classification                    | Sep | otember 28, 2019 | September 28, 2019       |       |  |  |
| Operating lease cost                   | Facilities and facilities related | \$  | 2,995            | \$                       | 8,237 |  |  |
| Finance lease cost                     |                                   |     |                  |                          |       |  |  |
| Amortization of financing lease assets | Depreciation and amortization     |     | 193              |                          | 517   |  |  |
| Interest on lease liabilities          | Interest expense                  |     | 26               |                          | 71    |  |  |
| Total lease cost                       |                                   | \$  | 3,214            | \$                       | 8,825 |  |  |

| Maturity of Lease Liabilities      | <b>Operating Leases</b> | Finance Leases |  |
|------------------------------------|-------------------------|----------------|--|
| 2019                               | \$ 3,257                | \$ 234         |  |
| 2020                               | 11,742                  | 975            |  |
| 2021                               | 10,079                  | 874            |  |
| 2022                               | 7,325                   | 611            |  |
| 2023                               | 5,934                   | 353            |  |
| Thereafter                         | 9,965                   | 103            |  |
| Total lease payments               | 48,302                  | 3,150          |  |
| Less: Interest                     | 4,753                   | 474            |  |
| Present value of lease liabilities | \$ 43,549               | \$ 2,676       |  |

(in thousands, except share data)

| Weighted - Average Remaining Lease Term (Years) | September 28, 2019 |
|---|--------------------|
| Operating leases                                | 5.3                |
| Finance leases                                  | 2.7                |
| Weighted - Average Discount Rate                |                    |
| Operating leases                                | 4%                 |
| Finance leases                                  | 7%                 |

|  |    | Nine Months Ended  |  |  |
|--|----|--------------------|--|--|
| Supplemental Cash Flow Information                             |    | September 28, 2019 |  |  |
| Operating cash flows from operating leases                     | \$ | 7,833              |  |  |
| Financing cash flows from finance leases                       | \$ | 543                |  |  |
| Right-of-use assets obtained in exchange for lease obligations |    |                    |  |  |
| Operating leases   | \$ | 14,251             |  |  |

Future minimum payments under non-cancelable operating leases as of December 29, 2018 were as follows:

| Years Ended                  |    | Amount |  |
|------------------------------|----|--------|--|
| 2019                         | \$ | 9,506  |  |
| 2020                         |    | 8,054  |  |
| 2021                         |    | 7,224  |  |
| 2022                         |    | 5,364  |  |
| 2023                         |    | 4,504  |  |
| Thereafter                   |    | 7,704  |  |
| Total minimum lease payments | \$ | 42,356 |  |

(in thousands, except share data)

#### Note 17 – Subsequent Events

On November 6, 2019, the Company and one of its direct subsidiaries ("Merger Sub," and together with NV5, the "NV5 Parties"), and Geospatial Holdings Inc. ("Geospatial") and Arlington Capital Partners III, L.P., a Delaware limited partnership, solely in its capacity as representative for the stockholders and optionholders of Geospatial, entered into a definitive Agreement and Plan of Merger (the "Merger Agreement"). The Merger Agreement provides that, at the effective time and upon the terms and subject to the conditions set forth therein and in accordance with applicable law, Merger Sub will merge with and into Geospatial (the "Merger") with Geospatial continuing as the surviving entity after the Merger and a direct, wholly-owned subsidiary of the Company. The board of directors of NV5 approved the Merger Agreement, the Merger and the other transactions contemplated by the Merger Agreement.

The Merger is structured as an all cash transaction, and the Company estimates that its aggregate obligations under the Merger Agreement will be approximately \$303 million. In connection with the Merger, the Company entered into a commitment letter (the "Commitment Letter") with Bank of America, N.A., BofA Securities, Inc., PNC Bank, National Association and PNC Capital Markets LLC (collectively, the "Lead Arrangers") to amend the Senior Credit Facility to provide additional borrowing facilities (the "Incremental Facility"), including a \$150 million term loan tranche and a \$90 million increase in the existing \$125 million revolving facility, for a total availability of \$365 million. The funding of the Incremental Facility provided for in the Commitment Letter is contingent on the satisfaction of customary conditions, including but not limited to (i) execution and delivery of definitive documentation with respect to the Incremental Facility in accordance with the terms set forth in the Commitment Letter, and (ii) consummation of the merger in accordance with the executed merger agreement provided to the Lead Arrangers not earlier than December 20, 2019.

#### ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS.

The following discussion and analysis of the financial condition and results of operations of NV5 Global, Inc. and its subsidiaries (collectively, the "Company," "we," "our," "us" or "NV5 Global") should be read in conjunction with the financial statements included elsewhere in this Quarterly Report and the audited financial statements for the year ended December 29, 2018, included in our Annual Report on Form 10-K. This Quarterly Report contains, in addition to unaudited historical information, forward-looking statements, which involve risk and uncertainties. The words "believe," "expect," "estimate," "may," "will," "could," "plan," or "continue" and similar expressions are intended to identify forward-looking statements. Our actual results could differ materially from the results those anticipated in such forward-looking statements. Factors that could cause or contribute to such differences in results and outcomes include, those discussed under the headings "Risk Factors" in our Annual Report on Form 10-K for the year ended December 29, 2018 and this Quarterly Report on Form 10-Q, if any. Readers are urged not to place undue reliance on these forward-looking statements, which speak only as of the date of this Quarterly Report on Form 10-Q. We undertake no obligation to (and we expressly disclaim any obligation to) revise or update any forward-looking statement, whether as a result of new information, subsequent events, or otherwise (except as may be required by law), in order to reflect any event or circumstance which may arise after the date of this Quarterly Report on Form 10-Q. Amounts presented are in thousands, except per share data.

## Overview

We are a provider of professional and technical engineering and consulting solutions to public and private sector clients. We focus on the infrastructure, energy, construction, real estate, and environmental markets. We primarily focus on the following business service verticals: construction quality assurance, infrastructure, energy, program management, and environmental solutions. Our primary clients include U.S. federal, state, municipal, and local government agencies, and military and defense clients. We also serve quasi-public and private sector clients from the education, healthcare, energy, and public utilities, including schools, universities, hospitals, health care providers, insurance providers, large utility service providers, and large to small energy producers.

## Recent Acquisitions

On July 2, 2019, the Company acquired WHPacific, Inc. ("WHPacific"), a leading provider of design engineering and surveying services serving Washington, Oregon, Idaho, New Mexico, Arizona and California for a cash purchase price of \$9,000. In order to determine the fair values of tangible and intangible assets acquired and liabilities assumed for WHPacific, the Company engaged a third-party independent valuation specialist to assist in the determination of fair values. The Company expects to finalize the purchase price allocation with respect to this transaction by the end of the fourth quarter 2019.

On July 1, 2019, the Company acquired GeoDesign, Inc. ("GeoDesign"), a geotechnical, environmental, geological, mining and pavement engineering company serving Washington, Oregon and California. The aggregate purchase price is up to \$12,800, including \$8,500 of cash, \$2,000 in promissory note (bearing interest at 4%), payable in four equal installments of \$500 due on the first, second, third and fourth anniversaries of July 1, 2019 and \$375 of the Company's common stock (4,731 shares) issued at the closing date. The purchase price also includes \$425 of the Company's common stock payable on the first and second anniversary of July 1, 2019. Further, the purchase price includes a \$1,500 earn-out of cash, which was recorded at an estimated fair value of \$1,456. In order to determine the fair values of tangible and intangible assets acquired and liabilities assumed for GeoDesign, the Company engaged a third-party independent valuation specialist to assist in the determination of fair values. The Company expects to finalize the purchase price allocation with respect to this transaction by the end of the fourth quarter 2019.

On June 3, 2019, the Company acquired Alta Environmental, L.P. ("Alta"), a consulting firm specializing in air quality, environmental building sciences, water resources, site assessment and remediation as well as environmental health and safety compliance services. The aggregate purchase price is up to \$6,500, including \$4,000 of cash and \$2,000 in promissory note (bearing interest at 4%), payable in four equal installments of \$500 due on the first, second, third and fourth anniversaries of June 3, 2019. Further, the purchase price includes a \$500 earn-out of cash, which was recorded at an estimated fair value of \$485. In order to determine the fair values of tangible and intangible assets acquired and liabilities assumed for Alta, the Company engaged a third-party independent valuation specialist to assist in the determination of fair values. The Company expects to finalize the purchase price allocation with respect to this transaction by the end of the fourth quarter 2019.

On June 3, 2019, the Company acquired Page One Consultants ("Page One"), a program management and construction quality assurance firm based in Orlando, Florida. The aggregate purchase price is up to \$3,900, including \$2,000 of cash, \$1,000 in promissory note (bearing interest at 3%), payable in three equal installments of \$333 due on the first, second and third anniversaries of June 3, 2019 and \$200 of the Company's common stock (2,647 shares) issued at the closing date. The purchase price also includes \$200 of the Company's common stock payable on the first anniversary of June 3, 2019. Further, the purchase price includes a \$500 earn-out of cash, which was recorded at an estimated fair value of \$448. In order to determine the fair values of tangible and intangible assets acquired and liabilities assumed for Page One, the Company engaged a third-party independent valuation specialist to assist in the determination of fair values. The Company expects to finalize the purchase price allocation with respect to this transaction by the end of the fourth quarter 2019.

On March 22, 2019, the Company acquired The Sextant Group, Inc. ("The Sextant Group"), a national leading provider of audiovisual, information and communications technology, acoustics consulting, and design services headquartered in Pittsburgh, PA. The Sextant Group provides services throughout the U.S. and is well-known for creating integrated technology solutions for a wide range of public and private sector clients. The aggregate purchase price is up to \$11,000, including \$7,000 of cash and \$4,000 in promissory note (bearing interest at 4%), payable in four equal installments of \$1,000 due on the first, second, third and fourth anniversaries of March 22, 2019. In order to determine the fair values of tangible and intangible assets acquired and liabilities assumed for The Sextant Group, the Company engaged a third-party independent valuation specialist to assist in the determination of fair values. The Company expects to finalize the purchase price allocation with respect to this transaction by the end of the fourth quarter 2019.

On December 31, 2018, the Company acquired certain assets of Celtic Energy, Inc. ("Celtic"), a nationally recognized energy consulting firm that specializes in energy project management and oversight. The aggregate purchase price is up to \$1,900, including \$1,000 in cash, \$300 in promissory note (bearing interest at 3%), payable in three equal installments of \$100 on the first, second and third anniversaries of December 31, 2018 and \$200 of the Company's common stock (3,227 shares) issued at the closing date. The purchase price also includes \$200 of the Company's common stock payable on the first anniversary of December 31, 2018. Further, the purchase price includes a \$200 earn-out of cash, which was recorded at an estimated fair value of \$181. In order to determine the fair values of tangible and intangible assets acquired and liabilities assumed for Celtic, the Company performed a purchase price allocation.

#### Segments

Our operations are organized into two reportable segments:

- Infrastructure (INF) includes our engineering, civil program management, and construction quality assurance, testing and inspection practices
- Building, Technology & Sciences (BTS) includes our energy, environmental and buildings program management practices

For additional information regarding our reportable segments, see "Reportable Segments" of the "Notes to Consolidated Financial Statements" included elsewhere herein.

## **Critical Accounting Policies and Estimates**

For a discussion of our critical accounting estimates, see "Management's Discussion and Analysis of Financial Condition and Results of Operations" that is included in the 2018 Form 10-K.

## **Results of Operations**

## **Consolidated Results of Operations**

The following table represents our condensed results of operations for the periods indicated (dollars in thousands):

|   | <b>Three Months Ended</b> |          |                    | Nine Months Ended |                    |          |                    |          |
|---|---------------------------|----------|--------------------|-------------------|--------------------|----------|--------------------|----------|
|   | <b>September 28, 2019</b> |          | September 29, 2018 |                   | September 28, 2019 |          | September 29, 2018 |          |
|   |                           |          |                    |                   |                    |          |                    |          |
| Gross revenues                                      | \$                        | 131,032  | \$                 | 104,185           | \$                 | 376,340  | \$                 | 302,737  |
| Less sub-consultant services and other direct costs |                           | (27,111) |                    | (19,736)          |                    | (82,213) |                    | (56,888) |
|   |                           |          |                    |                   |                    | _        |                    |          |
| Net revenues (1)                                    |                           | 103,921  |                    | 84,449            |                    | 294,127  |                    | 245,849  |
| Direct salary and wages costs                       |                           | 40,426   |                    | 34,475            |                    | 113,762  |                    | 98,542   |
|   |                           |          |                    |                   |                    |          |                    |          |
| Gross profit  |                           | 63,496   |                    | 49,974            |                    | 180,365  |                    | 147,307  |
|   |                           |          |                    |                   |                    |          |                    |          |
| Operating expenses                                  |                           | 55,671   |                    | 40,000            |                    | 155,533  |                    | 121,682  |
|   |                           |          |                    |                   |                    |          |                    |          |
| Income from operations                              |                           | 7,825    |                    | 9,974             |                    | 24,832   |                    | 25,625   |
|   |                           |          |                    |                   |                    |          |                    |          |
| Interest expense                                    |                           | (421)    |                    | (451)             |                    | (1,230)  |                    | (1,712)  |
|   |                           |          |                    |                   |                    |          |                    |          |
| Income tax expense                                  |                           | (1,560)  |                    | (2,238)           |                    | (3,422)  |                    | (4,716)  |
|   |                           |          |                    |                   |                    | <u> </u> |                    |          |
| Net income  | \$                        | 5,843    | \$                 | 7,285             | \$                 | 20,180   | \$                 | 19,197   |

<sup>(1)</sup> Net Revenues is not a measure of financial performance under GAAP. Gross revenues include sub-consultant costs and other direct costs which are generally pass-through costs. The Company believes that Net Revenues, which is a non-U.S. GAAP financial measure commonly used in our industry, enhances investors' ability to analyze our business trends and performance because it substantially measures the work performed by our employees.

## Three Months Ended September 28, 2019 Compared to the Three Months Ended September 29, 2018.

Gross and Net Revenues

Our consolidated gross revenues increased by \$26,847, or 26% in the three months ended September 28, 2019 compared to the three months ended September 29, 2018. Our consolidated net revenues increased by \$19,472, or 23% in the three months ended September 28, 2019 compared to the three months ended September 29, 2018. The increase in gross and net revenues was primarily due to the contribution from acquisitions completed since the third quarter of 2018. The growth in revenues was attributable to increases in the following:

- Energy distribution services
- Infrastructure engineering services
- · Energy and environmental services
- Civil and building program management services

Gross Profit

As a percentage of gross revenues, our gross profit margin was 48.5% and 48.0% for the three months ended September 28, 2019 and September 29, 2018, respectively.

Operating expenses

Our operating expenses increased \$15,671, or 39% for the three months ended September 28, 2019 compared to the three months ended September 29, 2018. The increases in operating expenses were primarily due to increases in payroll related costs of \$8,531, general and administrative costs of \$3,472 and depreciation and amortization of \$2,494. The increases in costs support our increases in revenues and are related primarily to the operating expenses of our acquired entities.

Income taxes

Our consolidated effective income tax rate was 22.9% and 24.5% for the three months ended September 28, 2019 and September 29, 2018, respectively. Our tax provision includes an income tax benefit related to the vesting of restricted stock totaling \$121 and \$95 for the three months ended September 28, 2019 and September 29, 2018, respectively.

#### Nine Months Ended September 28, 2019 Compared to the Nine Months Ended September 29, 2018.

Gross and Net Revenues

Our consolidated gross revenues increased by \$73,603, or 24% in the nine months ended September 28, 2019 compared to the nine months ended September 29, 2018. Our consolidated net revenues increased by \$48,278, or 20% in the nine months ended September 28, 2019 compared to the nine months ended September 29, 2018. The increase in gross and net revenues was primarily due to the contribution from acquisitions completed since the second quarter of 2018. The growth in revenues was attributable to increases in the following:

- Energy distribution services
- Infrastructure engineering services
- · Energy and environmental services
- · Civil and building program management services

Gross Profit

As a percentage of gross revenues, our gross profit margin was 47.9% and 48.7% for the nine months ended September 28, 2019 and September 29, 2018, respectively. The decrease in gross profit margin was due to an increased use of sub-consultants as well as higher other direct costs in 2019 compared to 2018.

Operating expenses

Our operating expenses increased \$33,851, or 28% for the nine months ended September 28, 2019 compared to the nine months ended September 29, 2018. The increases in operating expenses were primarily a result of increases in payroll related costs of \$17,309, general and administrative costs of \$7,438 and depreciation and amortization of \$7,248. The increases in costs support our increases in revenues and are related primarily to the operating expenses of our acquired entities.

Income taxes

Our consolidated effective income tax rate was 24.1% and 25.0% for the nine months ended September 28, 2019 and September 29, 2018, respectively. Our tax provision includes an income tax benefit related to the vesting of restricted stock totaling \$2,592 and \$1,210 for the nine months ended September 28, 2019 and September 29, 2018, respectively.

## Segment Results of Operations

The following tables set forth summarized financial information concerning our reportable segments (dollars in thousands):

|                             |    | <b>Three Months Ended</b> |    |                    |    | <b>Nine Months Ended</b> |    |                  |  |
|-----------------------------|----|---------------------------|----|--------------------|----|--------------------------|----|------------------|--|
|                             | 5  | September 28, 2019        |    | September 29, 2018 |    | September 28, 2019       |    | otember 29, 2018 |  |
| Gross revenues              | _  |                           |    |                    |    |                          |    |                  |  |
| INF                         | \$ | 89,035                    | \$ | 64,053             | \$ | 250,354                  | \$ | 180,611          |  |
| BTS                         | \$ | 43,060                    | \$ | 41,441             | \$ | 128,706                  | \$ | 125,344          |  |
| Segment income before taxes |    |                           |    |                    |    |                          |    |                  |  |
| INF                         | \$ | 14,008                    | \$ | 11,108             | \$ | 41,273                   | \$ | 28,951           |  |
| BTS                         | \$ | 6,871                     | \$ | 7,236              | \$ | 20,287                   | \$ | 21,069           |  |

For additional information regarding our reportable segments, see Note 15 of the notes to the unaudited consolidated financial statements included elsewhere in this Quarterly Report on Form 10-Q.

#### Three Months Ended September 28, 2019 Compared to Three Months Ended September 29, 2018

## INF Segment

Our gross revenues from INF reportable segment increased \$24,982, or 39% during the three months ended September 28, 2019 compared to the three months ended September 29, 2018. The increase in gross revenues is due to the contribution from acquisitions completed since thethird quarter of 2018. The growth in revenues was attributable to increases in:

- Energy distribution services
- Infrastructure engineering services
- Civil program management services

Segment Income before Taxes from INF increased \$2,900, or 26% during the three months ended September 28, 2019 compared to the three months ended September 29, 2018. The increase was primarily due to acquisitions completed since the third quarter of 2018.

## **BTS Segment**

Our gross revenues from BTS increased \$1,619, or 4% during the three months ended September 28, 2019 compared to the three months ended September 29, 2018. The increase in gross revenues was primarily due to the contribution from acquisitions completed since the third quarter of 2018, partially offset by lower revenues from our international operations.

Segment Income before Taxes from BTS decreased \$365, or 5% during the three months ended September 28, 2019 compared to the three months ended September 29, 2018. The decrease was primarily the result of lower earnings from our international operations.

## Nine Months Ended September 28, 2019 Compared to Nine Months Ended September 29, 2018

## INF Segment

Our gross revenues from INF reportable segment increased \$69,743, or 39% during the nine months ended September 28, 2019 compared to the nine months ended September 29, 2018. The increase in gross revenues is due to the contribution from acquisitions completed since the third quarter of 2018. The growth in revenues was attributable to increases in:

- Energy distribution services
- Infrastructure engineering services
- · Civil program management services

Segment Income before Taxes from INF increased \$12,322, or 43% during the nine months ended September 28, 2019 compared to the nine months ended September 29, 2018. The increase was primarily due to contributions from acquisitions completed since the third quarter of 2018.

#### **BTS Segment**

Our gross revenues from BTS increased \$3,362, or 3% during the nine months ended September 28, 2019 compared to the nine months ended September 29, 2018. The increase in gross revenues is due to the contribution from acquisitions completed since the third quarter of 2018.

Segment Income before Taxes from BTS decreased \$782, or 4% during the nine months ended September 28, 2019 compared to the nine months ended September 29, 2018. The decrease was primarily a result of a slowdown in the gaming industry and our international operations, partially offset by increases in revenues associated with building program management and mechanical, electrical and plumbing services.

## **Liquidity and Capital Resources**

Our principal sources of liquidity are our cash and cash equivalents balances, cash flows from operations, borrowing capacity under our Senior Credit Facility, and access to financial markets. Our principal uses of cash are operating expenses, working capital requirements, capital expenditures, repayment of debt, and acquisition expenditures. We believe our sources of liquidity, including cash flows from operations, existing cash and cash equivalents and borrowing capacity under our Senior Credit Facility (as proposed to be amended) will be sufficient to meet our projected cash requirements for at least the next twelve months. We will monitor our capital requirements thereafter to ensure our needs are in line with available capital resources.

Operating activities

Our business provided \$21,557 of net cash from operations during the nine months ended September 28, 2019, an increase of \$3,987, or 23% compared to \$17,570 during the nine months ended September 29, 2018. The change was primarily due to higher earnings after adding back non-cash adjustments, which totaled \$17,997, partially offset by a \$14,010 change in working capital compared to the nine months ended September 29, 2018.

Investing activities

During the nine months ended September 28, 2019 and September 29, 2018, net cash used in investing activities totaled \$31,175 and \$30,042, respectively. The increase in cash used in investing activities was primarily a result of increased acquisition activity.

Financing activities

Cash flows provided by financing activities during the nine months ended September 28, 2019 totaled \$304 compared to net cash provided by financing activities of \$49,923 during the nine months ended September 29, 2018. The decrease was primarily due to the net proceeds from the August 2018 public offering of \$93,469 offset by principal repayments of \$43,910 towards the Senior Credit Facility and notes payable during the nine months ended September 29, 2018.

#### **Financing**

Senior Credit Facility

On December 20, 2018, we entered into an amendment to a Credit Agreement (the "Credit Agreement") dated December 7, 2016 with Bank of America, N.A. ("Bank of America") and Merrill Lynch, Pierce, Fenner & Smith Incorporated ("MLPFS"). Pursuant to the amended Credit Agreement, Bank of America agreed to be the sole administrative agent for a five-year \$125,000 Senior Secured Revolving Credit Facility ("Senior Credit Facility") to the Company and, together with PNC Bank, National Association and Regions Bank as the other lenders under the Senior Credit Facility, has committed to lend to the Company all of the Senior Credit Facility, subject to certain terms and conditions. The Senior Credit Facility is secured by a first priority lien on substantially all of the assets of the Company. MLPFS has undertaken to act as sole lead arranger and sole book manager for the Senior Credit Facility. In addition, the Senior Credit Facility includes an accordion feature permitting the Company to request an increase in the Senior Credit Facility by an additional amount of up to \$100,000. The Senior Credit Facility includes a \$20,000 sublimit for the issuance of standby letters of credit and a \$15,000 sublimit for swingline loans. The proceeds of the Senior Credit Facility are intended to be used (i) to finance permitted acquisitions, (ii) for capital expenditures, and (iii) for general corporate purposes.

Borrowings under the Credit Agreement are at variable rates which are, at our option, tied to a Eurocurrency rate equal to LIBOR (London Interbank Offered Rate) plus an applicable rate or a base rate denominated in U.S. dollars. Interest rates are subject to change based on our Consolidated Senior Leverage Ratio (as defined in the Credit Agreement).

The Senior Credit Facility contains certain financial covenants, including a maximum leverage ratio of 4.0:1 and a minimum fixed charge coverage ratio of 1.20:1. Furthermore, the Senior Credit Facility also contains financial reporting covenant provisions and other covenants, representations, warranties, indemnities, and events of default that are customary for facilities of this type. As of September 28, 2019 and December 29, 2018, the Company is in compliance with the financial covenants. As of September 28, 2019, there was \$10,000 outstanding on the Senior Credit Facility. As of December 29, 2018, we had no outstanding balance on the Senior Credit Facility.

In connection with a pending acquisition, the Company entered into a commitment letter (the "Commitment Letter") with Bank of America, N.A., BofA Securities, Inc., PNC Bank, National Association and PNC Capital Markets LLC (collectively, the "Lead Arrangers") to amend the Senior Credit Facility to provide additional borrowing facilities (the "Incremental Facility"), including a \$150 million term loan tranche and a \$90 million increase in the existing \$125 million revolving facility, for a total availability of \$365 million. The funding of the Incremental Facility provided for in the Commitment Letter is contingent on the satisfaction of customary conditions, including but not limited to (i) execution and delivery of definitive documentation with respect to the Incremental Facility in accordance with the terms set forth in the Commitment Letter, and (ii) consummation of the merger in accordance with the executed merger agreement provided to the Lead Arrangers not earlier than December 20, 2019.

## Other Obligations

On July 1, 2019, the Company acquired GeoDesign. The purchase price allowed for the payment of \$425 in shares of the Company's stock or a combination of cash and shares of the Company's stock, at our discretion, payable on the first and second anniversary of July 1, 2019. At September 28, 2019, the outstanding balance of this obligation was \$382.

On June 3, 2019, the Company acquired Page One. The purchase price allowed for the payment of \$200 in shares of the Company's stock or a combination of cash and shares of the Company's stock, at our discretion, payable on the first anniversary of June 3, 2019. At September 28, 2019, the outstanding balance of this obligation was \$181.

On December 31, 2018, the Company acquired certain assets of Celtic. The purchase price allowed for the payment of \$200 in shares of the Company's stock or a combination of cash and shares of the Company's stock, at our discretion, payable on the first anniversary of December 31, 2018. At September 28, 2019, the outstanding balance of this obligation was \$181.

On November 2, 2018, the Company acquired CHI. The purchase price allowed for the payment of \$3,000 in shares of the Company's stock or a combination of cash and shares of the Company's stock, at our discretion, payable in three equal annual installments. At September 28, 2019 and December 29, 2018, the outstanding balance of this obligation was \$2,631.

On February 2, 2018, the Company acquired CSA. The purchase price allowed for the payment of \$250 in shares of the Company's stock or a combination of cash and shares of the Company's stock, at our discretion, payable in two equal annual installments. At September 28, 2019 and December 29, 2018, the outstanding balance of this obligation was \$111 and \$222, respectively.

On January 12, 2018, the Company acquired all of the outstanding equity interest in Butsko. The purchase price allowed for the payment of \$600 in shares of the Company's stock or a combination of cash and shares of the Company's stock, at our discretion, payable in two equal annual installments. At September 28, 2019 and December 29, 2018, the outstanding balance of this obligation was \$267 and \$534, respectively.

On September 6, 2017, the Company acquired all of the outstanding equity interest in Marron. The purchase price allowed for the payment of \$133 in shares of the Company's stock or a combination of cash and shares of the Company's stock, at our discretion, payable in two equal annual installments. There was no outstanding balance on this obligation as of September 28, 2019. At December 29, 2018, the outstanding balance of this obligation was \$55.

On June 6, 2017, the Company acquired all of the outstanding equity interest in RDK. The purchase price allowed for the payment of \$1,333 in shares of the Company's stock or a combination of cash and shares of the Company's stock, at our discretion, payable in two equal annual installments. There was no outstanding balance on this obligation as of September 28, 2019. At December 29, 2018, the outstanding balance of this obligation \$504.

On May 20, 2016, the Company acquired all of the outstanding equity interests of Dade Moeller. The purchase price allowed for the payment of \$3,000 in shares of the Company's stock or a combination of cash and shares of the Company's stock, at our discretion, payable in three equal annual installments of \$1,000. There was no outstanding balance on this obligation as of September 28, 2019. At December 29, 2018, the outstanding balance of this obligation was \$936.

Uncollateralized Promissory Notes

On July 1, 2019, the Company acquired GeoDesign. The purchase price included an uncollateralized \$2,000 promissory note bearing interest at 4.0% ("GeoDesign Note") and payable in four equal annual installments. The outstanding balance of the GeoDesign Note was \$2,000 as of September 28, 2019.

On June 3, 2019, the Company acquired Alta. The purchase price included an uncollateralized \$2,000 promissory note bearing interest at 4.0% ("Alta Note") and payable in four equal annual installments. The outstanding balance of the Alta Note was \$2,000 as of September 28, 2019.

On June 3, 2019, the Company acquired Page One. The purchase price included an uncollateralized \$1,000 promissory note bearing interest at 3.0% ("Page One Note") and payable in three equal annual installments. The outstanding balance of the Page One Note was \$1,000 as of September 28, 2019.

On March 22, 2019, we acquired The Sextant Group. The purchase price included an uncollateralized \$4,000 promissory note bearing interest at 4% ("The Sextant Group Note") and payable in four equal annual installments. The outstanding balance of The Sextant Group Note was \$4,000 as of September 28, 2019.

On December 31, 2018, we acquired certain assets of Celtic. The purchase price included an uncollateralized \$300 promissory note bearing interest at 3% (the "Celtic Note") payable in three equal annual installments. The outstanding balance of the Celtic note was \$300 as of September 28, 2019.

On November 2, 2018, we acquired CHI. The purchase price included an uncollateralized \$15,000 promissory note bearing interest at 3% (the "CHI Note") payable in four equal annual installments. The outstanding balance of the CHI Note was \$15,000 as of September 28, 2019 and December 29, 2018.

On August 24, 2018, the Company acquired CALYX. The purchase price included an uncollateralized \$4,000 promissory note bearing interest at 3.75% (the "CALYX Note") payable in four equal annual installments of \$1,000. The outstanding balance of the CALYX Note was \$3,000 and \$4,000 as of September 28, 2019 and December 29, 2018, respectively.

On February 2, 2018, the Company acquired CSA. The purchase price included an uncollateralized \$600 promissory note bearing interest at 3.0% (the "CSA Note") payable in four equal annual installments of \$150. The outstanding balance of the CSA Note was \$450 and \$600 as of September 28, 2019 and December 29, 2018, respectively.

On January 12, 2018, the Company acquired all of the outstanding equity interest in Butsko. The purchase price included an uncollateralized \$1,000 promissory note bearing interest at 3.0% (the "Butsko Note") payable in four equal annual installments of \$250. The outstanding balance of the Butsko Note was \$750 and \$1,000 as of September 28, 2019 and December 29, 2018, respectively.

On September 6, 2017, the Company acquired all of the outstanding interests in Marron. The purchase price included an uncollateralized \$300 promissory note bearing interest at 3.0% (the "Marron Note") payable in three equal annual installments of \$100. The outstanding balance of the Marron Note was \$100 and \$200 as of September 28, 2019 and December 29, 2018, respectively.

On June 6, 2017, the Company acquired all of the outstanding equity interest in RDK. The purchase price included an uncollateralized \$5,500 promissory note bearing interest at 3.0% (the "RDK Note") payable in four equal annual installments of \$1,375. The outstanding balance of the RDK Note was \$2,750 and \$4,125 as of September 28, 2019 and December 29, 2018, respectively.

On May 4, 2017, the Company acquired all of the outstanding equity interest in H&K. The purchase price included an uncollateralized \$600 promissory note bearing interest at 3.0% (the "H&K Note") payable in four equal annual installments of \$150. The outstanding balance of the H&K Note was \$300 and \$450 as of September 28, 2019 and December 29, 2018, respectively.

On May 1, 2017, the Company acquired all of the outstanding equity interest in Lochrane. The purchase price included an uncollateralized \$1,650 promissory note bearing interest at 3.0% (the "Lochrane Note") payable in four equal annual installments of \$413. The outstanding balance of the Lochrane Note was \$825 and \$1,238 as of September 28, 2019 and December 29, 2018, respectively.

On December 6, 2016, the Company acquired all of the outstanding interests of CivilSource. The purchase price included an uncollateralized \$3,500 promissory note bearing interest at 3.0% (the "CivilSource Note") payable in four equal annual

installments of \$875. The outstanding balance of the CivilSource Note was \$1,606 and \$2,625 as of September 28, 2019 and December 29, 2018, respectively.

On November 30, 2016, the Company acquired all of the outstanding interests of Hanna. The purchase price included an uncollateralized \$2,700 promissory note bearing interest at 3.0% (the "Hanna Note") payable in four equal annual installments of \$675. The outstanding balance of the Hanna Note was \$1,350 as of September 28, 2019 and December 29, 2018.

On October 26, 2016, the Company acquired all of the outstanding interests of JBA. The purchase price included an uncollateralized \$7,000 promissory note bearing interest at 3.0% (the "JBA Note") payable in five equal annual installments of \$1,400. The outstanding balance of the JBA Note was \$4,200 as of September 28, 2019 and December 29, 2018.

On September 12, 2016, the Company acquired certain assets of Weir. The purchase price included an uncollateralized \$500 promissory note bearing interest at 3.0% (the "Weir Note") payable in four equal annual installments of \$125. The outstanding balance of the Weir Note was \$125 and \$250 as of September 28, 2019 and December 29, 2018, respectively.

On May 20, 2016, the Company acquired all of the outstanding equity interests of Dade Moeller. The purchase price included an aggregate of \$6,000 of uncollateralized promissory notes bearing interest at 3.0% (the "Dade Moeller Notes") payable in four equal annual installments of \$1,500. The outstanding balance of the Dade Moeller Notes was \$1,497 and \$3,036 as of September 28, 2019 and December 29, 2018, respectively.

On July 1, 2015, the Company acquired all of the outstanding equity interests of RBA. The purchase price included an uncollateralized \$4,000 promissory notes bearing interest at 3.0% (the "RBA Note") payable in four equal annual installments. There was no outstanding balance on the RBA Note as of September 28, 2019. The outstanding balance of the RBA Note was \$1,000 as of December 29, 2018.

On January 30, 2015, the Company acquired all of the outstanding equity interests of JLA. The purchase price included an uncollateralized \$1,250 promissory note bearing interest at 3.5% (the "JLA Note") payable in four equal annual installments of \$313. There was no outstanding balance on the JLA Note as of September 28, 2019. As of December 29, 2018, the outstanding balance of the JLA note was \$313.

#### **Off-Balance Sheet Arrangements**

We did not have any off-balance sheet arrangements as of September 28, 2019.

#### **Effects of Inflation**

Based on our analysis of the periods presented, we believe that inflation has not had a material effect on our operating results. There can be no assurance that future inflation will not have an adverse impact on our operating results and financial condition.

## **Recently Issued Accounting Pronouncements**

For information on recently issued accounting pronouncements, see Note 3 of the notes to the unaudited consolidated financial statements included elsewhere in this Quarterly Report on Form 10-Q.

#### **Cautionary Statement about Forward-Looking Statements**

Our disclosure and analysis in this Quarterly Report on Form 10-Q, contain "forward-looking" statements within the meaning of Section 27A of the Securities Act Section 21E of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), and the Private Securities Litigation Reform Act of 1995. From time to time, we also provide forward-looking statements in other materials we release to the public, as well as oral forward-looking statements. Forward-looking statements include, statements regarding our "expectations," "hopes," "beliefs," "intentions," or "strategies" regarding the future. In addition, any statements that refer to projections, forecasts, or other characterizations of future events or circumstances, including any underlying assumptions, are forward-looking statements. We have tried, wherever possible, to identify such statements by using words such as "anticipate," "believe," "expect," "intend," "estimate," "predict," "project," "may," "might," "should," "would," "will," "likely," "will likely result," "continue," "could," "future," "plan," "possible," "potential," "target," "forecast," "goal," "observe," "seek," "strategy" and other words and terms of similar meaning, but the absence of these words does not mean that a statement is not forward looking. The forward-looking statements in this Current Report on Form 10-Q reflect the Company's current views with respect to future events and financial performance.

Forward-looking statements are not historical factors and should not be read as a guarantee or assurance of future performance or results, and will not necessarily be accurate indications of the times at, or by, or if such performance or results will be achieved. Forward-looking statements are based on information available at the time those statements are made or management's good faith beliefs, expectations and assumptions as of that time with respect to future events. Because forward-looking statements relate to the future, they are subject to risks and uncertainties that could cause actual performance or results to differ materially from those expressed in or suggested by the forward-looking statements. Important factors that could cause such differences include:

- our ability to retain the continued service of our key professionals and to identify, hire and retain additional qualified professionals;
- changes in demand from the local and state government and private clients that we serve;
- general economic conditions, nationally and globally, and their effect on the demand and market for our services;
- fluctuations in our results of operations;
- the government's funding and budgetary approval process;
- the possibility that our contracts may be terminated by our clients;
- our ability to win new contracts and renew existing contracts;
- our dependence on a limited number of clients;
- · our ability to complete projects timely, in accordance with our customers' expectations, or profitability;
- our ability to successfully execute our mergers and acquisitions strategy, including the integration of new companies into our business;
- our ability to successfully manage our growth strategy;
- our ability to raise capital in the future;
- · competitive pressures and trends in our industry and our ability to successfully compete with our competitors;
- our ability to avoid losses under fixed-price contracts;
- the credit and collection risks associated with our clients;
- our ability to comply with procurement laws and regulations;
- changes in laws, regulations, or policies;
- the enactment of legislation that could limit the ability of local, state and federal agencies to contract for our privatized services;
- our ability to complete our backlog of uncompleted projects as currently projected;
- the risk of employee misconduct or our failure to comply with laws and regulations;
- our ability to control, and operational issues pertaining to, business activities that we conduct with business partners and other third parties;
- our need to comply with a number of restrictive covenants and similar provisions in our senior credit facility that generally limit our ability to (among other things) incur additional indebtedness, create liens, make acquisitions, pay dividends and undergo certain changes in control, which could affect our ability to finance future operations, acquisitions or capital needs;
- · significant influence by our principal stockholder and the existence of certain anti-takeover measures in our governing documents; and
- other factors identified throughout this Current Report on Form 10-Q, including those discussed under the headings "Risk Factors," "Management's Discussion and Analysis of Financial Condition and Results of Operations," and "Business."

The forward-looking statements contained in this report are based on our current expectations and beliefs concerning future developments and their potential effects on us. There can be no assurance that future developments affecting us will be those that we have anticipated. These forward-looking statements involve a number of risks, uncertainties, or assumptions, many of which are beyond our control, which may cause actual results or performance to be materially different from those expressed or

implied by these forward-looking statements. These risks and uncertainties include, those factors described in "Item 1A. Risk Factors" in our Annual Report on Form 10-K for the year ended December 29, 2018. We undertake no obligation to update or revise any forward-looking statements, whether as a result of new information, future events, or otherwise, except as may be required under applicable securities laws. You are advised, however, to consult any further disclosures we make on related subjects in our Form 10-Q, 8-K and 10-K reports filed with the SEC. Our Annual Report on Form 10-K filing for the fiscal year ended December 29, 2018 listed various important factors that could cause actual results to differ materially from expected and historic results. We note these factors for investors as permitted by the Private Securities Litigation Reform Act of 1995, as amended. Readers can find them in "Item 1A. Risk Factors" of that filing and under the same heading of this filing. You may obtain a copy of our Annual Report on Form 10-K through our website, <a href="https://www.nv5.com">www.nv5.com</a>. Information contained on our website is not incorporated into this report. In addition to visiting our website, you may read and copy any document we file with the SEC at <a href="https://www.sec.gov">www.sec.gov</a>.

## ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK.

We are exposed to certain market risks from transactions that are entered into during the normal course of business. We have not entered into derivative financial instruments for trading purposes. We have no significant market risk exposure to interest rate changes related to the promissory notes related to acquisitions since these contain fixed interest rates. Our only debt subject to interest rate risk is the Senior Credit Facility which rates are variable, at our option, tied to a Eurocurrency rate equal to LIBOR (London Interbank Offered Rate) plus an applicable rate or a base rate denominated in U.S. dollars. Interest rates are subject to change based on our Consolidated Senior Leverage Ratio (as defined in the Credit Agreement). As of September 28, 2019, there was \$10,000 outstanding on the Senior Credit Facility. A one percentage point change in the assumed interest rate of the Senior Credit Facility would not have a material impact on our market risk.

#### ITEM 4. CONTROLS AND PROCEDURES.

#### **Disclosure Controls and Procedures**

In connection with management's evaluation of the effectiveness of our internal control over financial reporting as of December 29, 2018, we identified a material weakness in our internal control over financial reporting related to revenues. This material weakness related solely to internal control deficiencies over the initial set up of project contracts in our project management system and adequate documentation to support the analysis of certain percentage of completion projects. The material weakness described herein did not result in a material misstatement to the Company's previously issued consolidated financial statements, nor in the consolidated financial statements included in the Company's Annual Report on Form 10-K for the year ended December 29, 2018.

The Company has made progress toward remediating this material weakness (as described below under "Remediation Status of Reported Material Weakness"). As of September 28, 2019, the Company had not completed its remediation.

#### **Evaluation of Disclosure Controls and Procedures**

As of the end of the period covered by this Quarterly Report on Form 10-Q, we carried out an evaluation, under the supervision and with the participation of our management, including our Chief Executive Officer and Chief Financial Officer, of the effectiveness of the design and operation of our disclosure controls and procedures (as such term is defined in Rules 13a-15(e) and 15d-15(e) under the Exchange Act). Although our Chief Executive Officer and Chief Financial Officer concluded that, as of the end of the period covered by this Quarterly Report on Form 10-Q, the Company's disclosure controls and procedures were not effective as a result of a material weakness in our internal control over financial reporting, remedial steps have been taken in the to address the weakness and improve the process. The material weakness described herein did not result in a material misstatement to the Company's previously issued consolidated financial statements, nor in the condensed consolidated unaudited financial statements included in this Quarterly Report on Form 10-Q.

## Remediation Status of Reported Material Weakness

Management continues to execute its plan to remediate the material weakness. As of September 28, 2019, management had performed the following activities:

• The Company completed its examination and analysis of the facts and circumstances giving rise to the material weakness under the supervision of the Chairman of the Audit Committee. The Company is addressing the examination findings through ongoing remediation. The Company continues to believe the remediation plan remains appropriate;

#### **Table of Contents**

- The Company has enhanced its processes over the initial set up of project contracts and adequate documentation to support the analysis of percentage
  of completion projects and has made system enhancements to its project management system; and
- The Company has enhanced its processes for analyzing trends in margins in order to strengthen controls for the proper recognition of revenue.

The remediation steps outlined above are expected to strengthen the Company's internal control over financial reporting. Management has begun testing the ongoing operating effectiveness of all new and modified controls and will consider the material weakness remediated after the applicable controls operate effectively for a sufficient period.

## Changes in Internal Control Over Financial Reporting

There were no changes to the Company's internal control over financial reporting as defined in Exchange Act Rules 13a-15(e) and 15d-15(e) that occurred during the quarter ended September 28, 2019 that have materially affected, or are reasonably likely to materially affect, the Company's internal control over financial reporting other than as described above.

#### PART II - OTHER INFORMATION

#### ITEM 1. LEGAL PROCEEDINGS.

From time to time, we are subject to various legal proceedings that arise in the normal course of our business activities. As of the date of this Quarterly Report on Form 10-Q, we are not a party to any litigation the outcome of which, if determined adversely to us, would individually or in the aggregate be reasonably expected to have a material adverse effect on our results of operations or financial position.

#### ITEM 1A. RISK FACTORS.

There have been no material changes to any of the principal risks that we believe are material to our business, results of operations and financial condition, from the risk factors previously disclosed in our Annual Report on Form 10-K for the year ended December 29, 2018, except as set forth below:

#### Risks Related to our Pending Acquisition of Geospatial Holdings Inc.

The failure to successfully integrate Geospatial's business and operations in the expected time frame may adversely affect the combined company's future results.

On November 6, 2019, NV5 and one of its direct subsidiaries ("Merger Sub," and together with NV5, the "NV5 Parties"), and Geospatial Holdings Inc. ("Geospatial") and Arlington Capital Partners III, L.P., a Delaware limited partnership, solely in its capacity as representative for the stockholders and optionholders of Geospatial, entered into a definitive Agreement and Plan of Merger (the "Merger Agreement") providing for the merger of Merger Sub with and into Geospatial (the "Merger"). NV5 believes that the acquisition of Geospatial will result in certain benefits, including certain cost synergies, significant new service offerings and operational efficiencies. However, to realize these anticipated benefits, the businesses of NV5 and Geospatial must be successfully combined. The success of the Merger will depend on the combined company's ability to realize these anticipated benefits from combining the businesses of NV5 and Geospatial. The combined company may fail to realize the anticipated benefits of the Merger for a variety of reasons, including the following:

- unanticipated issues in integration of information, communications, and other systems;
- · unanticipated incompatibility of logistics, marketing, and administration methods;
- maintaining employee morale and retaining key employees;
- integrating the business cultures of both companies:
- · preserving important strategic client relationships;
- · consolidating corporate and administrative infrastructures and eliminating duplicative operations; and
- coordinating geographically separate organizations

In addition, even if the operations of Geospatial are integrated successfully, we may not realize the full benefits of the acquisition, including the synergies, cost savings, or growth opportunities that we expect. These benefits may not be achieved within the anticipated time frame, or at all.

We expect to fund the Geospatial acquisition with significant additional indebtedness, which may not be available when we are obligated to consummate the Merger. In addition, if we are unable to generate or borrow sufficient cash to make payments on

#### **Table of Contents**

#### our indebtedness, our financial condition would be materially harmed, our business could fail, and shareholders may lose all of their investment.

In connection with the pending Merger, NV5 obtained a financing commitment to provide an incremental loan facility in the original principal amount of \$240 million (the "Incremental Facility") to fund the Merger under its existing credit facility, pursuant to a commitment letter (the "Commitment Letter") from Bank of America, N.A., BofA Securities, Inc., PNC Bank, National Association and PNC Capital Markets LLC. The funding of the Incremental Facility provided for in the Commitment Letter is contingent on the satisfaction of customary conditions, including but not limited to (i) execution and delivery of definitive documentation with respect to the Incremental Facility in accordance with the terms set forth in the Commitment Letter, and (ii) consummation of the Merger in accordance with the Merger Agreement not earlier than December 20, 2019. The actual documentation governing the Incremental Facility has not been finalized, and accordingly, the actual terms may differ from the description of such terms in the Commitment Letter. Additionally, pursuant to the Merger Agreement, we may become obligated to consummate the Merger as soon as December 6, 2019, which may result in our inability to close such acquisition unless we obtain relevant waivers from either Geospatial or the Lead Arrangers. While we believe the risk of such a timing issue is low, it is possible we will be obligated to complete the Merger prior to the availability of financing pursuant to the Commitment Letter.

Our ability to make scheduled payments on or to refinance our obligations under our amended Senior Credit Facility will depend on our financial and operating performance, which will be affected by economic, financial, competitive, business, and other factors, some of which are beyond our control. We cannot assure you that our business will generate sufficient cash flow from operations to service our indebtedness or to fund our other liquidity needs. If we are unable to meet our debt obligations or fund our other liquidity needs, we may need to restructure or refinance all or a portion of our indebtedness on or before maturity or sell certain of our assets. We cannot assure you that we will be able to restructure or refinance any of our indebtedness on commercially reasonable terms, if at all, which could cause us to default on our debt obligations and impair our liquidity. Any refinancing of our indebtedness could be at higher interest rates and may require us to comply with more onerous covenants, which could further restrict our business operations.

Unavailability or cancellation of third-party insurance representation and warranty coverage regarding the Merger Agreement would increase our overall risk exposure as well as result in the unavailability of the Incremental Facility.

One of the conditions to the closing of the Merger is NV5 having secured representation and warranty insurance coverage regarding the Merger Agreement, which is our only recourse in the event the representations and warranties in the Merger Agreement prove to be inaccurate subsequent to the Merger. Pursuant to the Commitment Letter, we are obligated to consummate the Merger on the terms set forth in the Merger Agreement, without certain waivers that would adversely affect the interests of the lenders under the Incremental Facility. If we are unable to obtain such insurance coverage from third-party insurers as part consummating the Merger, Geospatial may seek our waiver of such condition, which if granted would result in the unavailability of the Incremental Facility. In addition, such representation and warranty insurance will likely include certain deductibles and exclusions, which may result in our being directly subject to the risk associated with any such inaccuracies in the Merger Agreement. If the third-party insurer providing the coverage fails, suddenly cancels our coverage, or is otherwise unable to provide us with adequate insurance coverage, our overall risk exposure and our operational expenses would increase and the management of our business operations would be disrupted.

## ITEM 2. UNREGISTERED SALES OF EQUITY SECURITIES AND USE OF PROCEEDS.

#### **Recent Sales of Unregistered Securities**

During the three months ended September 28, 2019, we issued the following securities that were not registered under the Securities Act:

In September 2019, we issued 4,731 shares of our common stock as partial consideration for our acquisition of GeoDesign.

These shares were issued in reliance upon Section 4(a)(2) of the Securities Act as a transaction by an issuer not involving a public offering. For a description of these acquisitions, see Note 5, Business Acquisitions, to the consolidated interim financial statements appearing under Part I of this Quarterly Report on Form 10-Q.

## **Issuer Purchase of Equity Securities**

None.

#### ITEM 3. DEFAULTS UPON SENIOR SECURITIES.

None.

## ITEM 4. MINE SAFETY DISCLOSURES.

Not applicable.

## ITEM 5. OTHER INFORMATION.

## Agreement and Plan of Merger

On November 6, 2019, we and one of our direct subsidiaries ("Merger Sub," and together with NV5, the "NV5 Parties"), and Geospatial Holdings Inc. ("Geospatial") and Arlington Capital Partners III, L.P., a Delaware limited partnership, solely in its capacity as representative for the stockholders and optionholders of Geospatial, entered into a definitive Agreement and Plan of Merger (the "Merger Agreement"). The Merger Agreement provides that, at the effective time and upon the terms and subject to the conditions set forth therein and in accordance with applicable law, Merger Sub will merge with and into Geospatial (the "Merger") with Geospatial continuing as the surviving entity after the Merger. The board of directors of NV5 approved the Merger Agreement, the Merger and the other transactions contemplated by the Merger Agreement.

Pursuant to the terms and subject to the conditions set forth in the Merger Agreement, immediately prior to the effective time of the Merger, each issued and outstanding share of Series A preferred stock of Geospatial shall be automatically converted into Geospatial Common Shares (as defined below) and be entitled to the payment of certain accrued or declared but unpaid dividends as provided in Geospatial's current certificate of incorporation.

Immediately thereafter, at the effective time of the Merger, each issued and outstanding share of common stock, par value \$0.01 per share, of

Geospatial ("Geospatial Common Shares") (other than shares as to which the holder shall have exercised appraisal rights in accordance with Delaware law, any such shares being "Dissenting Shares") will be converted into the right to receive a pro rata portion (the "Per Share Merger Consideration") of the aggregate of (i) \$302.5 million in cash, plus (ii) the aggregate exercise price of all in-the-money options (based on the Per Share Merger Consideration) to purchase Geospatial Common Shares as of the closing date (the "Closing Date"), plus (iii) the amount of cash held by Geospatial as of the Closing Date, less (iv) the aggregate indebtedness of Geospatial as of the Closing Date, less (v) transaction expenses of both NV5 and Geospatial (together with certain working capital adjustments as of the Closing Date and escrowed expense amounts, collectively, the "Merger Consideration"); and

Immediately prior to the effective time of the Merger, each vested in-the-money option to purchase Geospatial Common Shares granted under the Geospatial equity incentive plan will be automatically cancelled and replaced with the right to receive at the closing of the Merger the excess, if any, of the Per Share Merger Consideration over the exercise price of each such in-the-money option and all vested out-of-the-money options and unvested options of Geospatial Common Shares granted under the Geospatial equity incentive plan will be cancelled and forfeited for no consideration.

The NV5 Parties and Geospatial have made certain customary representations and warranties in the Merger Agreement and have agreed to customary covenants, including, among others, a representation by NV5 that it will have at the Closing Date sufficient cash available to pay the Merger Consideration and all related fees and expenses in connection with the Merger and certain related transactions, covenants by Geospatial with respect to the conduct of its business during the period between execution of the Merger Agreement and the closing of the Merger, a covenant by NV5 to obtain a customary representation and warranty insurance policy, and covenants by both parties regarding regulatory clearance of the Merger by appropriate anti-trust authorities, including. (in the case of NV5) divesting entities, facilities or assets, terminating, amending or assigning existing licenses or other agreements and entering into new licenses or other agreements as is necessary or reasonably advisable as a condition to obtaining such regulatory clearance.

The completion of the Merger is subject to various customary closing conditions, including, among others: (i) clearance by appropriate anti-trust authorities; (ii) the absence of a material adverse effect on Geospatial between signing and closing of the Merger; and (iii) that no more than 5% of Geospatial's Common Shares shall be Dissenting Shares.

The Merger Agreement may be terminated under certain circumstances, including by either party (i) if the Merger has not been consummated on or before February 6, 2020 (unless the terminating party is in material breach of the Merger Agreement that is the cause of, or results in the failure of the Merger to close), (ii) if a final and non-appealable order, decree or ruling is entered prohibiting or restraining the Merger, or (iii) upon a material uncured breach by the other party that would cause the closing conditions not to be satisfied.

A copy of the Merger Agreement is filed herewith as Exhibit 2.1 and is incorporated herein by reference. The foregoing description of the Merger Agreement does not purport to be complete and is qualified in its entirety by reference to the full text of the Merger Agreement. The Merger Agreement has been filed to provide investors with information regarding its terms. It is not intended to provide any other factual information about the NV5 Parties or Geospatial. In particular, the assertions embodied in the representations and warranties in the Merger Agreement were made as of a specified date, are modified or qualified by information in confidential disclosure letters provided by each party to the other in connection with the signing of the Merger Agreement, may be subject to a contractual standard of materiality different from what might be viewed as material to stockholders, or may have been used for the purpose of allocating risk between the parties. Accordingly, the representations and warranties in the Merger Agreement are not necessarily characterizations of the actual state of facts about the NV5 Parties or Geospatial at the time they were made or otherwise and should only be read in conjunction with the other information that NV5 makes publicly available in reports, statements and other documents filed with the SEC.

#### **Table of Contents**

#### **Commitment Letter**

In connection with the pending Merger, NV5 obtained a financing commitment to provide an incremental loan facility in the original principal amount of \$240 million (the "Incremental Facility") to fund the Merger under its existing credit facility, pursuant to a commitment letter (the "Commitment Letter") from Bank of America, N.A., BofA Securities, Inc., PNC Bank, National Association and PNC Capital Markets LLC. The Incremental Facility includes a \$150 million term loan tranche and a \$90 million increase in NV5's existing \$125 million revolving facility, for a total availability of \$365 million. The funding of the Incremental Facility provided for in the Commitment Letter is contingent on the satisfaction of customary conditions, including but not limited to (i) execution and delivery of definitive documentation with respect to the Incremental Facility in accordance with the terms set forth in the Commitment Letter, and (ii) consummation of the Merger in accordance with the Merger Agreement not earlier than December 20, 2019. The actual documentation governing the Incremental Facility has not been finalized, and accordingly, the actual terms may differ from the description of such terms in the Commitment Letter.

A copy of the Commitment Letter is filed herewith as Exhibit 10.1 and is incorporated herein by reference. The foregoing description of the Commitment Letter

does not purport to be complete and it is qualified in its entirety by reference to the full text of the Commitment Letter.

## ITEM 6. EXHIBITS.

| <u>Number</u> | <u>Description</u>  |
|---------------|---|
| 2.1           | Agreement and Plan of Merger by and among NV5 Global, Inc., NV5 Merger Sub Corp., Geospatial Holdings Inc. and Arlington Capital Partners III, L.P., dated as of November 6, 2019.          |
| 10.1          | Commitment Letter, dated as of November 6, 2019, by and among NV5 Global, Inc., Bank of America, N.A., BofA Securities, Inc., PNC Bank, National Association and PNC Capital Markets LLC.   |
| 31.1          | Certification of Chief Executive Officer pursuant to Rule 13a-14(a) and 15d-14(a) under the Securities Exchange Act of 1934, as adopted pursuant to § 302 of the Sarbanes-Oxley Act of 2002 |
| 31.2          | Certification of Chief Financial Officer pursuant to Rule 13a-14(a) or 15d-14(a) under the Securities Exchange Act of 1934, as adopted pursuant to § 302 of the Sarbanes-Oxley Act of 2002  |
| 32.1          | Certifications of Chief Executive Officer and Chief Financial Officer pursuant to 18 U.S.C. 1350, as adopted pursuant to § 906 of the Sarbanes-Oxley Act of 2002 **                         |
| 101.INS       | XBRL Instance Document  |
| 101.SCH       | XBRL Taxonomy Extension Schema Document   |
| 101.CAL       | XBRL Taxonomy Extension Calculation Linkbase Document   |
| 101.LAB       | XBRL Taxonomy Extension Label Linkbase Document   |
| 101.PRE       | XBRL Taxonomy Extension Presentation Linkbase Document  |
| 101.DEF       | XBRL Taxonomy Extension Definition Linkbase Document  |
|               |   |

- \* Schedules and exhibits have been omitted pursuant to Item 601(a)(5) of Regulation S-K. A copy of any omitted schedule or exhibit will be furnished supplementally to the Securities and Exchange Commission upon request.
- \*\* Furnished herewith. This certification is being furnished solely to accompany this report pursuant to 18 U.S.C. Section 1350, and is not being filed for purposes of Section 18 of the Exchange Act of 1934, as amended, and is not to be incorporated by reference into any filings of the Company, whether made before or after the date hereof, regardless of any general incorporation language in such filing.

# **SIGNATURES**

Pursuant to the requirements of Section 13 or 15(d) of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned, thereunto duly authorized.

# NV5 GLOBAL, INC.

By: <u>/s/ Edward Codispoti</u>
Edward Codispoti
Chief Financial Officer
(Principal Financial and Accounting Officer)

Date: November 7, 2019

# AGREEMENT AND PLAN OF MERGER

**BY AND AMONG** 

**NV5 GLOBAL, INC.** 

**NV5 MERGER SUB CORP.,** 

GEOSPATIAL HOLDINGS INC.,

**AND** 

THE REPRESENTATIVE IDENTIFIED HEREIN

**DATED AS OF NOVEMBER 6, 2019** 

# TABLE OF CONTENTS

<u>Page</u>

| ART   | ICLE I EFFECTS OF THE MERGER   | ; MERGE                      | R CONSID                   | ERATION | N 2          |    |
|---|--|------------------------------|----------------------------|---------|--------------|----|
| 1.1   | Merger 2   |                              |                            |         |              |    |
| 1.2   | Effect of the Merger   | 2                            |                            |         |              |    |
| 1.3   | Certificate of Incorporation; Bylaws   |                              | 2                          |         |              |    |
| 1.4   | Directors and Officers of the Survivir   | ng Compan                    | y                          | 2       |              |    |
| 1.5   | Common Stock of Merger Sub   | -                            | 3                          |         |              |    |
| 1.6   | Dissenting Shares  | 3                            |                            |         |              |    |
| 1.7   | Effect on the Shares and Options   |                              | 3                          |         |              |    |
| 1.8   | Merger Consideration   | 4                            |                            |         |              |    |
| 1.9   | Closing Payments   | 5                            |                            |         |              |    |
| 1.10  | Adjustment of the Merger Considera   | ation                        | 6                          |         |              |    |
| 1.11  | Other Payments   | 10                           |                            |         |              |    |
| 1.12  | Withholding  | 10                           |                            |         |              |    |
| 1.13  | Letters of Transmittal   | 10                           |                            |         |              |    |
| 1.14  | No Liability   | 11                           |                            |         |              |    |
| ADT   | ICLE II CLOSING  | 11                           |                            |         |              |    |
| 2.1   | ICLE II CLOSING Closing 11   | 11                           |                            |         |              |    |
| 2.1   | Effective Time   | 12                           |                            |         |              |    |
| 2.3   | Deliveries by the Company and the R  |                              | ve at Closi                | nα      | 12           |    |
| 2.4   | Deliveries by Parent and Merger Sub  | _                            | ve at Closi                | 13      | 12           |    |
| 2.1   | Deliveries by I dient and Weiger Bub   | ut Closing                   |                            | 13      |              |    |
|   |  | C                            |                            |         |              |    |
| ART   | ICLE III REPRESENTATIONS AND   | ) WARRA                      | NTIES OF                   | THE COM | <b>IPANY</b> | 13 |
| ART   | ICLE III REPRESENTATIONS AND Organization and Power  | WARRA<br>13                  | NTIES OF                   | THE COM | MPANY        | 13 |
|   |  | 13                           | NTIES OF                   | THE COM | ИРАNY        | 13 |
| 3.1   | Organization and Power   | 13                           |                            | THE COM | ИРАNY        | 13 |
| 3.1<br>3.2  | Organization and Power<br>Authorization of Agreement   | 13                           | 13                         | THE COM | ИРАNY        | 13 |
| 3.1<br>3.2<br>3.3   | Organization and Power<br>Authorization of Agreement<br>Conflicts; Consents of Third Parties   | 13                           | 13<br>14                   | ТНЕ СОМ | MPANY        | 13 |
| 3.1<br>3.2<br>3.3<br>3.4  | Organization and Power<br>Authorization of Agreement<br>Conflicts; Consents of Third Parties<br>Capitalization; Operating Subsidiaries   | 13<br>s<br>16                | 13<br>14                   | ТНЕ СОМ | ИPANY        | 13 |
| 3.1<br>3.2<br>3.3<br>3.4<br>3.5   | Organization and Power<br>Authorization of Agreement<br>Conflicts; Consents of Third Parties<br>Capitalization; Operating Subsidiaries<br>Financial Statements   | 13<br>s<br>16                | 13<br>14<br>14             | THE COM | MPANY        | 13 |
| 3.1<br>3.2<br>3.3<br>3.4<br>3.5<br>3.6  | Organization and Power<br>Authorization of Agreement<br>Conflicts; Consents of Third Parties<br>Capitalization; Operating Subsidiaries<br>Financial Statements<br>Undisclosed Liabilities; Indebtedness  | 13<br>s<br>16                | 13<br>14<br>14<br>16       | THE COM | MPANY        | 13 |
| 3.1<br>3.2<br>3.3<br>3.4<br>3.5<br>3.6<br>3.7   | Organization and Power Authorization of Agreement Conflicts; Consents of Third Parties Capitalization; Operating Subsidiaries Financial Statements Undisclosed Liabilities; Indebtedness Absence of Certain Developments   | 13<br>S<br>16                | 13<br>14<br>14<br>16       | ТНЕ СОМ | MPANY        | 13 |
| 3.1<br>3.2<br>3.3<br>3.4<br>3.5<br>3.6<br>3.7<br>3.8  | Organization and Power Authorization of Agreement Conflicts; Consents of Third Parties Capitalization; Operating Subsidiaries Financial Statements Undisclosed Liabilities; Indebtedness Absence of Certain Developments Legal Proceedings   | 13<br>S<br>16                | 13<br>14<br>14<br>16<br>17 | THE COM | MPANY        | 13 |
| 3.1<br>3.2<br>3.3<br>3.4<br>3.5<br>3.6<br>3.7<br>3.8<br>3.9   | Organization and Power Authorization of Agreement Conflicts; Consents of Third Parties Capitalization; Operating Subsidiaries Financial Statements Undisclosed Liabilities; Indebtedness Absence of Certain Developments Legal Proceedings Compliance with Laws; Permits   | 13<br>S<br>16                | 13<br>14<br>14<br>16<br>17 | THE COM | <b>IPANY</b> | 13 |
| 3.1<br>3.2<br>3.3<br>3.4<br>3.5<br>3.6<br>3.7<br>3.8<br>3.9<br>3.10   | Organization and Power Authorization of Agreement Conflicts; Consents of Third Parties Capitalization; Operating Subsidiaries Financial Statements Undisclosed Liabilities; Indebtedness Absence of Certain Developments Legal Proceedings Compliance with Laws; Permits Taxes 19  | 13<br>S<br>16                | 13<br>14<br>14<br>16<br>17 | THE COM | MPANY        | 13 |
| 3.1<br>3.2<br>3.3<br>3.4<br>3.5<br>3.6<br>3.7<br>3.8<br>3.9<br>3.10<br>3.11   | Organization and Power Authorization of Agreement Conflicts; Consents of Third Parties Capitalization; Operating Subsidiaries Financial Statements Undisclosed Liabilities; Indebtedness Absence of Certain Developments Legal Proceedings Compliance with Laws; Permits Taxes 19 Title to Properties.   | 13<br>s 16<br>18             | 13<br>14<br>14<br>16<br>17 | THE COM | MPANY        | 13 |
| 3.1<br>3.2<br>3.3<br>3.4<br>3.5<br>3.6<br>3.7<br>3.8<br>3.9<br>3.10<br>3.11<br>3.12                                 | Organization and Power Authorization of Agreement Conflicts; Consents of Third Parties Capitalization; Operating Subsidiaries Financial Statements Undisclosed Liabilities; Indebtedness Absence of Certain Developments Legal Proceedings Compliance with Laws; Permits Taxes 19 Title to Properties. Environmental Matters.  | 13 S 16 18 21 22             | 13<br>14<br>14<br>16<br>17 | THE COM | <b>IPANY</b> | 13 |
| 3.1<br>3.2<br>3.3<br>3.4<br>3.5<br>3.6<br>3.7<br>3.8<br>3.9<br>3.10<br>3.11<br>3.12<br>3.13                         | Organization and Power Authorization of Agreement Conflicts; Consents of Third Parties Capitalization; Operating Subsidiaries Financial Statements Undisclosed Liabilities; Indebtedness Absence of Certain Developments Legal Proceedings Compliance with Laws; Permits Taxes 19 Title to Properties. Environmental Matters. Material Contracts   | 13 s 16 18 21 22 22          | 13<br>14<br>14<br>16<br>17 | THE COM | MPANY .      | 13 |
| 3.1<br>3.2<br>3.3<br>3.4<br>3.5<br>3.6<br>3.7<br>3.8<br>3.9<br>3.10<br>3.11<br>3.12<br>3.13                         | Organization and Power Authorization of Agreement Conflicts; Consents of Third Parties Capitalization; Operating Subsidiaries Financial Statements Undisclosed Liabilities; Indebtedness Absence of Certain Developments Legal Proceedings Compliance with Laws; Permits Taxes 19 Title to Properties. Environmental Matters. Material Contracts Government Contracts.   | 13 s 16 18 21 22 22 24       | 13<br>14<br>14<br>16<br>17 | THE COM | MPANY        | 13 |
| 3.1<br>3.2<br>3.3<br>3.4<br>3.5<br>3.6<br>3.7<br>3.8<br>3.9<br>3.10<br>3.11<br>3.12<br>3.13<br>3.14<br>3.15         | Organization and Power Authorization of Agreement Conflicts; Consents of Third Parties Capitalization; Operating Subsidiaries Financial Statements Undisclosed Liabilities; Indebtedness Absence of Certain Developments Legal Proceedings Compliance with Laws; Permits Taxes 19 Title to Properties. Environmental Matters. Material Contracts Government Contracts. Personal Property                       | 13 S 16 18 21 22 22 24 25    | 13<br>14<br>14<br>16<br>17 | THE COM | <b>MPANY</b> | 13 |
| 3.1<br>3.2<br>3.3<br>3.4<br>3.5<br>3.6<br>3.7<br>3.8<br>3.9<br>3.10<br>3.11<br>3.12<br>3.13<br>3.14<br>3.15<br>3.16 | Organization and Power Authorization of Agreement Conflicts; Consents of Third Parties Capitalization; Operating Subsidiaries Financial Statements Undisclosed Liabilities; Indebtedness Absence of Certain Developments Legal Proceedings Compliance with Laws; Permits Taxes 19 Title to Properties. Environmental Matters. Material Contracts Government Contracts. Personal Property Intellectual Property | 13 s 16 18 21 22 22 24 25 25 | 13<br>14<br>14<br>16<br>17 | THE COM | MPANY .      | 13 |

i

| 3.19   | Transactions With Related Parties     |            | 3           | 30                  |                    |     |     |
|--------|---------------------------------------|------------|-------------|---------------------|--------------------|-----|-----|
| 3.20   | Insurance                             | 30         |             |                     |                    |     |     |
| 3.21   | Financial Advisors                    | 30         |             |                     |                    |     |     |
| 3.22   | Privacy and Data Security             |            | 30          |                     |                    |     |     |
| 3.23   | Customers                             | 31         |             |                     |                    |     |     |
| 3.24   | Bank Accounts                         | 31         |             |                     |                    |     |     |
| 3.25   | Backlog 32                            | 2          |             |                     |                    |     |     |
| 3.26   | Accounts Receivable and Accounts      | Payable    |             | 32                  |                    |     |     |
| 3.27   | Product and Service Warranties        |            | 33          |                     |                    |     |     |
| 3.28   | LIMITATIONS OF REPRESENTA             | ATIONS     | AND W       | ARRANTIES           | 33                 |     |     |
| 4 D.T. | AGLE WAREPREGENTATIONS AND            | D 1111 D E |             | a of by being an    | ID MED GED GM      |     |     |
|        | ICLE IV REPRESENTATIONS AN            |            |             | ES OF PARENT AN     | ND MERGER SU       | 3   |     |
|        | Organization and Power                | 3.         |             |                     |                    |     |     |
|        | Authorization of Agreement            |            | 34          | 2.4                 |                    |     |     |
|        | Conflicts; Consents of Third Parties  | 2.5        |             | 34                  |                    |     |     |
|        | Legal Proceedings                     | 35         |             |                     |                    |     |     |
|        | Financial Capability                  | 35         |             |                     |                    |     |     |
|        | Solvency 35                           |            |             |                     |                    |     |     |
|        |                                       | 35         |             |                     |                    |     |     |
|        | Financial Advisors                    | 36         | D 11        |                     | ~ 1 ×              |     | 2.6 |
| 4.9    | No Other Representations and Warra    | inties; No | ) Kelian    | ce; Parent and Merg | ger Sub Investigat | on. | 36  |
| ART    | ICLE V COVENANTS                      |            | 37          |                     |                    |     |     |
| 5.1    | Conduct of Business                   | 37         |             |                     |                    |     |     |
| 5.2    | Access to Information.                | 38         |             |                     |                    |     |     |
| 5.3    | Efforts; Regulatory Filings and Cons  | sents.     |             | 39                  |                    |     |     |
| 5.4    | Confidentiality                       | 41         |             |                     |                    |     |     |
| 5.5    | Preservation of Records               | 42         | 2           |                     |                    |     |     |
| 5.6    | Publicity 42                          |            |             |                     |                    |     |     |
| 5.7    | Director and Officer Liability; Inden | nnificatio | n.          | 42                  |                    |     |     |
| 5.8    | R&W Insurance Policy                  | 43         |             |                     |                    |     |     |
| 5.9    | Financing 4                           | 4          |             |                     |                    |     |     |
| 5.10   | Stockholder Approval                  |            | 45          |                     |                    |     |     |
| ADT    | ICLE VI CONDITIONS TO CLOSI           | NC         |             | 45                  |                    |     |     |
|        | Conditions to the Obligations of the  |            | Doron       |                     | 45                 |     |     |
|        | Other Conditions to the Obligations   |            | -           | •                   |                    |     |     |
|        | Other Conditions to the Obligations   |            |             | 46 46               |                    |     |     |
|        | •                                     | or the Co  | шрапу<br>47 |                     |                    |     |     |
| 0.4    | Frustration of Closing Conditions     |            | 4/          |                     |                    |     |     |
| ART    | ICLE VII TERMINATION                  |            | 47          |                     |                    |     |     |
| 7.1    | Termination                           | 47         |             |                     |                    |     |     |
| 7.2    | Effect of Termination                 | 4          | 9           |                     |                    |     |     |
|        |                                       |            |             |                     |                    |     |     |
| ART    | ICLE VIII TAX MATTERS                 |            | 49          |                     |                    |     |     |
|        |                                       |            |             |                     |                    |     |     |

| 8.1 Tax Sharing Arrangements       | 49                               |    |
|------------------------------------|----------------------------------|----|
| 8.2 Cooperation                    | 49                               |    |
| 8.3 Allocation of Taxes            | 49                               |    |
| 8.4 338(g) Elections               | 50                               |    |
| 8.5 Other Tax Matters              | 50                               |    |
| ARTICLE IX MISCELLANEOUS           | 50                               |    |
| 9.1 No Survival; Certain Waivers.  | 50                               |    |
| 9.2 Expenses                       | 52                               |    |
| 9.3 Governing Law                  | 52                               |    |
| 9.4 Submission to Jurisdiction; Wa | aivers 52                        |    |
| 9.5 Further Assurances             | 53                               |    |
| 9.6 Entire Agreement               | 53                               |    |
| 9.7 Amendments and Waivers         | 53                               |    |
| 9.8 Notices                        | 54                               |    |
| 9.9 Severability                   | 55                               |    |
| 9.10 Specific Performance          | 55                               |    |
| 9.11 No Third-Party Beneficiaries  | ; No Recourse Against Affiliates | 56 |
| 9.12 Assignment                    | 56                               |    |
| 9.13 Authorization of Representati | ive 56                           |    |
| 9.14 Attorney Conflict Waiver      | 59                               |    |
| 9.15 No Recourse to Debt Financi   | ng Sources 59                    |    |
| 9.16 Counterparts                  | 60                               |    |
| ARTICLE X DEFINITIONS AND          | INTERPRETATIONS                  | 60 |
| 10.1 Certain Definitions           | 60                               |    |
| 10.2 Certain Interpretive Matters  | 73                               |    |
| •                                  |                                  |    |

# **Exhibits**

Surviving Company Certificate of Incorporation Sample Working Capital and Agreed Principles Form of Letter of Transmittal Exhibit A Exhibit B Exhibit C Exhibit D Pro Rata Share

#### AGREEMENT AND PLAN OF MERGER

THIS AGREEMENT AND PLAN OF MERGER (this "Agreement") is entered into as of November [6], 2019, by and among (i) NV5 Global, Inc., a Delaware corporation ("Parent"), (ii) NV5 Merger Sub Corp., a Delaware corporation and a whollyowned subsidiary of Parent ("Merger Sub"), (iii) Geospatial Holdings Inc., a Delaware corporation (the "Company"), and (iv) Arlington Capital Partners III, L.P., a Delaware limited partnership, solely in its capacity as representative for the Sellers (the "Representative"). Parent, Merger Sub, the Company and, as applicable, the Representative are sometimes referred to in this Agreement as a "Party" and collectively as the "Parties." Capitalized terms that are used in this Agreement and not otherwise defined herein shall have the respective meanings ascribed to such terms in ARTICLE X.

## WITNESETH:

**WHEREAS**, as of the date of this Agreement, the authorized shares of capital stock of the Company consist of, and are designated under the Company Charter as, Series A Preferred Stock, Series B Preferred Stock and Common Stock (each, a "Share" and, collectively, the "Shares");

**WHEREAS**, as of the date of this Agreement, there are no shares of Series B Preferred Stock or Common Stock issued and outstanding, and there are issued and outstanding options to purchase shares of Common Stock under the Equity Incentive Plan (each, an "Option" and, collectively, the "Options" and together with the Shares, the "Securities");

**WHEREAS**, the Parties intend to effectuate a merger (the "<u>Merger</u>") of Merger Sub with and into the Company in accordance with this Agreement and the Delaware General Corporation Law (the "<u>DGCL</u>"), with the Company to be the surviving company of the Merger (the "<u>Surviving Company</u>");

**WHEREAS**, immediately prior to the Effective Time, (i) the Series A Preferred Stock shall be converted into Common Stock and (ii) the Options shall be cancelled for the consideration, and upon the terms and conditions, described herein;

**WHEREAS**, immediately following such conversions and cancellations, the Merger shall be consummated and, at the Effective Time, each share of Common Stock shall be converted into the right to receive the consideration upon the terms and conditions described herein;

**WHEREAS**, the boards of directors of the Company, Parent and Merger Sub have, upon the terms and subject to the conditions set forth herein, (i) unanimously approved this Agreement, the Merger and the other transactions contemplated hereby and (ii) declared that this Agreement, the Merger and the other transactions contemplated hereby are advisable on the terms and conditions set forth herein;

**WHEREAS**, the board of directors of the Company has recommended that the Stockholders adopt and approve this Agreement and the Transactions; and

WHEREAS, Parent, as the sole stockholder of Merger Sub, has adopted and approved this Agreement and the Transactions in accordance with the DGCL.

**NOW**, **THEREFORE**, in consideration of the foregoing and the mutual representations, warranties, covenants and agreements contained herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, and intending to be legally bound hereby, the Parties hereby agree as follows:

#### **ARTICLE I**

# EFFECTS OF THE MERGER; MERGER CONSIDERATION

- 1.1 <u>Merger</u>. At the Effective Time and upon the terms and subject to the conditions of this Agreement and the applicable provisions of the DGCL, Merger Sub shall merge with and into the Company, the separate corporate existence of Merger Sub shall cease and the Company shall continue as the Surviving Company and a wholly-owned Subsidiary of Parent.
- 1.2 <u>Effect of the Merger</u>. At the Effective Time, the effect of the Merger shall be as provided in the applicable provisions of the DGCL. Without limiting the generality of the foregoing, and subject thereto, at the Effective Time by virtue of the Merger and without any action on the part of Merger Sub or the Company, all of the property, rights, privileges, powers and franchises of the Company and Merger Sub shall vest in the Surviving Company, and all debts, liabilities and duties of the Company and Merger Sub shall become the debts, liabilities and duties of the Surviving Company.
- 1.3 <u>Certificate of Incorporation; Bylaws.</u> At the Effective Time, by virtue of the Merger and without any action on the part of Merger Sub or the Company, (a) the certificate of incorporation of the Surviving Company shall be amended and restated in the form of <u>Exhibit A</u> and, as so amended, shall be the certificate of incorporation of the Surviving Company until thereafter amended in accordance with the terms thereof or as provided by applicable Law, and (b) the bylaws of the Merger Sub as in effect immediately prior to the Effective Time shall be the bylaws of the Surviving Company until thereafter amended in accordance with the terms thereof, the certificate of incorporation of the Surviving Company or as provided by applicable Law; provided that, in each case, the name of the corporation set forth therein shall be changed to the name of the Company.
- 1.4 <u>Directors and Officers of the Surviving Company</u>. The directors of Merger Sub immediately prior to the Effective Time shall, from and after the Effective Time, be the directors of the Surviving Company until their successors have been duly elected or appointed and qualified or until their earlier death, resignation or removal in accordance with the certificate of incorporation and bylaws of the Surviving Company. The officers of the Company immediately prior to the Effective Time (excluding the Persons who have resigned from such officer positions effective as of the Effective Time pursuant to <u>Section 2.3(g)</u>; <u>provided</u>, <u>however</u>, for the avoidance of doubt, the resignations contemplated by <u>Section 2.3(g)</u> shall be limited solely to the applicable person's capacity as an officer and/or director of the Group Companies and shall not be deemed to include a resignation of such person as an employee of the Group Companies) shall be the initial officers of the Surviving Company until their successors have been duly elected or appointed and qualified

or until their earlier death, resignation or removal in accordance with the certificate of incorporation and by-laws of the Surviving Company.

- 1.5 <u>Common Stock of Merger Sub</u>. At the Effective Time, by virtue of the Merger and without any action on the part of Merger Sub, the Company or Parent, each share of common stock of Merger Sub issued and outstanding immediately prior to the Effective Time shall each be converted into and exchanged for 1 share of common stock of the Surviving Company.
- Dissenting Shares. Notwithstanding anything in this Agreement to the contrary, Shares issued and outstanding 1.6 immediately prior to the Adjustment Time held by a Stockholder who has not voted in favor of the Merger or consented thereto in writing or executed an enforceable waiver of appraisal rights to the extent permitted by applicable Law, and who has properly exercised its appraisal rights with respect to such shares (the "Dissenting Shares") in accordance with Section 262 of the DGCL (the "Dissenters' Rights Statute") shall not be converted into a right to receive the applicable portion of the Merger Consideration, but shall instead have the rights set forth in the Dissenters' Rights Statute unless such Stockholder subsequently withdraws its demand for appraisal or waives, fails to perfect or otherwise loses such Stockholder's appraisal rights, if any (in which case such Shares shall thereupon be deemed to have been converted into, and to have become exchangeable for, as of the Effective Time, the right to receive the applicable portion of the Merger Consideration). At the Effective Time, Stockholders holding Dissenting Shares shall cease to have any rights with respect thereto, except the rights provided under applicable Law and as provided in this Section 1.6. If, after the Effective Time, such Stockholder fails to perfect or loses any such right to appraisal, each such Dissenting Share of such Stockholder shall be treated as a Share that had been converted as of the Effective Time into the right to receive the applicable portion of the Merger Consideration, without interest, in accordance with Section 1.7. The Company shall give Parent (i) prompt written notice of any demands for the Company to purchase the Stockholder's Shares for cash received by the Company and withdrawals or attempted withdrawals of any such demands, and (ii) after the Closing, the right to direct and control all negotiations and proceedings with respect to any such demands. Prior to the Effective Time, the Company shall not, without the prior written consent of the Parent, make any payment with respect to, or settle, offer to settle or otherwise negotiate, any such demands.
- 1.7 <u>Effect on the Shares and Options</u>. Upon the terms and subject to the conditions of this Agreement, by virtue of the Merger and without any action on the part of the Company, Parent, Merger Sub or the Sellers, as the case may be:
- (a) <u>Series A Preferred Stock</u>. Each share of Series A Preferred Stock issued and outstanding immediately prior to the Effective Time (other than Dissenting Shares) shall automatically be converted into Common Stock immediately prior to the Effective Time in accordance with the terms of the Company Charter and the Company Stockholder Agreement and, subject to the payment of the Series A Preferred Amount, shall be treated as outstanding Common Stock for purposes of this Agreement.
- (b) <u>Common Stock</u>. At the Effective Time, each share of Common Stock issued and outstanding immediately prior to the Effective Time (including each share of Common Stock into which a share of Series A Preferred Stock is converted pursuant to <u>Section 1.7(a)</u>, but excluding

Dissenting Shares) shall be converted automatically into the right to receive the Per Participating Share Merger Consideration (as reduced at the Closing by the Per Participating Share Portion of such share of Common Stock of each of the Escrow Amount and the Representative Expense Fund Amount, and subject to adjustment as provided in Section 1.10) in cash, payable in accordance with and subject to the conditions provided in this ARTICLE I.

- (c) Options. Immediately prior to the Effective Time, each vested Option shall automatically be canceled and replaced with the right to receive at the Closing an amount in cash equal to (i) the excess, if any, of (x) the Per Participating Share Merger Consideration (as reduced at the Closing by the Per Participating Share Portion of such Option of each of the Escrow Amount and the Representative Expense Fund Amount, and subject to adjustment as provided in Section 1.10) over (y) the exercise price per share of Common Stock subject to such Option, multiplied by (ii) the total number of shares of Common Stock subject to such vested Option immediately prior to its cancellation (such payment to be net of withholdings, if any, and without interest); provided that in the event the Per Participating Share Merger Consideration is less than the exercise price per share of Common Stock subject to such vested Option, then such vested Option shall be automatically canceled as of the Effective Time without payment of consideration in exchange therefor. As of the Effective Time, each Option that is not vested shall be cancelled and forfeited for no consideration.
- 1.8 Merger Consideration. The aggregate consideration in respect of the Common Stock (including each share of Common Stock into which Series A Preferred Stock is converted pursuant to Section 1.7(a)), for the payment of the Series A Preferred Amount and for the cancellation of the Options shall be an amount equal to (a) Three Hundred and Two Million Five Hundred Thousand Dollars (\$302,500,000) (the "Base Price"), plus (b) the amount, if any, by which the Working Capital exceeds the Working Capital Target, minus (c) the amount, if any, by which the Working Capital Target exceeds the Working Capital, plus (d) the Closing Cash, if any, minus (e) Closing Date Indebtedness, minus (f) Transaction Expenses, plus (g) the aggregate exercise price for all vested Options to be canceled and replaced at the Closing pursuant to Section 1.7(c), (such resulting amount pursuant to clauses (a)-(g), and as such amount may be adjusted pursuant to the provisions of Section 1.10, the "Merger Consideration"). The Merger Consideration shall be allocated among the Sellers as specified in the Pre-Closing Statement delivered pursuant to Section 1.10 below. The Parties acknowledge and agree that neither Parent nor Merger Sub shall have any liability to any Person relating to, or obligation to verify, the allocation of the Merger Consideration, including the Series A Preferred Amount, among the Sellers as set forth in the Pre-Closing Statement (including with respect to the determination of the Per Participating Share Merger Consideration, the Per Participating Share Portion and the Series A Preferred Amount), and upon payment of the amounts set forth in Section 1.9 in accordance with the Pre-Closing Statement, Parent and Merger Sub will have satisfied all of their respective obligations under this Agreement with respect thereto, subject to any adjustments to the Merger Consideration pursuant to Section 1.10 hereof.
- 1.9 <u>Closing Payments</u>. At the Closing, Parent shall make or cause to be made, by wire transfer of immediately available funds, the following payments (each such payment, a "<u>Closing Payment</u>"):

- (a) payment on behalf of the Group Companies, to the payees thereof, of an aggregate cash amount equal to the amount of all Closing Date Indebtedness of the type identified in item (i) of the definition of "Indebtedness" and all Transaction Expenses;
- (b) payment to the Escrow Agent of a cash amount equal to the Escrow Amount to the bank account designated in writing by the Escrow Agent, which shall be deposited into an escrow account (the "Escrow Account") established pursuant to the Escrow Agreement;
- (c) payment to the Representative of a cash amount equal to \$1,000,000 (such amount, the "Representative Expense Fund Amount") to a bank account designated in writing by the Representative;
- (d) payment to the Company (on behalf of the Optionholders) of a cash amount equal to the aggregate amount payable to the Optionholders in accordance with Section 1.7(c) and as set forth in the Pre-Closing Statement to the bank account designated in writing by the Representative, which amount shall, within five (5) days following receipt, be paid by the Company through the Company's payroll system to the applicable Optionholders (subject to applicable withholding Tax), allocated to each Optionholder in accordance with the Pre-Closing Statement; and
- (e) payment to the account designated by a paying agent selected by the Representative with Parent's prior approval (such approval not to be unreasonably withheld) (the "Paying Agent") of a cash amount equal to the aggregate amount of the Series A Preferred Amount and the Estimated Merger Consideration payable to the Stockholders in accordance with Section 1.7(a) and Section 1.7(b) and as set forth in the Pre-Closing Statement to be distributed by the Paying Agent to the Stockholders in accordance with this ARTICLE I; provided, that if any Stockholder has not delivered to the Representative and Parent a duly executed and completed Letter of Transmittal and Surrendered Certificate(s) prior to the Closing Date, the amount allocated with respect to such Stockholder will be paid to the Paying Agent on behalf of such Stockholder (and distributed thereto upon delivery of such executed and completed Letter of Transmittal and Surrendered Certificate(s)).

Each of the Closing Payments shall be made in the amounts and as set forth in the Pre-Closing Statement delivered pursuant to Section 1.10 below.

For the avoidance of any doubt, Parent shall not be obligated to make payment under this <u>Section 1.9</u>, which, in the aggregate, exceeds the Base Price, as adjusted pursuant to <u>Sections 1.8</u> and <u>1.10</u>.

# 1.10 Adjustment of the Merger Consideration.

(a) <u>Pre-Closing Statement</u>. No later than three (3) days prior to the Closing Date, the Company shall deliver to Parent a statement (the "<u>Pre-Closing Statement</u>") setting forth the Company's good faith estimate of the Merger Consideration as of the Adjustment Time (the "<u>Estimated Merger Consideration</u>"), together with reasonably detailed calculations demonstrating each component thereof, as well as the amount of each Closing Payment. The Pre-Closing Statement shall be prepared in a manner consistent with the definitions of the terms Working Capital, Closing

Cash, Closing Date Indebtedness, Transaction Expenses, and the Accounting Rules and practices referred to therein (including as reflected on Exhibit B).

- (b) <u>Final Merger Consideration Adjustment</u>. The Merger Consideration shall be adjusted following the Closing based on the difference between the Final Closing Date Merger Consideration (as determined in accordance with this <u>Section 1.10</u>) and the Estimated Merger Consideration, if any, and payment shall be made in respect of any such post-Closing adjustment as set forth in <u>Section 1.10(e)</u>.
- (c) Closing Statement. No later than ninety (90) days after the Closing Date, Parent shall cause to be prepared in good faith and delivered to the Representative a statement (the "Closing Statement") setting forth Parent's calculation of the Merger Consideration as of the Adjustment Time ("Closing Date Merger Consideration"). The Closing Statement shall be prepared in a manner consistent with the definitions of the terms Working Capital, Closing Cash, Closing Date Indebtedness, Transaction Expenses, and the Accounting Rules and practices referred to therein (including as reflected on Exhibit B). The Parties agree that the purpose of preparing the Closing Statement and determining the Working Capital, Closing Cash, Closing Date Indebtedness, and Transaction Expenses is to measure the amount of the Working Capital, Closing Cash, Closing Date Indebtedness, and Transaction Expenses and such processes are not intended to permit the introduction of different or new judgments, accounting methods, policies, principles, practices, procedures, classifications or estimation methodologies for the purpose of preparing the Closing Statement or determining the Working Capital, Closing Cash, Closing Date Indebtedness, or Transaction Expenses. The Closing Statement shall entirely disregard (i) any and all effects on the assets or liabilities of the Group Companies as a result of the Transaction or of any financing or refinancing arrangements entered into at any time by Parent or its Affiliates or any other transaction entered into by Parent or its Affiliates in connection with the consummation of the Transaction, and (ii) any of the plans, transactions, or changes which Parent intends to initiate or make or cause to be initiated or made after the Closing with respect to any Group Company or their respective businesses or assets, or any facts or circumstances that are unique or particular to Parent or its Affiliates or any of their assets or liabilities. For the avoidance of doubt, unless the Representative otherwise agrees in writing, Parent may not amend, adjust, supplement or modify the Closing Statement or the amount of Closing Date Merger Consideration following its delivery to the Representative. If Parent fails to deliver the Closing Statement within such ninety (90)-day period, then in addition to any other rights the Representative may have under this Agreement, the Representative shall have the right to elect that the Estimated Merger Consideration be deemed to be the amount of the Closing Date Merger Consideration and be final and binding and used for purposes of calculating the adjustment pursuant to Section 1.10(e). The Parties acknowledge that no adjustments may be made to the Working Capital Target.

# (d) Disputes.

(i) The Representative shall have thirty (30) days to review the Closing Statement; <u>provided</u> that if Parent does not promptly provide access or other information as specified in <u>Section 1.10(f)</u> (and in any event within five (5) Business Days of any request by the Representative, then the Representative shall have thirty (30) days plus the number

of days between the date of the Representative's request for such access or information and the date Parent grants or provides such access or information. If the Representative disagrees with Parent's calculation of the Closing Date Merger Consideration as set forth in the Closing Statement, the Representative may, within thirty (30) days after receipt of the Closing Statement, deliver a written notice to Parent (a "<u>Dispute Notice</u>") disagreeing with such calculation and, to the extent reasonably able to so specify, setting forth the Representative's basis for such disagreement (the "<u>Disputed Items</u>"). If the Representative fails to deliver a Dispute Notice during such thirty (30)-day period, the Representative shall have waived its rights to contest the Closing Statement and the calculations of the Closing Date Merger Consideration set forth therein shall be deemed to be final and binding upon the Parties, and such amount shall be used as the Final Closing Date Merger Consideration for purposes of calculating the adjustment pursuant to <u>Section 1.10(e</u>).

(ii) If a Dispute Notice is duly delivered pursuant to Section 1.10(d)(i), the Representative and Parent shall, during the fifteen (15) days following such delivery, attempt in good faith to reach an agreement on all or a portion of the Disputed Items. If Parent and the Representative reach an agreement on any Disputed Item during such period, the resolution of such Disputed Items shall be in writing and shall be final and binding upon the Parties. If, during such fifteen (15)-day period, the Representative and Parent are unable to reach an agreement on all of the Disputed Items, then all Disputed Items remaining in dispute following such fifteen (15)-day period shall be submitted by the Representative and Parent to the Accounting Referee (the "Referred Disputed Items") as promptly as reasonably practicable for a determination resolving such Referred Disputed Items (it being agreed and understood that the Accounting Referee shall act as an arbitrator to determine the Referred Disputed Items (and, as a result thereof, the Closing Date Merger Consideration) and shall do so based solely on presentations and information provided by Parent and the Representative, as further specified below, and not by independent review), provided that the Parties can respond to questions or requests for information from the Accounting Referee. In conducting its review, the decision of the Accounting Referee shall be solely based on (A) the definitions and other applicable provisions of this Agreement, (B) a single presentation by each of the Representative and Parent limited to the Referred Disputed Items (which presentations the Accounting Referee shall be instructed to forward to Parent and the Representative, as applicable) and (C) one (1) written response submitted to the Accounting Referee by each of the Representative and Parent within ten (10) Business Days after receipt of each such presentation (which responses the Accounting Referee shall be instructed to forward to Parent and the Representative, as applicable), and not on independent review. The scope of the disputes to be resolved by the Accounting Referee shall be limited to resolving the Referred Disputed Items, and, in connection therewith, fixing mathematical errors and determining whether the Referred Disputed Items were determined in accordance with this Agreement (including the definition of the terms Working Capital, Closing Cash, Closing Date Indebtedness, Transaction Expenses, and the Accounting Rules) and the Accounting Referee is not to make any other determination. The Accounting Referee shall deliver to the Representative and Parent, as promptly as practicable (but in any case no later than thirty (30) days from the date of engagement of the Accounting Referee), a report setting forth its calculations of the Closing Date Merger Consideration based solely upon (x) the

Accounting Referee's final determination of the Referred Disputed Items and (v) the items which were finally determined pursuant to pursuant to Sections 1.10(d)(i) and 1.10(d)(ii) and not submitted to the Accounting Referee for resolution, which such Closing Date Merger Consideration amount shall not be less than the applicable amount thereof shown in Parent's calculation delivered pursuant to Section 1.10(c) nor more than the amount thereof shown in the Representative's calculation delivered pursuant to Section 1.10(d)(i). Such report shall be final and binding upon the Parties, absent manifest error, and shall be used for purposes of calculating the adjustment pursuant to Section 1.10(b) above. Notwithstanding anything herein to the contrary, the dispute resolution mechanism contained in this Section 1.10(d) shall be the exclusive mechanism for resolving disputes regarding the Merger Consideration adjustment. Judgment may be entered upon the determination of the Accounting Referee in any court having jurisdiction over the Party against which such determination is to be enforced. The fees, costs and expenses of the Accounting Referee shall be borne by the Parties in proportion to the relative amount each Party's determination has been modified, with any amount borne by the Representative to be paid out of the Representative Expense Fund Amount. For example, if the Representative challenges the calculation of the Closing Date Merger Consideration by an amount of \$100,000, but the Accounting Referee determines that the Representative has a valid claim for only \$60,000, the Representative (on behalf of the Sellers) shall bear 40% of the fees and expenses of the Accounting Referee and Parent shall bear the other 60% of such fees and expenses of the Accounting Referee. The Representative and Parent shall, and Parent shall cause the Group Companies, and each of its and their representatives to, reasonably cooperate and assist in any review by the Accounting Referee of the Closing Statement and the calculations of the Closing Date Merger Consideration.

- (e) <u>Final Closing Date Merger Consideration</u>. Following the time that the Closing Date Merger Consideration is finally determined pursuant to this <u>Section 1.10</u> (such finally determined amount, the "<u>Final Closing Date Merger Consideration</u>"), payment shall be made as follows:
  - (i) If the Final Closing Date Merger Consideration is greater than or equal to the Estimated Merger Consideration (the amount of such excess, the "Increase Amount"), then within five (5) Business Days after the Final Closing Date Merger Consideration is finally determined pursuant to this Section 1.10, (x) Parent shall pay by wire transfer of immediately available funds an amount equal to the Increase Amount up to \$500,000 to the Paying Agent (for further distribution to the Stockholders who delivered to the Representative and Parent a duly completed and executed Letter of Transmittal and Surrendered Certificate(s) prior to such time) and the Surviving Company (for further distribution to the Optionholders) in accordance with Section 1.11, and (y) Parent and the Representative shall deliver joint written instructions to the Escrow Agent instructing the Escrow Agent to disburse the Escrow Funds to the Paying Agent (for further distribution to the Stockholders who delivered to the Representative and Parent a duly completed and executed Letter of Transmittal and Surrendered Certificate(s) prior to such time) and the Surviving Company (for further distribution to the Optionholders) in accordance with Section 1.11.

- (ii) If the Final Closing Date Merger Consideration is less than the Estimated Merger Consideration, then within five (5) Business Days after the Final Closing Date Merger Consideration is finally determined pursuant to this Section 1.10, Parent and the Representative shall deliver joint written instructions to the Escrow Agent instructing the Escrow Agent to (x) disburse to an account designated in writing by Parent all or a portion of the Escrow Funds equal to such deficiency (it being understood that, notwithstanding anything to the contrary contained herein, the Escrow Funds shall be the sole source of recovery for any payment required to be made to Parent pursuant to this Section 1.10(e)(ii)),and (y) disburse in accordance with Section 1.11 to the Paying Agent (for further distribution to the Stockholders who delivered to the Representative and Parent a duly completed and executed Letter of Transmittal and Surrendered Certificate(s) prior to such time) and the Surviving Company (for further distribution to the Optionholders) in accordance with Section 1.11 the balance of the Escrow Funds, if any, remaining after payment to Parent pursuant to the preceding clause (x).
- (iii) Notwithstanding anything to the contrary in this <u>Section 1.10(e)</u>, if, at the time that Parent is required to make any payments to the Paying Agent (on behalf of the Stockholders) pursuant to this <u>Section 1.10(e)</u>, any Stockholder has not delivered to the Representative and Parent a duly completed and executed Letter of Transmittal and Surrendered Certificate(s) prior to such time, then the amount allocated with respect to such Stockholder will be paid to the Paying Agent on behalf of such Stockholder (and distributed thereto upon delivery of such executed and completed Letter of Transmittal and Surrendered Certificate(s)).

Upon payment of the amounts provided in this <u>Section 1.10(e)</u> in accordance herewith, none of the Parties may make or assert any claim under this <u>Section 1.10</u>.

- (f) <u>Cooperation</u>. During the period of time from and after the Closing Date through the final determination and payment of the Final Closing Date Merger Consideration in accordance with this <u>Section 1.10</u>, Parent shall afford, and shall cause the Group Companies to afford, to the Representative and any reasonably necessary accountants, counsel or financial or other advisers retained by the Representative in connection with the review of the Closing Date Merger Consideration, and afford to the Accounting Referee in connection with any review by it in accordance with <u>Section 1.10(d)(ii)</u>, reasonable access during normal business hours upon reasonable advance written notice to all the books, records, information and the Chief Financial Officer of the Company, in each case, to the extent relevant to the review or preparation of the Closing Statement and the calculation of the Closing Date Merger Consideration. The determination of the Closing Date Merger Consideration shall not take into account any developments or events taking place after the Closing Date or any actions taken by Parent on its own behalf or on behalf of the Company on or following the Closing Date.
  - 1.11 Other Payments. In order to facilitate the payment of any Other Seller Payments pursuant to this Agreement:
- (a) the Stockholders' aggregate Pro Rata Share of such funds shall be paid to an account designated by the Paying Agent (on behalf of the Stockholders) for distribution to the

Stockholders entitled thereto on a pro rata basis (based on each Stockholder's Pro Rata Share), subject to such Stockholders delivering to the Representative and Parent a duly completed and executed Letter of Transmittal and Surrendered Certificate(s) prior to such time; and

(b) the Optionholders' aggregate Pro Rata Share of such funds shall be paid to the Surviving Company for further distribution to the Optionholders on a pro rata basis (based on each Optionholder's Pro Rata Share), subject to applicable withholding Tax, through the Surviving Company's payroll system within five (5) days following receipt.

The Parties acknowledge and agree Parent shall have no liability to any Person relating to, or obligation to verify, the allocation of any Other Seller Payment among the Sellers as set forth in this <u>Section 1.11</u>, and upon payment of any Other Seller Payment in accordance with this <u>Section 1.11</u> or as directed by the Representative, Parent will have satisfied all of its obligations under this Agreement with respect thereto.

1.12 Withholding. Parent, the Company and Escrow Agent shall be entitled to deduct and withhold from the amounts payable pursuant to this Agreement such amounts as it reasonably determines it is required to deduct and withhold with respect to the making of such payments under any provision of U.S. federal, state, local or foreign Tax Law and instead shall pay such amount to the applicable Taxing Authority. Before making any such deduction or withholding described in the previous sentence, except for (a) any withholding due to the failure of the Company to provide the certificate described in Section 2.3(i), and (b) any withholding on payments under any compensatory payments made in connection with the transactions contemplated by this Agreement, Parent shall give the Representative notice of the intention to make such deduction or withholding, and such notice, which shall include the authority, basis and method of calculation for the proposed deduction or withholding, at least three (3) Business Days before such deduction or withholding is required, in order for the Representative to obtain reduction of or relief from such deduction or withholding. Amounts so withheld by Parent or other withholding agent and paid to the applicable Taxing Authority shall be treated for all purposes of this Agreement as having been paid to the recipient in respect of which such deduction and withholding was made by Parent or other withholding agent.

## 1.13 Letters of Transmittal.

(a) Promptly following the date hereof, the Company shall deliver to each Stockholder a Letter of Transmittal in the form of Exhibit C hereto, with such changes as mutually negotiated by the Parties in good faith to reflect the addition of a customary release provision and a customary indemnity provision with respect to the representations, warranties and covenants contained in such Letter of Transmittal (a "Letter of Transmittal"). Promptly following the receipt by the Paying Agent of the Closing Payment contemplated by Section 1.9(e), the Paying Agent shall deliver to each Stockholder who delivered to the Representative and Parent a duly completed and executed Letter of Transmittal and Surrendered Certificates(s) at least two (2) Business Days prior to the Closing Date the Series A Preferred Amount and a portion of the Estimated Merger Consideration allocated to each such Stockholder in accordance with the Pre-Closing Statement and to the bank account designated in such Stockholder's Letter of Transmittal. Following the Closing, but subject to Section 1.13(c), upon delivery by a Stockholder that did not receive such

portion of the Closing Payment contemplated by <u>Section 1.9(e)</u> at the Closing pursuant to the immediately preceding sentence to the Representative and Parent of a duly completed and executed Letter of Transmittal and Surrendered Certificate(s), the Paying Agent shall pay to such Stockholder within five (5) Business Days after such delivery, (x) the amounts to which such Stockholder is entitled to pursuant to the immediately preceding sentence and (y) if the Final Closing Date Merger Consideration has been finally determined as of such time, the amounts to which such Stockholder is entitled to pursuant to <u>Section 1.11(a)</u>, in each case, by wire transfer of immediately available funds to the account designated by such Stockholder in such Stockholder's Letter of Transmittal. No interest or dividends will be paid or accrued on the consideration payable to any Stockholder hereunder. Until surrendered in accordance with the provisions of this <u>Section 1.13</u>, the Shares shall represent, for all purposes, only the right to receive an amount in cash equal to the applicable portion of the Merger Consideration payable in respect thereto pursuant to this Agreement.

- (b) In the event that any stock certificate representing the Shares has been lost, stolen or destroyed, upon the making of a customary affidavit (in such form as attached to the Letter of Transmittal) of that fact by the Stockholder claiming such certificate to be lost, stolen or destroyed, the Paying Agent or the Surviving Company, as applicable, will pay, in exchange for the Shares represented by such lost, stolen or destroyed certificate, the consideration to which such Stockholder would otherwise be entitled pursuant to Section 1.7 without any requirement to post any bond or other security.
- (c) Promptly following the date that is one (1) year after the Closing Date, Parent may instruct the Paying Agent to deliver to the Surviving Company all cash delivered to the Paying Agent pursuant to this Agreement that is still in its possession at such time, in which case the Paying Agent's duties shall terminate. Thereafter, each Stockholder may deliver a duly completed and executed Letter of Transmittal and Surrendered Certificates(s) to the Surviving Company and (subject to applicable abandoned property, escheat and similar Laws) receive in consideration therefor, and the Surviving Company shall promptly pay, the portion of the Merger Consideration deliverable to such Stockholder in respect of its Surrendered Certificates(s) as determined in accordance with this <u>ARTICLE I</u> without any interest thereon.
- 1.14 <u>No Liability</u>. Notwithstanding anything to the contrary in this <u>ARTICLE I</u>, none of the Company, Parent, the Representative or the Surviving Company shall be liable to any Person for any amount properly paid in good faith to a public official pursuant to any abandoned property, escheat or similar law.

#### ARTICLE II

# **CLOSING**

2.1 <u>Closing</u>. Subject to the terms and conditions of this Agreement, the consummation of the Merger and the other Transactions (the "<u>Closing</u>") shall take place by electronic means of communication to avoid the necessity of a physical Closing, provided that if a physical Closing is required by either Party, it shall occur at 10:00 a.m., Eastern Time, at the offices of Sheppard Mullin Richter and Hampton LLP, 2099 Pennsylvania Ave., NW, Suite 100, Washington, DC 20006, on a date to be mutually specified by Parent and the Representative, which shall be no later than the

second (2<sup>nd</sup>) Business Day after the satisfaction or waiver (by the applicable Party hereto in writing) of the conditions set forth in <u>ARTICLE VI</u> (not including conditions which are to be satisfied by actions taken at the Closing but subject to the satisfaction or waiver (by the applicable Party hereto in writing) of those conditions at the Closing) (the date on which the Closing actually occurs, the "<u>Closing Date</u>"); <u>provided</u>, <u>however</u>, that without prior written consent of Parent, in no event shall the Closing occur prior to December 6, 2019.

- 2.2 <u>Effective Time</u>. Subject to the provisions of this Agreement, on the Closing Date, the Parties shall cause to be filed a certificate of merger, executed in accordance with, and in such form as is required by, the relevant provisions of the DGCL with respect to the Merger (the "<u>Certificate of Merger</u>") with the Secretary of State of the state of Delaware. The Merger shall become effective at 11:59 p.m. on the Closing Date (the time the Merger becomes effective being hereinafter referred to as the "<u>Effective Time</u>").
- 2.3 <u>Deliveries by the Company and the Representative at Closing</u>. At or prior to the Closing, the Company and/or the Representative, as applicable, shall deliver, or cause to be delivered, to Parent the following:
  - (a) the Pre-Closing Statement in accordance with Section 1.10(a);
  - (b) the Certificate of Merger, duly executed by the Company;
- (c) payoff letters in a commercially reasonable form with respect to the Closing Date Indebtedness to be repaid at Closing in accordance with <u>Section 1.9(a)</u>, which payoff letters shall provide for the release of all Liens relating to such Closing Date Indebtedness following satisfaction of the terms contained in such payoff letters;
  - (d) the Escrow Agreement, duly executed by the Representative on behalf of the Sellers;
- (e) an agreement with the Paying Agent (in a form reasonably satisfactory to each of Parent and the Representative, the "Paying Agent Agreement"), duly executed by the Representative on behalf of the Stockholders and the Paying Agent;
  - (f) a certificate required to be delivered pursuant to <u>Section 6.2(f)</u>;
- (g) evidence of resignation or removal, effective as of the Closing, of those directors and officers of any Group Company designated by Parent to the Company in writing at least five (5) Business Days prior to the Closing Date;
- (h) evidence of the termination of the Management Services Agreement, except with respect to the provisions related to limitation of liability for third party claims; and
- (i) a certificate conforming to the requirements of Treasury Regulation Section 1.1445-2(c)(3) to the effect that the Company is not, and has not been during the relevant period specified in Section 897(c)(1)(A)(ii) of the Code, a "United States real property holding corporation" within the meaning of Section 897(c) of the Code together with the appropriate notice to the Internal

Revenue Service pursuant to Treasury Regulation Section 1.897-2(h), which Representative shall file with the Internal Revenue Service immediately after the Closing.

- 2.4 <u>Deliveries by Parent and Merger Sub at Closing</u>. At or prior to the Closing, Parent or Merger Sub, as applicable, shall deliver, or cause to be delivered, the following:
- (a) to the Representative, the Paying Agent, the Company and all other payees referenced in <u>Section 1.9</u>, the Closing Payments;
  - (b) to the Representative, a copy of the R&W Insurance Policy;
  - (c) to the Representative, a certificate required to be delivered pursuant to Section 6.3(d); and
  - (d) to the Representative, the Escrow Agreement, duly executed by Parent and the Escrow Agent.

### ARTICLE III

#### REPRESENTATIONS AND WARRANTIES OF THE COMPANY

Except as set forth on the disclosure schedule delivered by the Company to Parent and Merger Sub on the date hereof (the "<u>Company Disclosure Schedule</u>"), which are subject to the limitations and qualifications set forth in <u>Section 10.2(c)</u>, the Company hereby represents and warrants to Parent and Merger Sub as of the date hereof as follows:

- 3.1 Organization and Power. Each of the Group Companies is a corporation or other legal entity duly formed, validly existing and in good standing under the Laws of its respective jurisdiction of formation as set forth on Section 3.1 of the Company Disclosure Schedule. Each of the Group Companies is duly licensed or qualified to do business in each jurisdiction in which the nature of its business or the character or location of any properties or assets owned or leased by it makes such licensing or qualification necessary, except for those jurisdictions where the failure to be so licensed or qualified would not have a Company Material Adverse Effect. The Company has the requisite power and authority to execute and deliver this Agreement and each other Transaction Agreement to which it is a party, to perform its obligations hereunder and thereunder and to consummate the Transactions. Each of the Group Companies has the requisite power and authority to own, lease and operate all of its properties and assets and to carry on its business as it is now being conducted in all material respects.
- 3.2 <u>Authorization of Agreement</u>. The execution and delivery of the Transaction Agreements to which the Company is a party, the performance by the Company of its obligations thereunder and the consummation of the Transactions have been duly authorized by the requisite corporate action on the part of the Company. Each of the Transaction Agreements to which the Company is a party has been duly and validly executed and delivered by the Company and (assuming the due authorization, execution and delivery by the other parties thereto) each such Transaction Agreement, when so executed and delivered, will constitute, the legal, valid and binding obligations

of the Company, enforceable against it in accordance with its terms, subject to applicable Equitable Principles.

## 3.3 Conflicts; Consents of Third Parties.

- (a) Except as listed on Section 3.3(a) of the Company Disclosure Schedule, and assuming all Governmental Approvals set forth on Section 3.3(b) of the Company Disclosure Schedule have been obtained and are effective and all applicable waiting periods have expired or been terminated and all filings and notifications described in Section 3.3(b) of the Company Disclosure Schedule have been made, none of the execution, delivery and performance by the Company of the Transaction Agreements to which it is a party or the consummation of the Transactions by the Company will contravene, conflict with, violate or constitute a default (with or without notice or lapse of time, or both) under, give rise to a right of termination, acceleration, modification or cancellation under, or otherwise require the consent or waiver of, notice or declaration to, or filing with any Person, including any Governmental Authority, pursuant to, any provision of (A) the Organizational Documents of any Group Company; (B) any Material Contract, Material Government Contract or Permit to which any Group Company is a party to or bound by, or by which any Group Company's properties or assets are bound; or (C) any Law applicable to any Group Company, except, in the case of clauses (B) and (C), where such conflict, violation or default, consent, waiver, notice, declaration or filing would not have, individually or in the aggregate, a Company Material Adverse Effect.
- (b) Except as set forth on Section 3.3(b) of the Company Disclosure Schedule, no consent, waiver, approval, waiting period expiration or termination, Order, Permit or authorization of, or declaration or filing with, or notification to, any Governmental Authority (a "Governmental Approval") is required on the part of any Group Company in connection with the execution and delivery by the Company of the Transaction Agreements to which it is a party or the consummation of the Transactions by the Company, except those Governmental Approvals, which, if not obtained, would not have, individually or in the aggregate, a Company Material Adverse Effect.

# 3.4 <u>Capitalization; Operating Subsidiaries</u>.

(a) The authorized capital stock of the Company consists of 350,000 shares, consisting of (a) 200,000 shares of Common Stock, and (b) 150,000 shares of Preferred Stock, of which 100,000 shares are designated as Series A Preferred Stock and 10,000 shares are designated as Series B Preferred Stock. As of the date hereof, there are zero (0) shares of Common Stock issued and outstanding, 46,788.643 shares of Series A Preferred Stock issued and outstanding, and zero (0) shares of Series B Preferred Stock issued and outstanding. Section 3.4(a)(i) of the Company Disclosure Schedule sets forth the number and classes of issued and outstanding Securities as of the date of this Agreement, the names of the holders thereof and the number and class of Securities held by each such holder. Each of the Securities have been duly and validly authorized and issued, are fully paid and non-assessable, and were not issued in violation of the preemptive rights of any Person or any Law or Order by which any Group Company at the time of issuance was bound. Except as set forth on Section 3.4(a)(ii) of the Company Disclosure Schedule, there are no outstanding (i) equity interests or voting securities of the Company, (ii) securities convertible or

exchangeable into or exchangeable or exercisable for any Securities or other equity interests or voting securities of the Company, (iii) options, warrants or rights (including purchase rights, subscription rights, preemptive rights, conversion rights, exchange rights, calls, puts or rights of first refusal) or other Contracts that require the Company to issue, sell or otherwise cause to become outstanding or to acquire, subscribe for, purchase, repurchase or redeem Securities or other equity interests of the Company, (iv) stock appreciation, phantom stock, profit participation or similar rights with respect to the Company, (v) shares of capital stock of the Company held as treasury shares, or (vi) dividends which have accrued or been declared but are currently unpaid on the capital stock of the Company (except for the Series A Preferred Amount, which will be paid out of the Merger Consideration and except for any dividends or distributions payable solely in cash prior to the Closing). The Securities set forth on Section 3.4(a)(i) of the Company Disclosure Schedule constitute all of the outstanding equity securities or securities convertible into or exchangeable for equity securities of the Company.

- Section 3.4(b) of the Company Disclosure Schedule sets forth a true and correct list of all direct and indirect Subsidiaries of the Company (the "Operating Subsidiaries"), listing for each such Operating Subsidiary (i) its name, (ii) its jurisdiction of organization, (iii) the number and type of its issued and outstanding equity interests and (iv) the record owners of such equity interests. The Company has no Subsidiaries or ownership interest in any other entity other than the Operating Subsidiaries. Except for the Operating Subsidiaries or as set forth on Section 3.4(b) of the Company Disclosure Schedule, no Group Company owns, or holds the right to acquire, any stock, partnership interest, joint venture interest or other equity ownership interest in any other Person. All of the issued and outstanding equity interests of each of the Operating Subsidiaries have been duly and validly authorized and issued, are fully paid and non-assessable, and are owned (either directly or indirectly) by the Company or one of the Operating Subsidiaries, free and clear of any Liens (other than Permitted Liens and limitations imposed by their Organizational Documents or any applicable securities Laws). Except as otherwise set forth in Section 3.4(b) of the Company Disclosure Schedule, there are no outstanding (v) shares of capital stock of any Group Company held as treasury shares, (w) equity interests or voting securities of any Operating Subsidiary, (x) securities convertible or exchangeable into equity interests of any Operating Subsidiary, (y) options, warrants or rights (including purchase rights, subscription rights, preemptive rights, conversion rights, exchange rights. calls, puts or rights of first refusal) or other Contracts that require any Operating Subsidiary to issue, sell or otherwise cause to become outstanding or to acquire, subscribe for, purchase, repurchase or redeem any equity interests of such Operating Subsidiary or (z) stock appreciation, phantom stock, profit participation or similar rights with respect to any Operating Subsidiary.
- (c) Except for this Agreement and as set forth on <u>Section 3.4(c)</u> of the Company Disclosure Schedule, the Company is not a party to any agreement restricting the transfer of, relating to the voting of, requiring registration of, or granting any preemptive rights, anti-dilution rights, rights of first refusal or any similar rights with respect to any securities of the Company.

## 3.5 Financial Statements.

- (a) The Company has made available to Parent the following financial statements (collectively, the "Company Financial Statements"):
  - (i) the audited consolidated balance sheets of the Group Companies as of December 30, 2017 and December 29, 2018, and the related consolidated statements of operations, consolidated statements of changes in stockholders' equity and consolidated statements of cash flows of the Group Companies for the fiscal years then ended; and
  - (ii) an unaudited consolidated balance sheet of the Group Companies as of September 28, 2019, and the related unaudited statements of operations and cash flows of the Group Companies for the 9-month period then ended.

September 28, 2019 shall be referred to herein as the "Balance Sheet Date", the balance sheet of the Group Companies as of such date shall be referred to herein as the "Balance Sheet".

- (b) The Company Financial Statements have been prepared in accordance with GAAP and present fairly, in all material respects, the consolidated financial condition and results of operations of the Group Companies as of the dates and for the periods indicated therein except, in each case, as disclosed therein or as set forth on Section 3.5(b) of the Company Disclosure Schedule, and, in the case of the unaudited Company Financial Statements, (i) that such Company Financial Statements may be subject to normal year-end adjustments and (ii) for the absence of notes thereto throughout the periods covered thereby. The books and records of the Group Companies fully and fairly reflect, in each case in all material respects, all of the Group Companies' transactions, properties, assets and liabilities and have been maintained in good faith and, except as reflected on the Financial Statements, the balance sheets included in the Financial Statements do not reflect any write-up or revaluation increasing the book value of any assets.
- (c) The Company has established and maintains systems of internal accounting controls over financial reporting that are designed to provide reasonable assurance in all material respects that (i) transactions are executed in accordance with management's general or specific authority, (ii) transactions are recorded and disclosed as necessary to permit preparation of financial statements in accordance with GAAP, (iii) access to the Group Companies' property and assets is permitted only in accordance with management's general or specific authority; and (iv) appropriate action is taken to address any difference between the recorded accountability for items compared with the actual levels. To the Knowledge of the Company, in the past three (3) years, there has not occurred any fraud involving the Company's internal accounting control.

# 3.6 <u>Undisclosed Liabilities</u>; <u>Indebtedness</u>.

(a) Except as set forth on Section 3.6(a) of the Company Disclosure Schedule, no Group Company has any liabilities that would have been required to be reflected on the Balance Sheet or in the notes thereto in accordance with GAAP and were not so reflected, other than (i) as disclosed in, set forth on, or reflected or reserved against in the Company Financial Statements, (ii) those incurred in the Ordinary Course of Business since the Balance Sheet Date, (iii) those included

in the calculation of the Closing Payments, (iv) those that are repaid, terminated, forgiven, settled, cancelled or otherwise extinguished at Closing pursuant to the terms of this Agreement, (v) those incurred in connection with the Transactions or (vi) those that would not exceed \$1,000,000 in the aggregate.

- (b) <u>Section 3.6(b)</u> of the Company Disclosure Schedule sets forth a true and complete list as of the date of this Agreement of all Indebtedness of the Group Companies.
- 3.7 <u>Absence of Certain Developments</u>. Except as set forth on <u>Section 3.7</u> of the Company Disclosure Schedule, between the Balance Sheet Date and the date hereof, the business of the Group Companies has been conducted in all material respects in the Ordinary Course of Business and there has not been any:
  - (a) Company Material Adverse Effect;
- (b) material change in any method of accounting or accounting practice of the Company, except as required by GAAP or as disclosed in the notes to the Financial Statements;
  - (c) entry into any Contract that would constitute a Material Contract;
- (d) transfer, assignment, sale or other disposition of any of the material assets shown or reflected in the Balance Sheet or cancellation of any debts or entitlements, other than in the Ordinary Course of Business;
- (e) transfer or assignment of or a grant of any license or sublicense under or with respect to any material Company Intellectual Property or Company IP Agreements, except (i) in the Ordinary Course of Business, (ii) pursuant to licenses and related support or maintenance agreements for commercial software, (iii) pursuant to any Contract in which the any such transfer, assignment or grant is incidental and not the principal purpose of such Contract;
- (f) acceleration, termination or cancellation of any Material Contract to which the Company is a party or by which it is bound;
- (g) grant of any bonuses, whether monetary or otherwise, or increase in any wages, salary, severance, pension or other compensation or benefits in respect of its current or former employees, officers, directors, independent contractors or consultants, other than in the Ordinary Course of Business or as provided for in any written agreements or required by applicable Law;
- (h) adoption, material modification or termination of any: (i) employment, severance, retention or other agreement with any current or former employee, officer, director, independent contractor or consultant, (ii) Benefit Plan or (iii) collective bargaining or other agreement with a Union, in each case written, other than in the Ordinary Course of Business;
- (i) acquisition by merger or consolidation with, or by purchase of a substantial portion of the assets or stock of, or by any other manner, any business or any Person or any division thereof; or

- (j) any Contract to do any of the foregoing, or any action or omission that would result in any of the foregoing
- 3.8 <u>Legal Proceedings</u>. Except as set forth on <u>Section 3.8</u> of the Company Disclosure Schedule, as of the date hereof, there are no pending or, to the Knowledge of the Company, threatened, Legal Proceedings against any Group Company. As of the date hereof, there is no pending Order imposed upon any of the Group Companies. None of the Group Companies has any suit, litigation, arbitration, claim, charge, grievance, action or proceeding pending against any Governmental Authority or other Person.

# 3.9 Compliance with Laws; Permits.

- (a) Except as set forth on Section 3.9(a) of the Company Disclosure Schedule, each Group Company is, and since January 1, 2017 has been, in material compliance with all Laws applicable to them and their respective businesses, operations and other assets, except for such instances of noncompliance that would not, individually or in the aggregate, result in a material liability to the Group Companies, taken as a whole. Between the Balance Sheet Date and the date hereof, no Group Company has received any written notice of, or been formally charged by a Governmental Authority with, the violation of, or failure to comply with, any Laws.
- (b) Except as set forth on Section 3.9(b) of the Company Disclosure Schedule, the Group Companies have obtained all Permits that are required or are necessary for the operation of their respective businesses as presently conducted, each of which are valid and binding and in full force and effect, other than any such Permits that, if not held by the Group Companies, would not, individually or in the aggregate, result in a Company Material Adverse Effect. As of the date hereof, no Legal Proceeding is pending or, to the Knowledge of the Company, threatened concerning any actual, alleged or potential violation of, failure to comply with, revocation, withdrawal, suspension, cancellation, termination, modification or limitation of any of the Permits.
- (c) Without limiting the generality of the foregoing, the Group Companies are in compliance in all material respects and have, during all periods for which any applicable statute of limitations has not expired, complied in all material respects with, the applicable provisions of the U.S. Foreign Corrupt Practices Act of 1977, and any other applicable non-U.S. anti-corruption laws and regulations; applicable Laws related to the imposition of economic sanctions or embargoes by the U.S. government, including all regulations, laws and policies administered by the U.S. Department of Treasury, Office of Foreign Assets Control; and applicable U.S. export controls, including the Export Administration Regulations administered by the U.S. Department of Commerce, Bureau of Industry and Security.
- (d) The Company Charter, bylaws and other Organizational Documents and similar Organizational Documents for each Group Company have been made available to Parent and are in full force and effect and none of the Group Companies is in material violation of any of the provisions of such Organizational Documents.
- (e) None of the representations and warranties contained in this <u>Section 3.9</u> shall be deemed to relate to tax matters (which are governed by <u>Section 3.10</u>), environmental matters

(which are governed by <u>Section 3.12</u>), government contract matters (which are governed by <u>Section 3.14</u>), or employee benefits matters (which are governed by <u>Section 3.17</u>).

# 3.10 <u>Taxes</u>.

- (a) Each of the Group Companies has prepared and timely filed, or caused to be prepared and timely filed, with the appropriate Taxing Authorities, all income and other material Tax Returns required to be filed with respect to the Group Companies, taking into account any extensions of time to file, and such Tax Returns are true, correct and complete in all material respects.
- (b) Each of the Group Companies has timely paid, or caused to be timely paid, all income and other material Taxes (whether or not shown as due and payable on such Tax Returns) with respect to the Group Companies. Each Group Company has complied with all applicable Laws relating to the reporting, collection, withholding and payment of all Taxes and has duly and timely withheld or collected, paid over and reported all Taxes required to have been withheld or collected by such Group Company.
- (c) No material deficiencies for any Taxes have been proposed, asserted or assessed in writing against any Group Company that are still pending.
- (d) No extensions of the period for assessment or collection of any Taxes are in effect with respect to any Group Company (other than by virtue of extensions of time to file Tax Returns obtained in the Ordinary Course of Business as set forth on Section 3.10(d) of the Company Disclosure Schedules).
- (e) No Tax Return filed by the Group Companies is under current Legal Proceeding by or with any Taxing Authority.
- (f) No claim has been made in writing by any Taxing Authority in a jurisdiction where any of the Group Companies does not file Tax Returns with respect to a particular Tax that such Group Company is or may be subject to taxation in such jurisdiction that has not been resolved.
- (g) There are no Liens for Taxes upon any of the assets or properties of any Group Company, except for Permitted Liens for Taxes not yet delinquent.
- (h) No Group Company has participated in any reportable transaction within the meaning of Code Section 6707A(c) (1) or Treasury Regulations Section 1.6011-4(b) (or any predecessor thereof).
- (i) Each of the Group Companies has provided to Parent true, correct and complete copies of all income and other material Tax Returns filed by or with respect to each member of the Group Companies for taxable periods beginning after December 31, 2015.
- (j) The unpaid Taxes not yet due and payable owed by or with respect to, as the case may be, each Group Company: (i) did not, as of the Balance Sheet Date, exceed the reserve for Tax liability (rather than any reserve for deferred Taxes established to reflect timing differences between book and Tax income), and (ii) do not exceed that reserve as adjusted for the passage of

time through the Closing Date in accordance with the past custom and practice of any Group Company in filing its Tax Returns. Since the Balance Sheet Date, no Group Company has incurred any liability for Taxes outside the Ordinary Course of Business and all Taxes not yet due and payable for a Pre-Closing Tax Period have been accrued and adequately disclosed and fully provided for in accordance with GAAP on the Financial Statements provided to Parent. Any unpaid taxes not yet due and payable that are owed by or with respect to a Group Company are properly reflected in the Closing Statement.

- (k) No Group Company will be required to include any item of income in, or exclude any item of deduction from, taxable income for any taxable period (or portion thereof) beginning after the Closing Date as a result of (i) any change in method of accounting for a Pre-Closing Tax Period; (ii) any closing agreement described in Section 7121 of the Code (or similar provision of state, local or foreign Law); (iii) any installment sale or open sale transaction disposition made in a Pre-Closing Tax Period; (iv) any prepaid amount received in a Pre-Closing Tax Period; (v) any election by the Company under Section 108(i) of the Code; (vi) any intercompany transaction or excess loss account described in Treasury regulations under Section 1502 of the Code (or any corresponding or similar provision of state, local, or foreign income Tax Law); or (vii) any other action taken outside of the Ordinary Course of Business for the purpose of deferring a Tax from a Pre-Closing Tax Period to a period following the Closing Date.
- (l) Each Group Company has the U.S. federal income tax classification set forth on <u>Schedule 3.10(l)</u> of the Company Disclosure Schedule.
- (m) There is no outstanding power of attorney authorizing anyone to act on behalf of any Group Company in connection with any Tax, Tax Return or audit, examination or other proceeding relating to any Tax or Tax Return of such Group Company.
- (n) There is no outstanding closing agreement, ruling request, request to consent to a change in method of accounting, subpoena or request for information with or by any Taxing Authority with respect to any Group Company, its income, gain, loss, deductions, credits, assets, liabilities or business, or any Tax or Tax Return.
- (o) No Group Company is required to include any adjustment under Code Section 481 (or similar provision of state, local or foreign Law) in income for any period ending after the Balance Sheet Date.
- (p) No Group Company is or has ever been a party to a Tax allocation agreement, Tax sharing agreement, or a Tax indemnity agreement (other than a commercial Contract entered into in the Ordinary Course of Business that does not primarily relate to Taxes).
- (q) No Group Company has any "tax-exempt bond financed property" or "tax-exempt use property" within the meaning of Code Section 168(g) or (h), respectively.
- (r) No Group Company is a "distributing corporation" or a "controlled corporation" under Code Section 355 in any distribution within the last two years or pursuant to a plan or series of related transactions (within the meaning of Code Section 355) with the Transactions.

- (s) No Group Company is or has ever been a "personal holding company", or a shareholder in a "foreign personal holding company" or a "passive foreign investment company" (all as defined in the Code). No Company has or has ever had a fixed place of business or permanent establishment outside the United States.
- (t) No Group Company has been a "United States real property holding corporation" (within the meaning of Code Section 897(c)(2)) at any time during the applicable period specified in Code Section 897(c)(1)(A)(ii).
- (u) No "section 197 intangible" (within the meaning of Code Section 197) of any Group Company is subject to the limitations under Code Section 197(f)(9).

# 3.11 <u>Title to Properties</u>.

- (a) <u>Section 3.11(a)</u> of the Company Disclosure Schedules set forth a list of all real property owned by any Group Company (the "<u>Owned Real Property</u>"). Except as set forth on <u>Section 3.11(a)</u>, of the Company Disclosure Schedule, with respect to each parcel of Owned Real Property, the Group Company identified as the owner on <u>Section 3.11(a)</u> of the Company Disclosure Schedule (i) holds fee simple title to any such Owned Real Property, free and clear of all Liens other than Permitted Liens; and (ii) has not granted to any Person the right to use or occupy such Owned Real Property or any material portion thereof.
- (b) <u>Section 3.11(b)</u> of the Company Disclosure Schedule sets forth a list of all real property leased, or subleased to, or otherwise used or occupied by any Group Company (the "<u>Leased Real Property</u>", together with the Owned Real Property, the "<u>Real Property</u>") pursuant to leases, subleases and occupancy agreements thereof (individually, a "<u>Real Property Lease</u>").
  - (c) Except as disclosed on <u>Section 3.11(c)</u> of the Company Disclosure Schedule:
  - (i) each Real Property Lease is a legal, valid and binding obligation of the Group Company party thereto (except (i) as enforceability may be limited by applicable Equitable Principles or (ii) where the failure to be legal, valid, or binding would not, individually or in the aggregate, result in a Company Material Adverse Effect) and, assuming the due authorization and execution by any other party thereto, is in full force and effect;
  - (ii) no Group Company (A) is in material default under any Real Property Lease, or (B) has any Knowledge of any current default by any other party to any Real Property Lease; and
  - (iii) the applicable Group Company has good and valid leasehold interests in each Real Property Lease, free and clear of all Liens created by the applicable Group Company other than Permitted Liens.

## 3.12 Environmental Matters.

(a) Except as set forth on Section 3.12(a) of the Company Disclosure Schedule:

- (i) the Group Companies are in compliance with all Environmental Laws applicable to them or their respective businesses or operations, except where the failure to be in compliance would not, individually or in the aggregate, result in a Company Material Adverse Effect.
- (ii) (x) the Group Companies maintain all Permits that are required under Environmental Laws for the operation of their respective businesses as presently conducted (collectively, the "Environmental Permits") and (y) the Group Companies are not in default or violation of any term, condition or provision of any Environmental Permit, except, in the case of clauses (x) and (y), as would not, individually or in the aggregate, result in a Company Material Adverse Effect.
- (iii) Since January 1, 2017, the Group Companies have not received any written notice of a Legal Proceeding or Order alleging that any of the Group Companies are in material violation of or have any material liability for cleanup or remediation of Hazardous Materials under any Environmental Law.
- (b) This <u>Section 3.12</u> sets forth the sole and exclusive representations and warranties of the Group Companies under this Agreement with respect to Environmental Permits, Environmental Laws, Hazardous Materials, or other environmental matters.

## 3.13 Material Contracts.

- (a) Excluding Government Contracts, <u>Section 3.13(a)</u> of the Company Disclosure Schedule sets forth a list of all of the following Contracts as of the date of this Agreement (other than (A) any such Contract solely by or between the Group Companies, (B) purchase or sale orders entered into in the Ordinary Course of Business or (C) confidentiality or non-disclosure Contracts entered into in the Ordinary Course of Business) to which any Group Company is a party or by which it is bound (collectively, the "<u>Material Contracts</u>"):
  - (i) Contracts with each current officer or director, or current employee of a Group Company who receives annual compensation (excluding commissions and bonuses) in excess of \$200,000;
  - (ii) Contracts entered into since the Balance Sheet Date relating to the acquisition by a Group Company of any operating business, or the equity interests of any other Person;
  - (iii) Contracts with any independent contractor or consultant who is a natural person or sole proprietorship providing compensation in excess of \$150,000 on an annual basis to which a Group Company is a party;
    - (iv) Contracts for or relating to the making of any loans or other extension of credit to any Person;

- (v) Contracts under which a Group Company directly or indirectly indemnifies any Person with respect to infringements of proprietary rights, except if such obligation is incidental to the purpose of the Contract;
- (vi) Contracts that are expected to involve (A) payment to a Group Company or (B) payment by a Group Company, in either case, of more than \$1,000,000 in the aggregate for any individual Contract during the fiscal year ending December 28, 2019 that are not terminable by such Group Company without penalty on 90 days' or less notice;
- (vii) Contract under which any Group Company is a lessee or lessor of any tangible property (other than real property), except for any such Contract under which the aggregate annual rental payments do not exceed \$100,000;
- (viii) Contracts containing covenants of a Group Company prohibiting or materially limiting the right of any of the Group Companies to compete in any line of business or prohibiting or restricting their ability to conduct business with any Person in any geographic area;
- (ix) Contracts for material joint ventures, partnerships, or other similar revenue-sharing arrangements with any Person (other than a Group Company);
- (x) Contracts relating to the incurrence, assumption or guarantee of any Indebtedness or imposing a Lien (other than Permitted Lien) on any of the assets of the Company or any Operating Subsidiary, including indentures, guarantees, loan or credit agreements (except for (x) those being terminated or cancelled in connection with the Closing and (y) security agreements ancillary to any Lease of personal property with respect to the property so Leased);
- (xi) Contracts pursuant to which a Group Company has, since the Balance Sheet Date, sold, exchanged or otherwise disposed of any assets in excess of \$500,000 in the aggregate, other than sales of inventory in the Ordinary Course of Business and property sold or otherwise disposed of in the Ordinary Course of Business and except for any tangible asset which is obsolete; and
- (xii) Contracts identified as a distributor agreement (or substantially similar agreement) involving amounts in excess of \$500,000 in the aggregate during the fiscal year ending December 28, 2019.
- (b) Except as set forth on Section 3.13(b) of the Company Disclosure Schedule, each Material Contract is in full force and effect and is a legal, valid, and binding obligation of the Group Company party thereto and, to the Knowledge of the Company, the other party or parties thereto, except (i) as enforceability may be limited by applicable Equitable Principles or (ii) where the failure to be legal, valid binding or enforceable would not, individually or in the aggregate, result in a Company Material Adverse Effect. True and correct copies of all Material Contracts have been made available to Parent or its representatives. Except as set forth on Section 3.13(b) of the Company Disclosure Schedule, (x) none of the Group Companies nor, to the Knowledge of the

Company, any other party thereto, is in material breach of or material default under any Material Contract and (y) since the Balance Sheet Date, no Group Company has received any written claim or notice of breach of or default under any Material Contract.

# 3.14 Government Contracts.

- (a) Section 3.14(a) of the Company Disclosure sets forth a list of each Government Contract that is currently in effect having expected annual revenue for the fiscal year ending December 28, 2019 in excess of \$3,000,000 and the name of the customer, name of the contract and number, effective date, and period of performance (each, a "Material Government Contract" and, collectively, the "Material Government Contracts").
- (b) Section 3.14(b)(i) of the Company Disclosure Schedule sets forth a true and complete list of each Government Bid (including the current status thereof) for an opportunity in excess of \$3,000,000 which the Company has participated during the fiscal year ending December 28, 2019. Except as set forth on Section 3.14(b)(ii) of the Company Disclosure Schedule, since January 1, 2018, with respect to each Material Government Contract or material Government Bid, (i) to the Knowledge of the Company, the Group Companies have complied in all material respects with all material terms and conditions thereof; (ii) no written notice has been received by any Group Company asserting that the Company, any of its Subsidiaries or any director, officer or employee of the Company or any of its Subsidiaries, is in material breach or violation of any Law or contractual requirement (other than pursuant to routine audits conducted pursuant to such Law or contractual requirement); and (iii) no written notice of termination, cure notice or show-cause notice has been received by any Group Company.
- (c) Except as set forth on Section 3.14(c) of the Company Disclosure Schedule, since January 1, 2018, to the Knowledge of the Company, no Governmental Authority nor any prime contractor, subcontractor or vendor has asserted in writing any claim or initiated any dispute proceeding against any Group Company relating to Material Government Contracts or material Government Bids, nor is the Company or any of its Subsidiaries asserting in writing any claim or initiating any dispute proceeding directly or indirectly against any such party concerning any Material Government Contract or material Government Bid.
- (d) Neither (i) the Company, its Subsidiaries nor any of its or their shareholders, members, officers, or directors, nor (ii) to the Knowledge of the Company, any of its or their employees is debarred, suspended, deemed non-responsible or otherwise excluded from participation in the award of any Material Government Contract or for any reason listed on the List of Parties Excluded from Federal Procurement and Nonprocurement Programs nor, to the Knowledge of the Company, is there any pending debarment, suspension or exclusion proceeding that has been initiated against the Company or any of its Subsidiaries or any of its or their predecessors, shareholders, members, officers, directors, managers or employees.
- (e) The applicable Group Company has complied in all material respects with all Government Contracts and Government Bids listed on Sections 3.14(a) and (b) of the Company Disclosure Schedule.

- (f) There has been no finding of fraud or (to the Company's Knowledge) any material liability as a result of defective pricing, labor mischarging or improper payments on the part of any Group Company in connection with any Government Contracts or Government Bids.
- (g) This <u>Section 3.14</u> sets forth the sole and exclusive representations and warranties of the Group Companies under this Agreement with respect to government contracts matters.
- 3.15 <u>Personal Property</u>. Except as set forth in <u>Section 3.15</u> of the Company Disclosure Schedule, each of the Group Companies has good and valid title to (or a valid leasehold interest in) the material tangible personal property, assets and equipment currently used in the conduct of the business of such Group Company, and such title or leasehold interests are free and clear of all Liens (other than Permitted Liens). All items of tangible personal property, assets and equipment that are used in the operation of the business of the Group Companies are in satisfactory operating condition and repair (ordinary wear and tear excepted).

# 3.16 Intellectual Property.

- (a) Except as set forth on <u>Section 3.16(a)</u> of the Company Disclosure Schedule, all Intellectual Property that is used in or necessary to the conduct of business of the Group Companies as currently conducted, including Company Technology and Company Software ("<u>Company Intellectual Property</u>") is either (i) owned by a Group Company (such Intellectual Property, "<u>Owned Intellectual Property</u>"), or (ii) licensed for use by a Group Company, except, in each case, where a failure to so own or license such Intellectual Property would not, individually or in the aggregate, result in any material liability to the Group Companies.
- (b) Section 3.16(b)(i) of the Company Disclosure Schedule sets forth a list of all active or pending (i) patents, patent applications, trademark registrations and applications, copyright registrations and applications, domain names, and other forms of registrations and applications for registration or issuance with a Governmental Authority of any Owned Intellectual Property, and the jurisdiction(s) in which any of the foregoing has been issued or registered or in which any application for such issuance and registration has been filed or (ii) any material Intellectual Property that is not registered (and in the case of material trade secrets, a general description only) but that is necessary for the operation of the business as conducted during the twelve (12) month period prior to the date of this Agreement (collectively, the "Scheduled IP"). To the Knowledge of the Company, the Scheduled IP is subsisting and in full force and effect. The applicable Group Company possesses all rights, title, and interests in and to the Scheduled IP, free and clear of any Lien other than Permitted Liens and licenses granted in the Outbound IP Contractual Obligations. Except as set forth on Section 3.16(b)(ii) of the Company Disclosure Schedule, there is no, and during the past six (6) years there has been no, pending claim against any of the Group Companies by any Person contesting the ownership or use by any of the Group Companies of any Intellectual Property and none of the Group Companies has received any charge, complaint, claim, demand or notice alleging infringement, violation or misappropriation of the Intellectual Property of any Person. None of the Group Companies is subject to any Claim, proceeding or outstanding Order, settlement agreement or stipulation restricting in any manner the use, transfer or licensing by any

of the Group Companies of any Owned Intellectual Property, or adversely affecting the validity, use or enforceability of any such Company Intellectual Property.

- (c) Except as set forth on Section 3.16(c) of the Company Disclosure Schedule: (i) the Group Companies have not in the past six (6) years infringed, violated or misappropriated any Intellectual Property of any third party; (ii) no Person has infringed, violated or misappropriated, and, to the Knowledge of the Company, no Person is infringing, violating or misappropriating, any Owned Intellectual Property, which has not had a material adverse effect on any Group Company; and (iii) no Person has infringed, violated or misappropriated, and no Person is infringing, violating or misappropriating, any Company Intellectual Property that is not Owned Intellectual Property, and which is exclusively licensed to any Group Company or for which any Group Company has the right to bring suit for infringement, misappropriation or other violation of such rights, which has not had a material adverse effect on any Group Company.
- (d) To the Knowledge of the Company, the Merger and the Transactions (A) have not caused, and will not cause, the forfeiture or termination of, or give rise to a right of forfeiture or termination of any material Company Intellectual Property, or impair the right of any Group Company to use, possess, sell or license any material Company Intellectual Property or portion thereof, in the manner any Group Company uses, possesses, sells or licenses such Company Intellectual Property as of the date of this Agreement, (B) have not caused, and will not cause, any Group Company to be in breach of any Contract governing any material Company Intellectual Property, and (C) have not resulted in, and will not result in, the modification, cancellation, termination, suspension of, or acceleration of, any payments with respect to any Contract governing any Company Intellectual Property, or give any Person the right to do any of the foregoing.
- (e) <u>Section 3.16(e)(i)</u> of the Company Disclosure Schedule separately lists and identifies all Software that is owned by the Group Companies (the "<u>Company Software</u>"). Except as set forth on <u>Section 3.16(e)(ii)</u> of the Company Disclosure Schedule, all right, title and interest in and to the Company Software is owned by a Group Company free and clear of all Liens except for Permitted Liens. The Company Software performs in all material respects in accordance with the documentation and other written materials related thereto and, to the Knowledge of the Company, is free from any disabling codes or instructions and any "back door," "time bomb," "Trojan horse," "worm," "drop dead device," "virus" or other software routines or hardware components that permit unauthorized access or the unauthorized disruption, impairment, disablement or erasure of such Company Software. Except as set forth on <u>Section 3.16(e)(iii)</u> of the Company Disclosure Schedule, the Group Companies have not disclosed, delivered or licensed to any Person, or obligated themselves to disclose, deliver or license to any Person (including any escrow agent), any software source code owned by the Group Companies.
- (f) <u>Section 3.16(f)</u> of the Company Disclosure Schedule sets forth a complete list of all agreements under which (i) any Intellectual Property is licensed, sold, assigned, or otherwise conveyed or provided to a Group Company except for non-exclusive shrink-wrap or click-wrap licenses and other non-exclusive licenses for Commercial-Off-The-Shelf Software, and (ii) a Group Company has granted to any Person any right, title or interest in any material Company Intellectual Property ("<u>Outbound IP Contractual Obligations</u>"), other than any non-disclosure

agreements entered into in the Ordinary Course of Business and non-exclusive licenses to customers in the Ordinary Course of Business. All such agreements are in full force and effect, and are binding obligations of the Group Company party thereto and, to the Knowledge of the Company, the other party or parties thereto. No Group Company that is a party to such agreement is in default under any such agreement, and to the Knowledge of the Company, no other party or parties to any such agreement is in default thereunder.

- (g) The Group Companies have taken commercially reasonable efforts to maintain the secrecy of their trade secrets and other confidential information. To the Knowledge of the Company, no trade secret of the Group Companies as presently conducted has been authorized to be disclosed or has been actually disclosed by any Group Company other than pursuant to a binding and enforceable nondisclosure agreement or other obligation of confidentiality restricting the disclosure and use of such trade secrets.
- (h) Except as set forth on Section 3.16(h) of the Company Disclosure Schedule, each Person who has participated in the development of any material Owned Intellectual Property has conveyed to a Group Company, by written agreement or operation of law, ownership of all such Intellectual Property created or developed by such Person within the scope of such Person's employment or engagement. No such Person has any claim, right or interest to or in any Owned Intellectual Property, excluding rights reserved by contractors of the Group Company in any Intellectual Property owned by such Persons prior to the date of their engagement with any Group Company.
- (i) The Group Companies have not modified or distributed with the Company Software any Software that is provided under a license commonly referred to as an open source, free software, copyleft or community source code license (including any library or code licensed under the GNU General Public License, GNU Lesser General Public License, Apache Software License, or any other public source code license arrangement) in any manner that: (a) prohibits or restricts their ability to charge a royalty or receive consideration in connection with the sublicensing or distribution of any Company Software; (b) requires the distribution or making available of source code of any Company Software; (c) except as specifically permitted by applicable law, grants any other Person the right to decompile, disassemble or otherwise reverse-engineer any Company Software; or (d) requires the licensing of any Company Software for the purpose of making derivative works
- (j) Except as set forth on <u>Section 3.16(j)</u> of the Company Disclosure Schedule, a Group Company has entered into written confidentiality agreements with each Person who has participated in the development of any material Owned Intellectual Property comprising a trade secret.

# 3.17 Employee Benefit Plans.

(a) Section 3.17(a) of the Company Disclosure Schedule sets forth a list of each "employee benefit plan" within the meaning of Section 3(3) of ERISA, and all other material pension, retirement, profit-sharing, deferred compensation, equity or equity-based compensation, health and welfare, severance pay, vacation, bonus, incentive compensation and fringe benefit plans (excluding

government programs and statutory benefits), and individual employment or award agreements, that are currently adopted, maintained by, sponsored or contributed to by the Group Companies or under which any Group Company has any outstanding liability, (each a "Benefit Plan," and collectively, the "Benefit Plans").

- (b) With respect to each Benefit Plan, the Company has provided or made available to Parent the following to the extent applicable: (i) the plan document, adoption agreement (or, in the case of an unwritten Benefit Plan, a written description thereof) and all amendments thereto; (ii) the most recent summary plan description (and any summaries of material modifications with respect thereto); (iii) the two (2) most recent annual reports on Form 5500 (with schedules and attachments); (iv) the most recent IRS opinion or determination letter, (v) where applicable, any trust agreement, insurance policies and other documents establishing other funding arrangement and (vi) any correspondence since January 1, 2017 with any Governmental Authority relating to any government investigation or audit or any submission under any voluntary compliance procedure.
- (c) Except as set forth on Section 3.17(c) of the Company Disclosure Schedule, (i) no Benefit Plan is subject to Title IV of ERISA or Section 412 of the Code, and (ii) no Benefit Plan is a "multiemployer plan" within the meaning of Section 3(37) or 4001(a)(3) of ERISA.
- (d) With respect to each Benefit Plan, the Group Companies are in compliance in all material respects with ERISA, the Code and all Laws applicable to such Benefit Plans, and each Benefit Plan has been administered in all material respects in accordance with its terms and all applicable Laws. Each Benefit Plan that is intended to be qualified under Section 401(a) of the Code has received a favorable determination letter from the IRS, or with respect to a prototype plan, can rely on an opinion letter from the IRS to the prototype plan sponsor, to the effect that such Benefit Plan is so qualified, and, to the Knowledge of the Company, nothing has occurred that could reasonably be expected to cause the revocation of such determination letter from the IRS or the unavailability of reliance on such opinion letter from the IRS. All material benefits, contributions and premiums required by and due under the terms of each Benefit Plan or applicable Law have been timely paid in accordance with the terms of such Benefit Plan, the terms of all applicable Laws and GAAP.
- (e) No claim, Legal Proceeding, investigation, audit or other action (other than routine claims for benefits in the Ordinary Course of Business) is pending, or to the Knowledge of the Company, threatened against any Benefit Plan.
- (f) None of the Benefit Plans provide retiree health or welfare insurance benefits to any current or former employee of the Company or its Subsidiaries, except as may be required by Section 4980B of the Code, Part 6 of Subtitle B of Title I of ERISA or any similar Law requiring group health plan continuation coverage.
- (g) Except as set forth on Section 3.17(g) of the Company Disclosure Schedule, neither the execution and delivery of this Agreement nor the consummation of the Transactions will result in the payment of any amount under any Benefit Plan that would, individually or in combination with any other such payment, result in an "excess parachute payment" within the meaning of Section 280G of the Code.

- (h) No Benefit Plan providing for deferred compensation is subject to Code Section 409A(a) or Code Section 409A(b).
- (i) Notwithstanding any other provision of this Agreement to the contrary, this <u>Section 3.17</u> contains the sole and exclusive representation and warranties of the Company with respect to employee benefit matters.

### 3.18 Labor.

- (a) No Group Company is a party to any labor or collective bargaining agreement in respect of any employee or group of employees of the Group Companies, and there is no pending representation question or union organizing activity respecting employees of the Group Companies. Except as set forth on Section 3.18(a) of the Company Disclosure Schedule, (i) there are no, and within the period starting on the Balance Sheet Date and ending on the date hereof, there have been no, material strikes, work stoppages, work slowdowns, lockouts, picketing or other material labor disputes pending or, to the Knowledge of the Company, threatened against any Group Company, and (ii) there are no material unfair labor practice charges, grievances or complaints pending or, to the Knowledge of the Company, threatened by or on behalf of any employee or group of employees of any of the Group Companies against any of the Group Companies before a Governmental Authority.
- (b) Section 3.18(b) of the Company Disclosure Schedule sets forth a list of all employees of the Company as of the date that is two (2) Business Days prior to the date hereof, including each employee's name, title, date of hire, employment status (active or inactive), status as "exempt" or "not exempt," resident alien status (if applicable), and office location. The Company has provided to Parent a true and complete list of each employee's current base salary (or wages), target bonus, and other cash incentive compensation.
- (c) Each of the Group Companies is in compliance in all material respects with all agreements and all applicable laws relating to employment or labor. To the Knowledge of the Company, no present or former employee, officer, director or manager of the Group Companies has any claim against the Group Companies for unpaid wages, salary, or vacation or sick pay, or otherwise under any employment-related agreement.
- (d) Except as set forth on <u>Section 3.18(d)</u> of the Company Disclosure Schedule, to the Knowledge of the Company, since the Balance Sheet Date no executive or senior management-level employee of the Group Companies has provided notice of their intention to terminate his or her employment.
- (e) Except as set forth on <u>Section 3.18(e)</u> of the Company Disclosure Schedule, the Group Companies do not have any severance policy or severance obligation to any current or former employees of the Group Companies as of the date hereof.
- 3.19 <u>Transactions With Related Parties</u>. Except as set forth on <u>Section 3.19</u> of the Company Disclosure Schedule, no present officer, director, member or stockholder of any of the Group Companies, nor any Affiliate of any Group Company (each a "<u>Related Party</u>"), is currently

a party to any transaction or Contract with a Group Company, other than (i) Contracts with respect to the acquisition by, or merger with, a Group Company, where the seller or surviving company (or an Affiliate of the seller or surviving company) thereunder became a Related Party in connection with such transaction, and (ii) Contracts which will be terminated at or prior to Closing.

- 3.20 <u>Insurance</u>. <u>Section 3.20</u> of the Company Disclosure Schedule contains a list of all insurance policies owned or held by or on behalf of the Group Companies as of the date of this Agreement (the "<u>Insurance Policies</u>"). As of the date of this Agreement, all Insurance Policies are in full force and effect and the Group Companies have complied in all material respects with the provisions of such policies. All premiums due and payable on the Insurance Policies have been paid as of the date hereof. As of the date hereof, neither the Company nor any of the Operating Subsidiaries have received a written notice of cancellation of any Insurance Policy. There are no pending claims by the Group Companies under any of the Insurance Policies, including as to which a Group Company has received written notice that such claim has been denied.
- 3.21 <u>Financial Advisors</u>. Except as set forth on <u>Section 3.21</u> of the Company Disclosure Schedule, no Person has acted, directly or indirectly, as a broker, finder, agent, investment banker or financial advisor for the Sellers or the Group Companies and no Person other than those Persons set forth on <u>Section 3.21</u> of the Company Disclosure Schedule is entitled to any fee or commission or like payment based on the arrangements made by the Sellers or any Group Company in connection with the Transaction.

# 3.22 Privacy and Data Security.

- (a) Except as would not result in a Company Material Adverse Effect, each Group Company complies, and has complied, with, and each of their respective data privacy and security practices conforms with, Applicable Privacy and Security Laws and with the respective data privacy and information security obligations to which it is subject, including under contract, privacy policy, or online terms of use. Each Group Company maintains policies and procedures that comply with (i) Applicable Privacy and Security Laws and (ii) data privacy and information security obligations to its customers, data subjects, or other Persons, including under contract, privacy policy, or online terms of use. Without limiting the foregoing, the Group Companies have used commercially reasonable efforts to ensure that the Group Companies' data privacy and security practices are in compliance with any applicable requirements under the CCPA. No Group Company has received any claims, charges or written notice (including written notice from third parties acting on its behalf) from any Governmental Authority or other Person that it is committing, has committed or is under investigation for a violation of any of the Applicable Privacy and Security Laws, data privacy and security policies and notices or contractual obligations relating to data or Personal Information. To the Company's Knowledge, no person (including any Governmental Authority) has commenced any Claim with respect to the loss, theft, misuse, damage or unauthorized access, use, Processing, disclosure or modification of any data or Personal Information by or on behalf of the Group Companies.
- (b) All systems, including the Company Software, firmware, hardware, networks, interfaces, and related systems owned or licensed by the Group Companies (collectively, the "Business Systems") are sufficient for the needs of their business as currently conducted and

have been designed and implemented using commercially reasonable standards to ensure (i) data and Personal Information collected by or on behalf of the Group Companies is protected against loss, theft, misuse, damage and unauthorized access, use, Processing, disclosure or modification. In the last thirty-six (36) months, no part of the Business Systems has been materially prone to malfunction or error and there has been (i) no material disruption, interruption or outage to any Business System, (ii) no material unauthorized disclosure of or access to data or Personal Information, (iii) no material unauthorized breaches of the security of or intrusion into the Business Systems and (iv) no Claim or circumstance requiring a Group Company to notify a Governmental Authority or any affected party of a data security breach or violation of any Applicable Privacy and Security Laws. The Group Companies have safeguarded their Business Systems with commercially reasonable information security controls, and disaster recovery and business continuity practices.

- (c) The Group Companies have all of the necessary rights, permission and licenses to the personally identifiable data received from customers or clients using Company Software, products and services, except where the lack of such rights, permission and licenses would not have a material adverse effect on any Group Company.
- 3.23 <u>Customers</u>. With respect to any Contract expected to involve annual payments or consideration furnished to any Group Company for the fiscal year ending December 28, 2019 of more than \$1,000,000, except as set forth on <u>Section 3.23</u> of the Company Disclosure Schedules, as of the date hereof, no customer (or former customer) of any Group Company during the last twelve (12) months has canceled, terminated or, to the Company's Knowledge, threatened in writing to cancel or otherwise terminate such Contract, or to materially decrease its usage of the Group Company's services or products. As of the date hereof, the Group Companies have not received any written notice, and the Company has no Knowledge, to the effect that any current customer under such Contract intends to terminate or materially alter its business relations with any Group Company as a result of the Transactions.
- 3.24 <u>Bank Accounts. Section 3.24</u> of the Company Disclosure Schedules sets forth the names and locations of all banks, trust companies, savings and loan associations and other financial institutions at which the Group Companies maintain accounts of any nature (or at which an account is maintained to which the Group Company has access as to which deposits are made on behalf of the Group Company), in each case listing the type of account, the account number therefor, and the names of all Persons authorized to draw thereon, make withdrawals therefrom or have access thereto, and lists all locations of all safe deposit boxes used by the Group Company. There are no outstanding powers of attorney executed on behalf of any Group Company with respect to any such accounts or safe deposit boxes.

## 3.25 Backlog.

(a) <u>Section 3.25</u> of the Company Disclosure Schedules sets forth the projects of the Group Companies that are outstanding and require the performance of additional services that have already been ordered by the applicable customer to complete them, as of the date hereof (the "<u>Backlog Contracts</u>"), including (a) each customer, (b) whether the applicable contract is for a fixed price or other type of contract, (c) the contract revenue for 2018 and the contract revenue as of the Balance Sheet Date, (d) the contract's dollar value, and (e) the dollar amount of the backlog (for

the avoidance of doubt, this clause (e) only includes customer orders that have been placed but not yet fulfilled and excludes additional orders that may be placed in the future at customer's discretion). Except as set forth on Section 3.25 of the Company Disclosure Schedules, no Backlog Contracts are at prices which, based on the past experience of any Group Company and current and anticipated costs, have resulted, or can reasonably be expected to result, in a material loss to the Group Company.

(b) Except as set forth on Section 3.25 of the Company Disclosure Schedules, no Group Company is and, to the Knowledge of the Company, no other party to any Backlog Contract is in violation or breach of, or in default under, nor has there occurred any event or condition that with or without the passage of time or the giving of notice (or both) would constitute a violation or breach of, or default under, or permit the default termination of, any Backlog Contract or would cause acceleration of any obligation of any party thereto or create any Lien on the property of any Group Company, except, in each case, as would not result in a Company Material Adverse Effect.

# 3.26 Accounts Receivable and Accounts Payable.

- (a) All Accounts Receivable of the Group Companies are, or in the case of Work-in-Progress, will be, valid receivables that have arisen from bona fide transactions entered into on an arms' length basis in the Ordinary Course of Business. Except as set forth on Schedule 3.26 of the Company Disclosure Schedule, the Accounts Receivable of the Group Companies are current and, to the Knowledge of the Company, are, or in the case of Work-in-Progress, will be, collectible net of the reserve shown on the balance sheet included in the Financial Statements dated as of the Balance Sheet Date (which reserve is adequate and calculated in accordance with GAAP) without any setoff, expense or other reduction. There is no contest, claim, defense or right of setoff, other than returns in the Ordinary Course of Business, with respect to any Accounts Receivable. Schedule 3.26 of the Company Disclosure Schedule sets forth (i) the aging of all Accounts Receivable and (ii) a listing of all Work-in-Progress, in each case, as of the Balance Sheet Date and with respect to balances over \$100,000.
- (b) The accounts payable of the Group Companies reflected on the Balance Sheet arose from bona fide transactions with third parties on an arms' length basis in the Ordinary Course of Business.
- 3.27 <u>Product and Service Warranties</u>. As of the date hereof, since January 1, 2017, no Group Company has received any written legal claims or had any legal claims threatened, in each case, involving amounts in excess of \$100,000, with respect to any warranties or guarantees pertaining to any products or services provided by the Group Companies.
- 3.28 LIMITATIONS OF REPRESENTATIONS AND WARRANTIES. EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES CONTAINED IN THIS ARTICLE III (AS MODIFIED, SUBJECT TO THE LIMITATIONS AND QUALIFICATIONS SET FORTH IN SECTION 10.2(c), IN EACH CASE, BY THE COMPANY DISCLOSURE SCHEDULE), THE COMPANY MAKES NO, AND HAS NOT AUTHORIZED ANY OF ITS AFFILIATES TO MAKE ANY OTHER, EXPRESS OR IMPLIED REPRESENTATION OR WARRANTY WITH RESPECT TO THE COMPANY, THE GROUP COMPANIES OR THE TRANSACTION, AND THE COMPANY DISCLAIMS ANY OTHER REPRESENTATIONS OR WARRANTIES, WHETHER

MADE BY THE COMPANY, ANY AFFILIATE OF THE COMPANY OR ANY OF THEIR RESPECTIVE OFFICERS, DIRECTORS, MANAGERS, EMPLOYEES, AGENTS OR REPRESENTATIVES AND IF MADE, SUCH REPRESENTATION OR WARRANTY MAY NOT BE RELIED UPON BY PARENT, MERGER SUB OR ANY OF THEIR RESPECTIVE AFFILIATES AND REPRESENTATIVES AS HAVING BEEN AUTHORIZED BY THE COMPANY OR ANY OF ITS AFFILIATES. EXCEPT FOR THE REPRESENTATIONS AND WARRANTIES CONTAINED IN THIS ARTICLE III (AS MODIFIED, SUBJECT TO THE LIMITATIONS AND QUALIFICATIONS SET FORTH IN SECTION 10.2(c), IN EACH CASE, BY THE COMPANY DISCLOSURE SCHEDULE), THE COMPANY HEREBY DISCLAIMS ALL LIABILITY AND RESPONSIBILITY FOR ANY REPRESENTATION, WARRANTY, OPINION, PROJECTION, FORECAST, STATEMENT, MEMORANDUM, PRESENTATION, ADVICE OR INFORMATION MADE, COMMUNICATED, OR FURNISHED (ORALLY OR IN WRITING) TO PARENT, MERGER SUB OR THEIR RESPECTIVE AFFILIATES OR REPRESENTATIVES (INCLUDING ANY OPINION, PROJECTION, FORECAST, STATEMENT, MEMORANDUM, PRESENTATION, ADVICE OR INFORMATION THAT MAY HAVE BEEN OR MAY BE PROVIDED TO PARENT OR MERGER SUB BY ANY DIRECTOR, OFFICER, EMPLOYEE, AGENT, CONSULTANT, OR REPRESENTATIVE OF THE COMPANY OR ANY OF ITS AFFILIATES, INCLUDING ANY INFORMATION MADE AVAILABLE IN ANY ELECTRONIC DATA ROOM HOSTED BY THE COMPANY OR ANY OF ITS REPRESENTATIVES IN CONNECTION WITH THE TRANSACTION). THE COMPANY DOES NOT MAKE ANY REPRESENTATIONS OR WARRANTIES TO PARENT OR MERGER SUB REGARDING THE PROBABLE SUCCESS OR PROFITABILITY OF THE BUSINESS CONDUCTED BY THE GROUP COMPANIES.

### **ARTICLE IV**

### REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUB

Except as set forth on the disclosure schedule delivered by Parent to the Representative and the Company concurrently with entry into this Agreement (the "<u>Parent Disclosure Schedule</u>"), which are subject to the limitations and qualifications set forth in <u>Section 10.2(c)</u>, each of Parent and Merger Sub hereby represents and warrants to the Company as of the date hereof as follows:

4.1 Organization and Power. Parent is a corporation duly organized, validly existing and in good standing under the Laws of the State of Delaware. Merger Sub is a Delaware corporation duly organized, validly existing and in good standing under the Laws of the State of Delaware. Each of Parent and Merger Sub has the requisite corporate power and authority to execute and deliver each Transaction Agreement to which it is a party, to perform its obligations hereunder and thereunder and to consummate the Transactions. Each of Parent and Merger Sub has the corporate power and authority to own or lease all of its properties and assets and to carry on its business as it is now being conducted. Each of Parent and Merger Sub is duly licensed or qualified to do business in each jurisdiction in which the nature of its business or the character or location of any properties or assets owned or leased by it makes such licensing or qualification necessary, except for those jurisdictions where the failure to be so licensed or qualified would not have, individually or in the aggregate, a Parent Material Adverse Effect.

4.2 <u>Authorization of Agreement</u>. The execution and delivery of the Transaction Agreements to which it is a party and the performance of its obligations thereunder have been duly authorized by the requisite corporate action on the part of each of Parent and Merger Sub. No other proceeding, action or approval on the part of either Parent or Merger Sub (including by its respective equityholders) is necessary to authorize the Transaction Agreements to which it is a party or to consummate the Transactions. Each of the Transaction Agreements to which it is a party has been duly and validly executed and delivered by each of Parent and Merger Sub and (assuming the due authorization, execution and delivery by the other parties thereto) each of the Transaction Agreements, when so executed and delivered, will constitute, the legal, valid and binding obligations of Parent and/or Merger Sub, as applicable, enforceable against it in accordance with its terms, subject to applicable Equitable Principles.

## 4.3 Conflicts; Consents of Third Parties.

- (a) Assuming all Governmental Approvals contemplated by Section 4.3(b) below have been obtained and are effective and all applicable waiting periods have expired or been terminated and all filings and notifications described in Section 3.3(b) of the Company Disclosure Schedule have been made, none of the execution, delivery and performance by either Parent or Merger Sub of the Transaction Agreements to which it is a party, or the consummation of the Transaction by Parent and Merger Sub, will conflict with, violate or constitute a default (with or without notice or lapse of time, or both) under or give rise to a right of termination, acceleration, modification or cancellation under any provision of (A) the Organizational Documents of Parent or Merger Sub, as applicable; (B) any Contract or Permit which Parent or Merger Sub or any of their Affiliates, as applicable, is a party to or bound by, or by which Parent's or Merger Sub's or any of their Affiliates, as applicable, properties or assets are bound; or (C) any Law applicable to Parent or Merger Sub or any of their Affiliates, as applicable, except, in the case of clauses (B) and (C), where such conflict, violation acceleration, termination, modification, cancellation or default, would not have, individually or in the aggregate, a Parent Material Adverse Effect.
- (b) Except as set forth on <u>Section 4.3(b)</u> of the Parent Disclosure Schedule, no Governmental Approval is required on the part of either Parent or Merger Sub in connection with the execution and delivery by such Party of the Transaction Agreements to which it is a party, or the consummation of the Transactions by such Party, except for any Governmental Approval (i) described in <u>Section 3.3(b)</u> of the Company Disclosure Schedule or (ii) the failure of which to make or obtain would not have, individually or in the aggregate, a Parent Material Adverse Effect.
- 4.4 <u>Legal Proceedings</u>. There are no pending or, to the knowledge of Parent or Merger Sub, threatened, Legal Proceedings against Parent or Merger Sub or their Affiliates that would have, individually or in the aggregate, a Parent Material Adverse Effect. There is no outstanding material Order imposed upon either Parent or Merger Sub or any of their assets or Affiliates, except for Legal Proceedings which, if adversely determined, would not have, individually or in the aggregate, a Parent Material Adverse Effect
- 4.5 <u>Financial Capability</u>. Parent will have as of the Closing, (i) sufficient cash available to pay the Merger Consideration and all related fees and expenses in connection with the Transactions, (ii) the resources and capabilities (financial and otherwise) to perform its obligations

hereunder and (iii) has not incurred, and as of the Closing will not have incurred, any obligation, condition, commitment, restriction or liability of any kind that would impair or adversely affect such resources and capabilities.

- 4.6 <u>Solvency</u>. Upon consummation of the Merger, Parent and the Group Companies, on a consolidated basis, will not, solely as a result of the consummation of such Transaction, (i) be insolvent or have incurred debts beyond their ability to pay such debts as they mature or (ii) have unreasonably small capital with which to engage in their respective businesses. No transfer of property is being made and no obligation is being incurred in connection with the Transaction with the intent to hinder, delay or defraud either present or future creditors of the Company or any of its Subsidiaries.
- 4.7 <u>Investment</u>. Parent is acquiring the equity securities of the Group Companies for its own account and for investment purposes and not with a view to the distribution thereof. Parent acknowledges that such equity securities have not been registered under the Securities Act or any state securities Law and Parent must bear the economic risk of its investment in such securities until and unless the offer and sale of such securities is subsequently registered under the Securities Act and all applicable state securities Laws or an exemption from such registration is applicable. Parent has conducted an examination of available information relating to the Group Companies and their respective businesses, Parent has such knowledge, sophistication and experience in business and financial matters that it is capable of evaluating an investment in such securities, and Parent can bear the substantial economic risk of an investment in such securities for an indefinite period of time and can afford a complete loss of such investment.
- 4.8 <u>Financial Advisors</u>. Except as set forth on <u>Section 4.8</u> of the Parent Disclosure Schedule, no Person has acted, directly or indirectly, as a broker, finder, agent, investment banker or financial advisor for Parent, Merger Sub or their respective Affiliates and no Person is entitled to any fee or commission or like payment based on the arrangements made by Parent, Merger Sub or their respective Affiliates in connection with the Transaction.
  - 4.9 No Other Representations and Warranties; No Reliance; Parent and Merger Sub Investigation.
- (a) Each of Parent and Merger Sub acknowledges and agrees that, except as expressly set forth in <u>ARTICLE III</u>, the Company makes no promise, representation or warranty, express or implied, relating to the Group Companies or any of their respective businesses, operations, assets, liabilities, conditions or prospects or the Transaction, including with respect to merchantability, fitness for any particular or ordinary purpose, or as to the accuracy or completeness of any information regarding any of the foregoing, or as to any other matter, notwithstanding the delivery or disclosure to Parent and/or Merger Sub or any of its Affiliates or representatives of any documents, opinions, projections, forecasts, statements, memorandums, presentations, advice or information (whether communicated orally or in writing), and any such other promises, representations or warranties, or liability or responsibility therefor, are hereby expressly disclaimed. In addition, each of Parent and Merger Sub acknowledges and agrees that it has not executed or authorized the execution of this Agreement in reliance upon any promise, representation or warranty not expressly set forth in <u>ARTICLE III</u>.

- (b) In respect of this Agreement and the Transaction, neither Parent nor Merger Sub has relied or is relying on any document or written or oral information (including, but not limited to, the Confidential Information Packet prepared by Houlihan Lokey Capital, Inc.), statement, representation or warranty furnished to or discovered by it or any of its Affiliates other than the representations and warranties set forth in this Agreement.
- (c) Each of Parent and Merger Sub acknowledges that it is has made its own inquiry and is relying on its own independent investigation and analysis in entering into the Transaction. Each of Parent and Merger Sub is knowledgeable about the industries in which the Company operates and is capable of evaluating the merits and risks of the Transaction. Each of Parent and Merger Sub has been afforded full access to the books and records, facilities and personnel of the Company for purposes of conducting a due diligence investigation and has conducted a full due diligence investigation of the Company. The foregoing provisions of this Section 4.9(c) shall not in any way limit the representations and warranties contained in ARTICLE III.
- (d) Except as otherwise expressly set forth in this Agreement, each of Parent and Merger Sub understands and agrees that the Company, its Operating Subsidiaries and each of their respective assets are being furnished "as is", "where is" and, subject to the representations and warranties contained in <u>ARTICLE III</u>, with all faults and without any other representation or warranty of any nature whatsoever or elsewhere in this Agreement.

### ARTICLE V

## **COVENANTS**

- 5.1 <u>Conduct of Business</u>. Except as contemplated by this Agreement, from and after the date hereof until the earlier of the Closing or the termination of this Agreement in accordance with its terms, the Company shall, and shall cause each other Group Company to, except as set forth on <u>Section 5.1</u> of the Company Disclosure Schedule or as consented to in writing by Parent (which consent shall not be unreasonably withheld, conditioned, or delayed), (a) conduct its business in the Ordinary Course of Business (including any conduct that is directly related, complementary or incidental thereto), (b) use commercially reasonable efforts to preserve substantially intact its business organization and to preserve the present commercial relationships with key Persons with whom it does business and (c) not do any of the following:
  - (i) make any capital expenditure in excess of \$350,000 individually or \$750,000 in the aggregate;
  - (ii) take or omit to take any action that would reasonably be expected to result in a Company Material Adverse Effect;
  - (iii) declare or pay a dividend on, or make any other distribution in respect of, its equity securities except dividends and distributions by an Operating Subsidiary to the Company or dividends or distributions solely in cash;

- (iv) acquire or agree to acquire in any manner (whether by merger or consolidation, the purchase of an equity interest in or a material portion of the assets of or otherwise) any business or any Person or other business organization or division thereof of any other Person (excluding the acquisition of assets not constituting substantially all of the assets of another Person and in the Ordinary Course of Business);
- (v) amend, extend, renew or terminate any Material Contract, Material Government Contract or Real Property Lease, as applicable, other than any Contracts or extensions (A) with a term of less than one year, (B) which involve \$500,000 or less, or (C) amended, extended, renewed or terminated in the Ordinary Course of Business;
- (vi) change in any material respect the base compensation of, or enter into any new bonus or incentive agreement or arrangement (other than any transaction, retention, change of control or similar payments which, in each case, are payable in connection with the transactions contemplated by this Agreement) with, any of its employees, other than changes made in accordance with normal compensation practices and consistent with past practices of the Group Companies or changes required by employment agreements or by any Law;
- (vii) materially amend or enter into a new, Benefit Plan (except as required by Law, a Contract in effect on the date hereof or customary renewals of existing Benefit Plans in the Ordinary Course of Business) or collective bargaining;
- (viii) incur any Indebtedness in excess of \$1,000,000 in the aggregate (with respect to the Group Companies, taken as a whole), except (1) current liabilities incurred in the Ordinary Course of Business or (2) borrowings under existing credit facilities, and (3) obligations under Contracts entered into in the Ordinary Course of Business;
- (ix) issue any equity interests or grant any option or issue any warrant to purchase or subscribe for any such securities or issue any securities convertible into such securities (except in connection with the exercise or conversion of equity securities, options and warrants issued and outstanding as of the date hereof);
  - (x) adopt any amendments to their respective Organizational Documents;
- (xi) make any change in the accounting principles, methods, practices or policies applied in the preparation of the Financial Statements, unless such change is required by applicable Law or GAAP;
- (xii) sell, or otherwise dispose of, any (A) intangible, or (B) material tangible assets in excess of \$500,000 in the aggregate, other than sales of inventory in the Ordinary Course of Business and personal property sold or otherwise disposed of in the Ordinary Course of Business and except for any tangible asset which is obsolete;

- (xiii) make, change or revoke any Tax election; change any annual Tax accounting period; change any Tax accounting principles, methods, practices or policies; file any amended Tax Return; or enter into any Tax allocation agreement, Tax sharing agreement, or Tax indemnity agreement (other than commercial Contracts entered into in the Ordinary Course of Business that do not primarily relate to Taxes); or
  - (xiv) agree in writing to do anything contained in this clause (c).

## 5.2 Access to Information.

- (a) From and after the date hereof until the earlier of the Closing Date or the termination of this Agreement in accordance with its terms, upon reasonable advance notice, and subject to the restrictions contained in the confidentiality agreements to which the Group Companies are subject, the Company shall provide to Parent and Parent's authorized representatives during normal business hours reasonable access to all books and records of the Group Companies (in a manner so as to not interfere with the normal business operations of any Group Company) for any reasonable purpose (provided, the continuation of due diligence shall not be deemed a reasonable purpose). All of such information shall be treated as confidential information pursuant to the terms of the Confidentiality Agreement. Notwithstanding anything to the contrary in this Agreement, the Company shall not be required to disclose any competitively sensitive information or disclose any other information to Parent or its representatives if (x) such disclosure would be reasonably likely to jeopardize any attorney-client or other legal privilege, (y) such disclosure would be reasonably likely to contravene any applicable Laws, fiduciary duty, or binding agreement entered into prior to the date hereof, or (z) the Parties are in an adversarial relationship in litigation or arbitration (in which case the furnishing of information, documents or records contemplated by this Section 5.2(a) shall be subject to applicable rules relating to discovery) (the matters referred to in this sentence with respect to any Person, the "Access Limitations").
- (b) From and after the Closing Date, in connection with any reasonable business purpose, including the determination of any matter relating to the rights or obligations of the Sellers under this Agreement, upon reasonable prior request and subject to the Access Limitations, Parent shall, and shall cause the Group Companies to, (i) afford the Representative and its authorized representatives reasonable access, during normal business hours, to the offices, properties, books, records and other documents of Parent and its Affiliates solely to the extent in respect of the Group Companies and (ii) make reasonably available to the Representative and its authorized representatives the employees of the Company, Parent and its Affiliates in respect of the Group Companies whose assistance, expertise, testimony, notes and recollections or presence is necessary to assist the Representative in connection with the inquiries for any of the purposes referred to above, including the presence of such Persons as witnesses in hearings or trials for such purposes; provided, however, that (x) such requests shall not interfere in any significant manner with the normal operations of Parent or any of its Affiliates or would be reasonably likely to jeopardize any attorney-client or other legal privilege, or would be reasonably likely to contravene any applicable Laws, fiduciary duty or binding agreement entered into prior to the date hereof, (y) the auditors and accountants of Parent or its Affiliates shall not be obligated to make any work papers (to the extent such documents exist) available to any Person unless and until such Person has signed a customary

agreement relating to such access to work papers in form and substance reasonably acceptable to such auditors or accountants, and (z) if the Parties are in an adversarial relationship in litigation or arbitration, the furnishing of information, documents or records contemplated by this Section 5.2(b) shall be instead subject to applicable rules relating to discovery.

(c) During the period from the date of this Agreement until the earlier of the Closing Date or the termination of this Agreement in accordance with its terms, Parent hereby agrees that it is not authorized to and shall not (and shall not permit any of its employees, agents, representatives or Affiliates to) contact any customer, supplier, distributor, officer, employee or other material business relation of any Group Company regarding the Transactions without the prior written consent of the Company.

## 5.3 Efforts; Regulatory Filings and Consents.

- (a) Without prejudice to Parent's obligations set forth in Section 5.3(d), each of the Company, on the one hand, and Parent and Merger Sub, on the other hand, shall use its respective commercially reasonable efforts to take, or cause to be taken, all actions and to do, or cause to be done, and to assist and cooperate with the other Parties in doing, all things necessary, proper or advisable (i) to consummate and make effective as promptly as possible, but in no event later than the Termination Date, the Transactions, (ii) to obtain all Governmental Approvals of any Governmental Antitrust Authority required to be obtained or made by the Company, Parent or Merger Sub, or any of their respective Subsidiaries or Affiliates in connection with the Transactions or the taking of any action contemplated by this Agreement, (iii) to defend, lift, mitigate or rescind the effect of any litigation or administrative proceeding involving any Governmental Antitrust Authority (including a private party challenge) adversely affecting this Agreement or the Transaction, including promptly appealing any adverse court or administrative decision; provided, however, that the Company and its Affiliates shall not be required to (A) make any material monetary expenditures or commence or be a plaintiff in any litigation to satisfy their obligations under this Section 5.3; or (B) sell or otherwise dispose of any portion of the business of the Group Companies or offer or grant any material accommodation (financial or otherwise) to any Person in order to satisfy their obligations under this Section 5.3.
- (b) Each of the Company and Parent shall (i) as soon as reasonably practicable (and in any event within five (5) Business Days following the date of this Agreement) file with the United States Federal Trade Commission (the "FTC") and the United States Department of Justice (the "DOJ") the notification and report form, if any, required for the Transaction and to supply as promptly as practicable any supplemental information requested in connection therewith pursuant to the HSR Act and (ii) as soon as reasonably practicable (and in any event within five (5) Business Days following the date of this Agreement) make all filings under the applicable Other Competition Laws, if any, required for the Transaction, and shall take all other actions necessary, proper or advisable to cause the expiration or termination of the applicable waiting period under the HSR Act and the applicable Other Competition Laws. Any such antitrust notification and report form or filing and supplemental information shall be in substantial compliance with the requirements of the HSR Act or the applicable Other Competition Laws, as the case may be. All other regulatory filings shall be in substantial compliance with the requirements of applicable Law. Each of Parent and the

Company shall furnish to the other Party such necessary information and reasonable assistance as the other party may reasonably request in connection with its preparation of any filing or submission that is necessary under the HSR Act, the applicable Other Competition Laws or other applicable Law, as the case may be. The Company and Parent shall use reasonable best efforts to comply promptly with any inquiries or requests for additional information from the FTC, the DOJ, other Governmental Authorities and any other Governmental Authority having jurisdiction.

- (c) Without limiting the generality of the undertakings set forth in Sections 5.3(a), 5.3(b) and 5.3(d) and subject to any appropriate confidentiality protections, the Company, on the one hand, and Parent and Merger Sub, on the other hand, shall each furnish to the other such necessary information and reasonable assistance as the other may reasonably request in connection with this Section 5.3 and shall each promptly provide counsel for the other party with copies of all filings made by such party, and all correspondence between such party (and its advisors) with any Governmental Antitrust Authority, other Governmental Authority or, in connection with any proceeding by a private party, and any other information supplied by such party and such party's Affiliates to a Governmental Antitrust Authority or other Governmental Authority in connection with his Agreement and the Transaction. Subject to applicable Law, the Company and Parent shall permit counsel for the other Party reasonable opportunity to review in advance, and shall consider in good faith the views of the other party in connection with, any proposed written or, if practicable, oral communication to any Governmental Antitrust Authority or other Governmental Authority relating to the Transaction. Each of the Company and Parent agrees not to participate in any substantive meeting or discussion, either in person or by telephone, with any Governmental Antitrust Authority or other Governmental Authority, it consults with the other Party in advance, and gives the other Party the opportunity to attend and participate.
- (d) Notwithstanding anything to the contrary in this Agreement, Parent and its Affiliates shall take, or cause to be taken, any and all actions, including the disposition of assets, required by any Governmental Antitrust Authority as a condition to the granting of any Governmental Approval necessary for the consummation of the Transactions or as may be required to avoid, lift, vacate, reverse or resolve any legislative, administrative or judicial action (including any suit instituted (or threatened to be instituted) by the FTC, the DOJ or any other applicable Governmental Authority or any private party challenging any of the Transaction as violative of the HSR Act or Other Competition Laws) that would otherwise reasonably be expected to materially impair or delay the consummation of the Transactions. Without limiting the generality of the foregoing, such actions may include, but are not limited to: (i) divesting, selling or otherwise disposing of, or holding separate and agreeing to sell or otherwise dispose of, any entity, facility or asset of Parent or its Affiliates (including the Group Companies), (ii) terminating, amending or assigning any existing relationships and contractual rights and obligations, and (iii) amending, assigning or terminating any existing licenses or other agreements and entering into such new licenses or other agreements (and, in each case, entering into agreements with the relevant Governmental Antitrust Authority giving effect thereto), in each case with respect to the foregoing clauses (i), (ii) or (iii) if such action is necessary or reasonably advisable as a condition to the granting of any Governmental Approval necessary for the consummation of the Transactions or as

may be required to avoid, lift, vacate reverse or resolve any legislative, administrative or judicial action (including any suit instituted (or threatened to be instituted) by the FTC, the DOJ or any other applicable Governmental Authority or any private party challenging any of the Transactions as violative of the HSR Act or Other Competition Laws) that would otherwise reasonably be expected to materially impair or delay the consummation of the Transactions; provided that any such action shall not be required to be effective prior to the Closing.

- (e) The filing fees under the HSR Act, Other Competition Laws and other applicable Laws, as well as the fees and disbursements of any legal counsel or other advisor jointly retained by the Parties in connection with any such filings and any other filings with Governmental Authorities, shall be borne by Parent.
- (f) Without limiting any other obligation under this Agreement, during the period from the date of this Agreement until the Closing Date, Parent, Merger Sub and their respective Subsidiaries and Affiliates shall not take or agree to take any action that would reasonably be expected to prevent or delay the parties from obtaining any Governmental Approval in connection with the Transactions, including entering into an agreement to acquire (whether via merger, consolidation, stock or asset purchase or otherwise) any material amounts of assets of or any equity in any other Person or any business or division thereof if such agreement would be reasonably expected to create a material risk of making it more difficult to obtain the Governmental Approval of the FTC or DOJ or any other Governmental Antitrust Authority required in connection with the Transactions.
- 5.4 <u>Confidentiality</u>. Each of Parent and Merger Sub acknowledges that the information provided to it and its representatives in connection with this Agreement (including <u>Section 5.2(a)</u> hereof) and the Transaction is subject to the terms of the Confidentiality Agreement, dated July 31, 2019, by and between Parent and the Company (the "<u>Confidentiality Agreement</u>"), the terms of which are incorporated herein by reference.
- 5.5 <u>Preservation of Records</u>. In addition to and not in limitation of the provisions of <u>Section 5.2(a)</u>, Parent agrees to preserve and keep the records relating to the businesses of the Group Companies for a period of seven (7) years from the Closing Date and shall make such records and personnel available to the Representative as may be reasonably requested in connection with any insurance claims by, Legal Proceedings (other than Legal Proceedings between the Representative and Parent related to this Agreement or the Transaction) or Tax audits against, or governmental investigations of, the Group Companies or in order to enable the Representative to comply with its obligations under this Agreement and each other Transaction Agreement.
- 5.6 <u>Publicity</u>. None of the Representative or, prior to the Closing, the Company, on the one hand, or Parent, Merger Sub or, following the Closing, the Surviving Company, on the other hand, shall issue any press release or public announcement concerning this Agreement, the other Transaction Agreements or the Transaction or make any other public disclosure containing or pertaining to the terms of this Agreement without obtaining the Representative's or Parent's, as applicable, prior written approval, which approval will not be unreasonably withheld or delayed, unless, in the judgment of the Party seeking to disclose, disclosure is otherwise required by applicable Law or by the applicable rules of any stock exchange on which such disclosing Party lists securities;

provided that, to the extent any disclosure is required by applicable Law or stock exchange rule, the Party intending to make such disclosure shall use its commercially reasonable efforts consistent with applicable Law or stock exchange rule to consult with the Representative or Parent, as applicable, with respect to the text thereof and; provided, further, that (i) the Company and Representative, on the one hand, and Parent and its equityholders and Affiliates, on the other hand, shall be entitled to disclose such information to their respective directors, officers, executive employees, equity owners, partners, prospective partners, investors, prospective investors, professional advisors and lenders who have a need to know the information and who agree to keep such information confidential or are otherwise bound to confidentiality, (ii) Parent's financing sources and other professional advisors may publish "tombstones" or other customary announcements which do not contain pricing details that are not otherwise publicly available, and (iii) after the Closing, the Parent may publicly announce that the Transaction has closed and comply with its reporting requirements pursuant to applicable Law.

## 5.7 <u>Director and Officer Liability; Indemnification</u>.

- (a) For a period of six (6) years after the Closing, (i) Parent shall not, and shall not permit the Group Companies to, amend, repeal or modify any provision in any of their Organizational Documents relating to the exculpation, indemnification or advancement of expenses of any present or former officers, managers and/or directors (each, a "D&O Indemnified Person") (unless and to the extent required by Law) in a manner that would materially adversely affect the rights of D&O Indemnified Persons and (ii) Parent shall, and shall cause the Group Companies to, to the fullest extent permitted by applicable Law, (A) indemnify and hold harmless the D&O Indemnified Persons against all D&O Expenses (as defined below) and all Losses, claims, damages, judgments and amounts paid in settlement in respect of any threatened, pending or completed claim, action or proceeding, whether criminal, civil, administrative or investigative, based on or arising out of or relating to the fact that such Person is or was a director or officer of any of the Group Companies or arising out of acts or omissions occurring on or prior to the Closing (a "D&O Indemnifiable Claim") and (B) advance to such D&O Indemnified Persons all D&O Expenses incurred in connection with any D&O Indemnifiable Claim promptly after receipt of statements therefor. Any D&O Indemnifiable Claims shall continue until such D&O Indemnifiable Claim is disposed of or all Orders in connection with such D&O Indemnifiable Claim are fully and finally satisfied. For the purposes of this Agreement, "D&O Expenses" shall include reasonable attorneys' fees and all other reasonable costs, charges and expenses paid or incurred in connection with investigating, defending, being a witness in or participating in (including on appeal), or preparing to defend, to be a witness in or participate in any D&O Indemnifiable Claim.
- (b) Parent shall cause the Company, at Parent's expense, to purchase and maintain in effect beginning on the Closing Date and for a period of six (6) years thereafter without any lapses in coverage, a "tail" policy providing directors' and officers' liability insurance coverage for the benefit of those Persons who are covered by any Group Company's directors' and officers' liability insurance policies as of the date hereof or at the Closing with respect to matters occurring prior to the Closing. Such policy shall provide coverage that is at least equal to the coverage provided under the Group Companies' current directors' and officers' liability insurance policies; provided that the Company may substitute therefor policies of at least the same coverage containing terms

and conditions which are no less advantageous to the beneficiaries thereof so long as such substitution does not result in gaps or lapses in coverage with respect to matters occurring prior to the Closing Date.

- (c) If Parent, the Surviving Company, any of the Operating Subsidiaries or any of their respective successors or assigns (i) consolidates with or merges into any other Person and is not the continuing or surviving corporation or entity of such consolidation or merger or (ii) transfers or conveys all or substantially all of its properties and other assets to any Person, then, and in each such case, Parent shall cause proper provision to be made so that the applicable successors and assigns or transferees expressly assume the obligations set forth in this <u>Section 5.7</u>.
- 5.8 R&W Insurance Policy. On or before the Closing, Parent shall deposit with the Insurer the portion of the premium required under the terms of the R&W Insurance Policy in order to bind the Insurer as of the Closing and take any other actions necessary to cause the R&W Insurance Policy to incept as of the Closing. Parent shall cause the R&W Insurance Policy to be issued promptly after the Closing. Prior to the Closing, Parent shall pay or cause to be paid all of the costs and expenses related to the R&W Insurance Policy, including, without limitation, the total premium, underwriting costs, brokerage commission, due diligence fees, Taxes related to such policy and other fees and expenses of such policy and take all steps and do all other actions that may be reasonably required in order to ensure that the R&W Insurance Policy is issued on or before the Closing Date. The R&W Insurance Policy shall be on customary terms and conditions (including, for the avoidance of doubt, (i) the Insurer expressly waiving, and agreeing not to pursue, directly or indirectly, except in the case of fraud by the Company in the making of representations and warranties in this Agreement, any subrogation rights against the Group Companies, any Seller or any Seller's equityholders with respect to any claim made by any insured thereunder and (ii) the Insurer expressly agreeing that Parent and its Affiliates shall have no obligation to pursue any claim against the Group Companies or any Seller in connection with any Loss, liability or damage related thereto). Parent covenants and agrees to not cancel, redeem or take any action that would adversely affect the terms and conditions of the R&W Insurance Policy. Parent and its Affiliates will not (i) amend, waive or otherwise modify the R&W Insurance Policy in any manner that would allow the insurer thereunder or any other Person to, except in the case of fraud by the Company in the making of representations and warranties in this Agreement, subrogate or otherwise make or bring any action or proceedings against any Seller, any Seller's equityholders, any Group Company or any of its Affiliates or any past, present or future director, manager, officer, employee or advisor of any Seller or Group Company based upon, arising out of, or related to this Agreement, or the negotiation, execution or performance of this Agreement or that would reasonably be likely to result in any such Person having any liability arising from any breach of any representation or warranty in this Agreement or any other Transaction Agreement or (ii) novate or otherwise assign its rights under such R&W Insurance Policy (or do anything which has a similar effect).

### 5.9 Financing.

(a) Prior to the Closing, the Company shall, and shall cause the Operating Subsidiaries to, use commercially reasonable efforts to, and to cause its and their respective officers, managers, employees, consultants, counsel, accountants, agents, advisors and other representatives

(the "Company's Representatives") to use their commercially reasonable efforts (except to the extent it would be reasonably expected to unreasonably interfere with the ongoing operations of the Group Companies or would include any actions that the Company reasonably believes would result in a violation of any confidentiality arrangement or material agreement or the loss of any attorney-client or other similar privilege) to cooperate with any debt financing that Parent may be arranging in connection with the Transaction (the "Financing") (it being understood that the Company shall have satisfied its obligations set forth in this sentence if the Company shall have used its reasonable best efforts to comply with such obligations whether or not any applicable deliverables are actually obtained or provided). The foregoing notwithstanding, (i) none of the Group Companies nor any of their respective directors, officers or employees shall be required to pass resolutions or consents to approve or authorize the execution of the Financing or execute or deliver any definitive agreements related to the Financing, (ii) no obligation of the Company, the Operating Subsidiaries or any other Person undertaken pursuant to the foregoing shall be effective until the Closing, (iii) none of the Group Companies nor any of the Company's Representatives shall be required to pay, or commit to pay, any fee or incur any other cost or expense in connection with the Financing prior to the Closing, (iv) such assistance shall not require the giving of representations or warranties to any third parties or the indemnification thereof, (v) such assistance shall not require the waiver or amendment of any terms of this Agreement. (vi) such assistance shall not cause any director, officer or employee of the Group Companies to incur any personal liability and (vii) such assistance shall not require delivery of any legal opinions or accountants' cold comfort letters or reliance letters. Notwithstanding anything to the contrary in this Agreement, (x) the condition set forth in Section 6.2(c), as it applies to the Company's obligations under this Section 5.9(a), shall be deemed satisfied and (y) the Company shall not be deemed to have breached or failed to perform or observe any covenants, obligations or other agreements contained in this Section 5.9(a), in each case, unless the Financing has not been obtained primarily as a result of the Company's willful and material breach of its obligations under this Section 5.9(a) that cannot be or has not been cured prior to the later of (A) the Business Day prior to the Termination Date or (B) the date that is thirty (30) days from the date that the Company are notified by Parent of such willful and material breach.

- (b) Regardless of whether the Closing shall occur, Parent shall indemnify and hold harmless the Group Companies and their Affiliates and the Company's Representatives for and against any and all Losses and other obligations suffered or incurred by them in connection with the arrangement of and assistance with the Financing and any information utilized in connection therewith or otherwise arising from the Financing, including Losses and other obligations suffered in connection with complying with Section 5.9(a) above, and shall indemnify and hold harmless the Group Companies and their Affiliates and the Company's Representatives for and against all costs and expenses incurred by the Group Companies and their Affiliates and the Company's Representatives in connection with performance of their respective obligations under Section 5.9(a) or otherwise arising from the Financing. Parent shall, promptly upon request by the Company, reimburse the Group Companies and the Company's Representatives for all reasonable costs and expenses incurred by such Persons in connection with Section 5.9(a).
- 5.10 <u>Stockholder Approval</u>. Concurrently with the execution of this Agreement, the Company has delivered to Parent a written consent, duly executed by Arlington Capital Partners III, L.P., approving and adopting this Agreement, the Transaction Agreements to which the Company

is or will be a party, the Merger and the other applicable Transactions. At least 10 days prior to the Closing, the Company shall deliver to each holder of Shares a "Company Sale Notice" exercising the Company's take-along rights under Section 2.1 of the Company Stockholder Agreement.

### **ARTICLE VI**

## **CONDITIONS TO CLOSING**

- 6.1 <u>Conditions to the Obligations of the Company, Parent and Merger Sub</u>. The obligations of the Company, Parent and Merger Sub to effect the Closing and to consummate the Transactions are subject to the satisfaction (or, if permitted by applicable Law, waiver in writing by the Party for whose benefit such condition exists) of the following conditions:
- (a) any applicable waiting period under the HSR Act relating to the Transaction shall have expired or been terminated; and
- (b) there shall not be in effect any Law or Order of a Governmental Authority of competent jurisdiction in the United States directing that the Transaction not be consummated as provided herein or which has the effect of rendering it impossible or illegal to consummate the Transaction; provided, however, that Parent shall have taken all actions required by <u>Section 5.3</u> to prevent the occurrence or entry of any such Law or Order and to remove or appeal as promptly as possible any such Law or Order.
- 6.2 Other Conditions to the Obligations of Parent and Merger Sub. The obligations of Parent and Merger Sub to effect the Closing and to consummate the Transaction are subject to the satisfaction (or, if permitted by applicable Law, waiver in writing by Parent) of the following further conditions:
- (a) the Fundamental Representations shall be true and correct in all material respects on and as of the Closing Date as though made on and as of the Closing Date (other than such representations and warranties that are made on and as of a specified date, in which case such representations and warranties shall be true and correct only as of the specified date);
- (b) the representations and warranties of the Company set forth in <u>ARTICLE III</u> (other than those referred to in clause (a) above), respectively, shall be true and correct on and as of the Closing Date as though made on and as of the Closing Date (other than such representations and warranties that are made on and as of a specified date, in which case such representations and warranties shall be so true and correct only as of the specified date), except to the extent that the facts, events and circumstances that cause such representations and warranties to not be true and correct as of such dates have not had a Company Material Adverse Effect (provided that for the purposes of the foregoing clause, qualifications as to materiality and Company Material Adverse Effect contained in such representations and warranties shall not be given effect);
- (c) the Company shall have performed and complied in all material respects with all covenants required to be performed or complied with by the Company under this Agreement on or prior to the Closing Date;

- (d) since the date hereof, there shall not have occurred a Company Material Adverse Effect;
- (e) not more than five percent (5%) of the Shares outstanding immediately prior to the Effective Time shall be (i) Dissenting Shares or (ii) Shares in respect of which the holder thereof shall continue to have the right to exercise dissenters rights under the DGCL; and
- (f) prior to or at the Closing, the Company shall have delivered to Parent a certificate of an authorized officer of the Company, dated as of the Closing Date, in form and substance reasonably acceptable to Parent, to the effect that the conditions specified in Sections 6.2(a), 6.2(b), 6.2(c) and 6.2(d) have been satisfied.
- 6.3 Other Conditions to the Obligations of the Company. The obligations of the Company to effect the Closing and to consummate the Transaction are subject to the satisfaction (or, if permitted by applicable Law, waiver in writing by the Representative) of the following further conditions:
- (a) the representations and warranties of Parent and Merger Sub set forth in Section 4.1 (Organization and Power), Section 4.2 (Authorization of Agreement) and Section 4.8 (Financial Advisors) shall be true and correct in all material respects on and as of the Closing Date as though made on and as of the Closing Date (other than such representations and warranties that are made on and as of a specified date, in which case such representations and warranties shall be true and correct only as of the specified date);
- (b) the representations and warranties of Parent and Merger Sub set forth in <u>ARTICLE IV</u> (other than those referred to in clause (a) above) shall be true and correct on and as of the Closing Date as though made on and as of the Closing Date (other than such representations and warranties that are made on and as of a specified date, in which case such representations and warranties shall be true and correct only as of the specified date), except to the extent that the facts, events and circumstances that cause such representations and warranties to not be true and correct as of such dates have not had a Parent Material Adverse Effect (provided that for the purposes of the foregoing clause, qualifications as to materiality and Parent Material Adverse Effect contained in such representations and warranties shall not be given effect);
- (c) Parent and Merger Sub shall have performed and complied in all material respects with all covenants required to be performed or complied with by it under this Agreement on or prior to the Closing Date; and
- (d) prior to or at the Closing, Parent shall have delivered to the Representative a certificate of an authorized officer of Parent, dated as of the Closing Date, in form and substance reasonably acceptable to the Representative, to the effect that the conditions specified in <u>Sections 6.3(a)</u>, <u>6.3(b)</u> and <u>6.3(c)</u> have been satisfied.
- 6.4 <u>Frustration of Closing Conditions</u>. No Party may rely on the failure of any condition set forth in this <u>ARTICLE VI</u> to be satisfied if such failure was caused by such Party's failure to use best efforts to cause the Closing to occur, as required by Section 5.3.

### **ARTICLE VII**

### **TERMINATION**

#### 7.1 Termination.

- (a) This Agreement may be terminated and the Transaction may be abandoned at any time prior to the Closing:
  - (i) by mutual written consent of Parent and the Representative;
- (ii) by either the Representative or Parent, if any Governmental Authority of competent jurisdiction in the United States shall have issued an Order or taken any other action restraining, enjoining or otherwise prohibiting the Transaction (after giving effect to Parent's and Merger Sub's respective obligations under <u>Section 5.3</u>) and such Order or other action shall have become final and nonappealable;
- (iii) by either the Representative or Parent, if the Closing does not occur on or prior to February 6, 2020 (the "<u>Termination Date</u>"); <u>provided</u>, that the right to terminate this Agreement pursuant to this <u>Section 7.1(a)(iii)</u> shall not be available to any Party whose material breach of any provision of this Agreement has been the cause of, or resulted in, the failure of the Closing to occur on or before the Termination Date;
- (iv) by the Representative, upon written notice to Parent, if there shall have been a breach of any of the representations, warranties, agreements or covenants set forth in this Agreement on the part of Parent or Merger Sub or any of such representations and warranties shall have become untrue in a manner that would result in any conditions set forth in Sections 6.3(a), 6.3(b) or 6.3(c) not being satisfied prior to the Termination Date, such breach or inaccuracy has not been waived by the Representative, and the breach or inaccuracy, if capable of being cured, has not been cured within thirty (30) days following the Representative's written notice to Parent of such breach or inaccuracy or is not capable of being cured on or prior to the Termination Date; provided that the right to terminate this Agreement under this Section 7.1(a)(iv) shall not be available to the Representative if the Company is then in material breach of any representation, warranty, covenant, or other agreement contained herein;
- (v) by Parent, upon written notice to the Representative, if there shall have been a breach of any of the representations, warranties, agreements or covenants set forth in this Agreement on the part of the Company or any of such representations and warranties shall have become untrue in a manner that would result in any conditions set forth in Sections 6.2(a), 6.2(b) or 6.2(c) not being satisfied prior to the Termination Date, such breach or inaccuracy has not been waived by Parent, and the breach or inaccuracy, if capable of being cured, has not been cured within thirty (30) days following Parent's written notice to the Representative of such breach or inaccuracy or is not capable of being cured on or prior to the Termination Date; provided that the right to terminate this Agreement

under this <u>Section 7.1(a)(v)</u> shall not be available to Parent if it is then in material breach of any representation, warranty, covenant, or other agreement contained herein;

- (vi) by the Representative, whether or not the Representative or the Company has sought or is entitled to seek specific performance pursuant to <u>Section 9.10</u>, if (A) all of the conditions set forth in <u>Sections 6.1</u> and <u>6.2</u> have been satisfied or waived (other than those conditions which by their terms cannot be satisfied until the Closing and those conditions that Parent's breach has caused not to be satisfied) and (B) after receiving at least a two (2) Business Day prior written notice from the Representative, Parent fails to consummate the Transactions within two (2) Business Days following the date on which the Closing was required to have occurred pursuant to <u>Section 2.1</u>; or
- (vii) by the Parent, whether or not the Parent has sought or is entitled to seek specific performance pursuant to <u>Section 9.10</u>, if (A) all of the conditions set forth in <u>Sections 6.1</u> and <u>6.3</u> have been satisfied or waived (other than those conditions which by their terms cannot be satisfied until the Closing and those conditions that the Company's breach has caused not to be satisfied) and (B) after receiving at least a two (2) Business Day prior written notice from the Parent, the Company fails to consummate the Transactions within two (2) Business Days following the date on which the Closing was required to have occurred pursuant to <u>Section 2.1</u>.
- (b) In the event of termination by the Representative or Parent pursuant to this <u>Section 7.1</u>, written notice thereof shall forthwith be given to the other and the Transaction shall be terminated, without further action by any Party. If the Transactions are terminated as provided herein, Parent shall return to the Company or destroy all documents and other material received from the Company or the Representative relating to the Transaction, whether so obtained before or after the execution hereof.
- 7.2 Effect of Termination. If this Agreement is terminated and the Transaction is abandoned as described in Section 7.1, this Agreement shall become null and void and of no further force and effect, without any liability or obligation on the part of any Party or their respective directors, officers, employees, owners, representatives or Affiliates, and the Transaction shall be abandoned without further action by the Parties, except for (i) the penultimate sentence of Section 5.2(a) (Access to Information) and (ii) Sections 7.2 (Effect of Termination), Section 5.9(b) (Financing) and ARTICLE IX (Miscellaneous), each of which, shall survive such termination. Nothing in this Section 7.2, however, shall be deemed to release any Party from any liability for any willful breach by such Party of the terms and provisions of this Agreement prior to termination. For purposes of this Sections 7.2, "willful" shall mean a breach that is a consequence of an act undertaken by the breaching Party with the knowledge (actual or constructive) that the taking of such act would, or would be reasonably expected to, cause a breach of this Agreement.

ARTICLE VIII

TAX MATTERS

- 8.1 <u>Tax Sharing Arrangements</u>. All Tax allocation agreements, Tax sharing agreements, and Tax indemnity agreements (other than commercial Contracts entered into in the Ordinary Course of Business that do not primarily relate to Taxes) with respect to or involving any Group Company (other than the provisions of this Agreement) shall be terminated on or prior to the Closing Date and after the Closing Date, neither the Parent nor any Group Company shall not be bound thereby or have any liability thereunder for amounts due in respect of any period on or prior to the Closing Date.
- 8.2 <u>Cooperation</u>. After the Closing Date, the Representative shall (and shall cause its Affiliates to) (a) assist Parent in preparing any Tax Returns for a Pre-Closing Tax Period, and in connection therewith provide Parent necessary powers of attorney, (b) cooperate fully in preparing for and conducting any Legal Proceeding with Taxing Authorities regarding, any Tax Returns of the Group Companies for a Pre-Closing Tax Period, and (c) make available to Parent and to any Taxing Authority as reasonably requested all information, records, and documents relating to Taxes of the Group Companies. In furtherance of the foregoing, Parent and the Representative shall retain (and shall cause their Affiliates to retain) copies of all Tax Returns and related workpapers for all taxable periods that include the Closing Date and all prior taxable periods until thirty (30) days after the expiration of the applicable statute of limitations, including any extensions or waivers thereof, to which such Tax Returns relate.
- Allocation of Taxes. To the extent permissible under applicable Laws, the Parties agree to elect (and have each Group Company elect) to have each Tax year of each Group Company to end on the Closing Date and, if such election is not permitted or required in a jurisdiction with respect to a specific Tax such that each Group Company is required to file a Tax Return for a Straddle Period, to utilize the following conventions for determining the amount of Taxes attributable to the portion of the Straddle Period ending on the Closing Date: (i) in the case of property Taxes and other similar Taxes imposed on a periodic basis, the amount attributable to the portion of the Straddle Period ending on the Closing Date shall equal the Taxes for the entire Straddle Period multiplied by a fraction, the numerator of which is the number of calendar days in the portion of the period ending on the Closing Date and the denominator of which is the total number of calendar days in the entire Straddle Period; and (ii) in the case of all other Taxes (including income Taxes, sales Taxes, employment Taxes, withholding Taxes, etc.), the amount attributable to the portion of the Straddle Period ending on the Closing Date shall be determined as if each Group Company filed a separate Tax Return with respect to such Taxes for the portion of the Straddle Period ending on and as of the end of the day on the Closing Date using a "closing of the books methodology." For purposes of clause (ii), (A) any item determined on an annual or periodic basis (including amortization and depreciation deductions) shall be allocated to the portion of the Straddle Period ending on the Closing Date based on the relative number of days in such portion of the Straddle Period ending on the Closing Date as compared to the number of days in the entire Straddle Period; (B) any Transaction Deduction shall be attributed to the portion of the Straddle Period ending on the Closing Date, to the fullest extent permitted by applicable Law; and (C) any item (or Tax) resulting from a Parent Closing Date Transaction shall be attributed to the portion of the Straddle Period beginning after the Closing Date. For the avoidance of doubt, for purposes of allocating amounts required to be included by Parent or any Group Company in income under Section 951(a) or 951A of the Code with respect to any Straddle Period of a foreign Group Company, the taxable

year of the relevant foreign Group Company giving rise to the income required to be included shall be deemed to close on the Closing Date in the same manner as described above.

- 8.4 <u>338(g) Elections</u>. Neither Parent nor any Group Company shall make an election under Section 338(g) of the Code with respect to the Transactions or make any other Tax election or take any other action that has the effect of increasing the amount of Taxes attributable to a Pre-Closing Tax Period that are included in the calculation of Working Capital.
- 8.5 Other Tax Matters. The parties hereto shall, to the extent permitted or required under applicable Law, treat the Closing Date as the last day of the taxable period of the Group Companies for all Tax purposes, and Parent shall cause the Group Companies to join Parent's "consolidated group" (as defined in Treasury Regulation Section 1.1502-76(h)) effective on the day after the Closing Date.

### **ARTICLE IX**

### **MISCELLANEOUS**

#### 9.1 No Survival: Certain Waivers.

- (a) The representations and warranties and covenants and agreements (to the extent contemplating or requiring performance prior to the Closing) of the Company and the Sellers set forth in this Agreement or in any instrument delivered in connection with this Agreement (other than those contained in the Letters of Transmittal to the extent provided therein) shall not survive the Closing. Each of the representations and warranties of the Company and the Sellers set forth in this Agreement or in any instrument delivered in connection with this Agreement (other than those contained in the Letters of Transmittal to the extent provided therein) shall terminate effective immediately as of the Closing such that no claim for breach of any such representation or warranty, detrimental reliance or other right or remedy (whether in contract, in tort or at law or in equity) may be brought after the Closing with respect thereto against the Company or the Sellers except to the extent based upon or attributable to those contained in the Letters of Transmittal. The covenants and agreements of the Company, the Sellers, Parent and Merger Sub set forth in this Agreement and in any other document delivered in connection herewith to the extent contemplating or requiring performance prior to the Closing shall terminate effective immediately as of the Closing such that no claim for breach of any such covenant or agreement, detrimental reliance or other right or remedy (whether in contract, in tort, at law or in equity) may be brought after the Closing with respect thereto against the Company or the Sellers. Each covenant or agreement herein requiring performance at or after the Closing, shall, in each case, expressly survive the Closing, and nothing in this Section 9.1(a) shall be deemed to limit any rights or remedies of any Party for breach of any such surviving covenant or agreement.
- (b) Parent, for itself and on behalf of its Affiliates (including, after the Closing with respect to Parent, the Surviving Company and the Operating Subsidiaries), acknowledges and agrees that, from and after the Closing, to the fullest extent permitted under applicable Law, any and all rights, claims and causes of action it may have against any Seller or its Affiliates or any of the former, current, or future general or limited partners, shareholders or equityholders, managers,

members, directors, officers, employees, representatives or agents or any former, current or future general or limited partner, direct or indirect shareholder or equityholder, manager, member, director, officer, employee, Affiliate, representative or agent of any of the foregoing (collectively, the "Seller Related Persons") relating to the operation of the Company and the Operating Subsidiaries or their respective businesses or relating to the subject matter of this Agreement or the Company Disclosure Schedule, or Exhibits hereto or the Transactions, whether arising under, or based upon, any Law or otherwise (including any right, whether arising at law or in equity, to seek indemnification, contribution, cost recovery, damages or any other recourse or remedy, including as may arise under common law) are hereby irrevocably waived. Furthermore, without limiting the generality of this Section 9.1, no claim shall be brought or maintained by, or on behalf of, the Parent or any of its Affiliates (including, after the Closing with respect to Parent, the Surviving Company and the Operating Subsidiaries) against any Seller Related Person, and no recourse shall be sought or granted against any of them, by virtue of, or based upon, any alleged misrepresentation or inaccuracy in, or breach of, any of the representations, warranties, covenants or agreements of the Company, the Sellers or any other Person set forth or contained in this Agreement, any certificate, instrument, opinion, agreement or other document of the Company, the Sellers or any other Person delivered hereunder, the subject matter of this Agreement or the Company Disclosure Schedule or Exhibits hereto or the Transactions, the business or the ownership, operation, management, use or control of the business of the Company or any Operating Subsidiaries, any of their assets, or any actions or omissions at, or prior to, the Closing.

(c) Parent acknowledges and agrees that the agreements contained in this <u>Section 9.1</u> are an integral part of the Transaction and that, without these agreements set forth in this <u>Section 9.1</u>, the Company and the Sellers would not enter into this Agreement or otherwise agree to consummate the Transaction.

## 9.2 Expenses.

- (a) Except as otherwise provided in this Agreement or the other Transaction Agreements, each Party shall bear its own costs and expenses incurred in connection with the negotiation and execution of this Agreement and the other Transaction Agreements and each other agreement, document and instrument contemplated hereby or thereby and the consummation of the Transaction; <u>provided</u> that for the avoidance of doubt, Parent shall pay fees and other amounts required to be paid in connection with the Financing. Parent shall be solely responsible for all governmental fees and charges applicable to any requests for Governmental Approvals or to the consummation of the Transaction. Parent, on the one hand, and the Representative, on the other hand, shall each pay one-half of all charges and expenses of the Paying Agent and the Escrow Agent in connection with this Agreement.
- (b) Any Transfer Taxes payable in connection with the Transaction shall be borne fifty percent (50%) by Parent and fifty percent (50%) by Sellers (as a component of Transaction Expenses). Except as otherwise required by Law, Parent shall duly and timely prepare and file any Tax Return relating to such Taxes. Parent shall give the Representative a copy of each such Tax Return for its review and comments at least ten (10) days prior to filing and shall give the

Representative a copy of such Tax Return as filed, together with proof of payment of the Taxes shown thereon to be payable.

- 9.3 <u>Governing Law</u>. This Agreement shall be governed by and construed in accordance with the Laws of the State of Delaware without giving effect to the choice of law principles of such state that would require or permit the application of the Laws of another jurisdiction.
- 9.4 <u>Submission to Jurisdiction; Waivers</u>. Subject to <u>Section 1.10(d)(ii)</u> (which will govern any dispute arising thereunder), the Parties agree that any dispute, controversy or claim arising out of or relating to the Transaction or to this Agreement, or the validity, interpretation, breach or termination thereof, including claims seeking redress or asserting rights under any Law, shall be resolved exclusively in the state or federal courts sitting in the State of Delaware (the "<u>Delaware Courts</u>"). In that context, and without limiting the generality of the foregoing, each Party irrevocably and unconditionally:
- (a) submits for itself and its property in any action relating to the Transaction or to this Agreement, or for recognition and enforcement of any judgment in respect thereof, to the exclusive jurisdiction of the Delaware Courts, and appellate courts having jurisdiction of appeals from any of the foregoing courts, and agrees that all claims in respect of any such action shall be heard and determined in such Delaware Courts or, to the extent permitted by Law, in such appellate courts;
- (b) consents that any such action may and shall be brought exclusively in such courts and waives any objection that it may now or hereafter have to the venue or jurisdiction of any such action in any such court or that such action was brought in an inconvenient forum, and agrees not to plead or claim the same;
- (c) waives all right to trial by jury in any action (whether based on contract, tort or otherwise) arising out of or relating to the Transaction or to this Agreement, or its performance under or the enforcement of this Agreement;
- (d) agrees that service of process in any such action may be effected by mailing a copy of such process by registered or certified mail (or any substantially similar form of mail), postage prepaid, to such Party at its address as provided in <u>Section 9.8</u>; and
- (e) agrees that nothing in this Agreement shall affect the right to effect service of process in any other manner permitted by the Laws of the State of Delaware.
- 9.5 <u>Further Assurances</u>. After the Closing, each Party shall from time to time, at the request of and without further cost or expense to the other, execute and deliver such other instruments of conveyance and assumption and take such other actions as may reasonably be requested in order to more effectively consummate the Transaction.
- 9.6 <u>Entire Agreement</u>. This Agreement (including the Schedules and Exhibits hereto), the documents delivered pursuant hereto and the other Transaction Agreements represent the entire understanding and agreement between the Parties with respect to the Transaction and supersedes

all prior agreements among the Parties respecting the Transaction. The Parties have voluntarily agreed to define their rights, liabilities and obligations respecting the Transaction exclusively in contract pursuant to the express terms and provisions of this Agreement; and the Parties expressly disclaim that they are owed any duties or are entitled to any remedies not expressly set forth in this Agreement.

- 9.7 Amendments and Waivers. Prior to Closing, this Agreement can be amended, supplemented or changed, and any provision hereof can be waived, only by written instrument making specific reference to this Agreement signed by Parent and the Company. Following Closing, this Agreement can be amended, supplemented or changed, and any provision hereof can be waived, only by written instrument making specific reference to this Agreement signed by Parent and the Representative. The waiver by any Party of a breach of any provision of this Agreement shall not operate or be construed as a further or continuing waiver of such breach or as a waiver of any other or subsequent breach. No failure on the part of any Party to exercise, and no delay in exercising, any right, power or remedy hereunder shall operate as a waiver thereof, nor shall any single or partial exercise of such right, power or remedy by such Party preclude any other or further exercise thereof or the exercise of any other right, power or remedy. In the event any provision of any other Transaction Agreement shall in any way conflict with the provisions of this Agreement (except where a provision therein expressly provides that it is intended to take precedence over this Agreement), this Agreement shall control. Notwithstanding anything to the contrary contained herein, the provisions of Sections 9.10 (Specific Performance), 9.11 (No Third-Party Beneficiaries), 9.12 (Assignments), 9.7 (Amendments and Waivers) and 9.15 (No Recourse to Debt Financing Sources) that relate to any Debt Financing Sources (and any provision of this Agreement to the extent a modification, waiver or termination of such provision would directly modify the substance of the provisions of such Sections that relate to any Debt Financing Sources) may not be modified, waived or terminated in a manner that is adverse to any Debt Financing Sources without the prior written consent of such Debt Financing Sources, which consent shall not be unreasonably withheld, conditioned or delayed.
- 9.8 <u>Notices</u>. All notices and other communications under this Agreement shall be in writing and shall be deemed given (a) when delivered personally by hand, (b) when sent by facsimile or e-mail (with written confirmation of receipt), or (c) one Business Day following the day sent by overnight courier (charges prepaid), in each case at the following addresses and facsimile numbers (or to such other address or facsimile number as a Party may have specified by notice given to the other Parties pursuant to this provision).

If, prior to the Closing, the Company, to:

c/o Arlington Capital Partners 5425 Wisconsin Avenue, Suite 200 Chevy Chase, MD 20815 Attention: Peter Manos and David Wodlinger Facsimile: (202) 337-7525

Email: pmanos@arlingtoncap.com dwodlinger@arlingtoncap.com

## With a copy (which shall not constitute notice) to:

Sheppard Mullin Richter & Hampton LLP 2099 Pennsylvania Avenue, NW, Suite 100

Washington, D.C. 20006

Attention: Lucantonio N. Salvi Facsimile: (202) 747-3811

Email: lsalvi@sheppardmullin.com

## If to the Representative, to:

c/o Arlington Capital Partners 5425 Wisconsin Avenue, Suite 200 Chevy Chase, MD 20815

Attention: Peter Manos and David Wodlinger

Facsimile: (202) 337-7525

Email: pmanos@arlingtoncap.com dwodlinger@arlingtoncap.com

## With a copy (which shall not constitute notice) to:

Sheppard Mullin Richter & Hampton LLP 2099 Pennsylvania Avenue, NW, Suite 100

Washington, D.C. 20006

Attention: Lucantonio N. Salvi Facsimile: (202) 747-3811

Email: lsalvi@sheppardmullin.com

## If to Parent or Merger Sub, or, following the Closing, the Surviving Company, to:

NV5 Global Inc. 20 South Park Road

Suite 350

Attention: Richard Tong, Esq., Executive Vice President and General Counsel

Facsimile: (954) 495-2101 Email: Richard.Tong@nv5.com

## With a copy (which shall not constitute notice) to:

Loeb & Loeb LLP 345 Park Avenue

Attention: Lloyd L. Rothenberg Facsimile: (212) 656-1076 Email: lrothenberg@loeb.com 9.9 <u>Severability</u>. If any term or other provision of this Agreement is invalid, illegal, or incapable of being enforced by any Law or public policy, all other terms or provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the Transaction is not affected in any manner materially adverse to any Party. Upon such determination that any term or other provision is invalid, illegal, or incapable of being enforced, the Parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the Parties as closely as possible in an acceptable manner in order that the Transaction is consummated as originally contemplated to the greatest extent possible.

## 9.10 Specific Performance.

- (a) Each Party acknowledges and agrees that the other Parties would be irreparably damaged if any of the provisions of this Agreement are not performed in accordance with their specific terms and that any breach of this Agreement by the Company, Parent or Merger Sub could not be adequately compensated in all cases by monetary damages alone, even if available. Accordingly, in addition to any other right or remedy to which any Party may be entitled at law or in equity, before or after the Closing, each Party shall be entitled to enforce any provision of this Agreement by a decree of specific performance and to temporary, preliminary and permanent injunctive relief to prevent breaches or threatened breaches of any of the provisions of this Agreement, without posting any bond or other undertaking. Each of the Parties further agrees that it shall not object to, or take any position inconsistent with respect to, whether in a court of law or otherwise, (i) the appropriateness of the specific performance contemplated by this Section 9.10 and (ii) the exclusive jurisdiction of the courts set forth in Section 9.4 with respect to any action brought for any such remedy.
- (b) Each Party further agrees that (i) by seeking the remedies provided for in this <u>Section 9.10</u>, a Party shall not in any respect waive its right to seek any other form of relief that may be available to such Party under this Agreement or in the event that the remedies provided for in this <u>Section 9.10</u> are not available or otherwise are not granted, and (ii) nothing set forth in this <u>Section 9.10</u> shall require any Party to institute any action for (or limit any Party's right to institute any action for) specific performance under this <u>Section 9.10</u> prior or as a condition to exercising any termination right under <u>ARTICLE VII</u>, nor shall the commencement of any action pursuant to this <u>Section 9.10</u> or anything set forth in this <u>Section 9.10</u> restrict or limit any such Party's right to terminate this Agreement in accordance with <u>ARTICLE VII</u>, or pursue any other remedies under this Agreement that may be available then or thereafter.
- 9.11 <u>No Third-Party Beneficiaries; No Recourse Against Affiliates</u>. Nothing in this Agreement, express or implied, is intended or shall be construed to give any rights to any Person or entity other than (i) the Parties and their successors and permitted assigns and (ii) each D&O Indemnified Person, who shall have the right to enforce the obligations of Parent and the Company solely with respect to Section 5.7; provided that the Debt Financing Sources shall be express third-party beneficiaries of, and have the right to enforce, the provisions of Sections 9.10 (Specific Performance), 9.11 (No Third-Party Beneficiaries), 9.12 (Assignments), 9.7 (Amendments), and 9.15 (No Recourse to Debt Financing Sources) that are related to the Debt Financing Sources. No past, present or future director, officer, employee, incorporator, member, partner, stockholder,

Affiliate, agent, attorney or representative of the Company, any Operating Subsidiary, any Seller or any of its respective Affiliates shall have any liability (whether in Law or in equity or in contract or in tort) for any obligations or liabilities of the Company arising under, in connection with or related to this Agreement or for any claim based on, in respect of, or by reason of, the Transaction, including any alleged nondisclosure or misrepresentations made by any such Persons.

9.12 <u>Assignment</u>. No Party may assign or transfer this Agreement or any right, interest or obligation hereunder, directly or indirectly (by operation of Law or otherwise), without the prior written approval of Parent, on the one hand, and the Representative, on the other hand; <u>provided</u>, that each of Parent and Merger Sub may assign its rights, but not its obligations, under this Agreement to (a) any of its Affiliates or (b) the Financing sources or any other financing sources for collateral purposes; and, <u>provided</u>, further that any such assignment shall not relieve Parent or Merger Sub of its obligations hereunder. Any assignment in violation of this <u>Section 9.12</u> shall be void. Subject to the foregoing, this Agreement shall be binding upon and inure to the benefit of the Parties and their respective successors and permitted assigns.

# 9.13 <u>Authorization of Representative</u>.

By virtue of adoption of this Agreement by the Sellers, and without further action by any such Seller, the Representative is hereby appointed, authorized and empowered to act as an agent, representative and attorney-in-fact for each of the Sellers in connection with and to facilitate the consummation of the Transactions, including pursuant to the Escrow Agreement and the Paying Agent Agreement and the matters related to (w) the Merger Consideration contemplated by Section 1.10, (x) the payment of amounts from the Representative Expense Fund Amount, (v) the preparation and filing of the Tax Returns with respect to the Group Companies contemplated by ARTICLE VIII and (z) all other such items and matters set forth in this Agreement and the other Transaction Agreements contemplating participation by the Representative (collectively, "Representative Actions"), in each case with the power and authority, including power of substitution, acting in the name of or for and on behalf of each Seller, and subject to the limitations set forth herein or therein: (i) to execute and deliver and receive such waivers and consents as the Representative. in its sole discretion, may deem necessary or desirable; (ii) to collect and receive all moneys and other proceeds and property payable to the Representative from the Representative Expense Fund Amount, and, subject to any applicable withholding Laws, and net of any out-of-pocket expenses incurred by the Representative, disburse and pay the same to each Seller in accordance with such Seller's Pro Rata Share; (iii) to authorize the release of any funds from the Escrow Account in accordance with this Agreement and the Escrow Agreement; (iv) to authorize the release of any funds by the Paying Agent in accordance with this Agreement and the Paying Agent Agreement; (v) to enforce and protect the rights and interests of the Sellers and the Representative arising out of or under or in any manner relating to any Representative Action, and to take any and all actions which the Representative believes are necessary or appropriate in respect thereof, including asserting or pursuing any claim, action, proceeding or investigation (a "Claim") against Parent. Merger Sub and/or any of the Group Companies (after the Closing), consenting to, compromising or settling any such Claims, conducting negotiations with Parent, Merger Sub, the Group Companies (after the Closing) and their representatives regarding such Claims; (vi) to refrain from enforcing any right of any Seller and/or the Representative arising out of or under or in any

manner relating to any Representative Action in connection with the foregoing; provided, that no such failure to act on the part of the Representative, except as otherwise provided in this Agreement shall be deemed a waiver of any such right or interest by the Representative or by the Sellers unless such waiver is in writing signed by the waiving party or by the Representative; (vii) to make, execute, acknowledge, deliver and receive all such other agreements, guarantees, Orders, receipts, endorsements, notices, requests, instructions, certificates, stock powers, letters and other writings, and, in general, to do any and all things and to take any and all action that the Representative, in its sole and absolute discretion, may consider necessary or proper or convenient in connection with or to carry out the Representative Actions, and all other agreements, documents or instruments executed in connection therewith; and (viii) take all such other actions as the Representative shall deem necessary or appropriate, in its discretion, for the accomplishment of the foregoing and the consummation of the Transactions. The Parties acknowledge and agree that the appointment, authorization and empowerment of the Representative set forth in this Section 9.13(a) shall not include any matter specifically reserved for a Seller in this Agreement.

The Representative shall be entitled to the payment of all its out-of-pocket expenses incurred as the (b) Representative subject to and in accordance with the terms and conditions set forth in this Agreement, including Section 1.9(c), which such amounts to be used by the Representative to pay expenses incurred by the Representative in its capacity as the Representative; provided, that if the Transaction is not consummated, the Company shall reimburse the Representative for all costs and expenses reasonably incurred by the Representative in connection with the Transaction and neither Parent nor Merger Sub shall have any liability to the Representative or the Company in connection therefor. Once the Representative determines, in its sole discretion, that the Representative will not incur any additional expenses in its capacity as the Representative, then the Representative will distribute the remaining unused Representative Expense Fund Amount, if any, to the Sellers in accordance with their Pro Rata Shares. If, however, the Representative incurs expenses, in its capacity as the Representative, in an amount exceeding the Representative Expense Fund Amount, then the Representative shall be entitled to receive from the Sellers in accordance with their Pro Rata Shares an amount for the difference between the total expenses incurred by the Representative and the Representative Expense Fund Amount, Furthermore, the Representative shall be entitled to cause the Paving Agent to withhold and pay a portion of any Other Seller Payments to the Representative, by providing written notice thereof to the Paying Agent and Parent prior to its distribution of such Other Seller Payment, for the purpose of the Representative making any payments or paying any expenses under or in connection with this Agreement on behalf of the Sellers to satisfy costs, expenses and/or liabilities of the Representative in connection with the performance of its duties under this Agreement. In connection with this Agreement, and any instrument, agreement or document relating hereto or thereto, and in exercising or failing to exercise all or any of the powers conferred upon the Representative hereunder, (i) the Representative shall incur no responsibility whatsoever to any of the Sellers by reason of any error in judgment or other act or omission performed or omitted hereunder or any such other agreement, instrument or document, excepting only responsibility for any act or failure to act which represents willful misconduct, (ii) the Representative shall not be liable to Sellers for any apportionment or distribution of payments made by the Representative in good faith, and if any such apportionment or distribution is subsequently determined to have been made in error, the sole recourse of any Seller to whom payment was due, but not made or not made in full, shall be to recover from the other Sellers any

payment in excess of the amount to which such Seller is determined to have been entitled, and (iii) the Representative shall be entitled to rely on the advice of counsel, public accountants or other independent experts experienced in the matter at issue, and any error in judgment or other act or omission of the Representative pursuant to such advice shall in no event subject the Representative to liability to any of the Sellers. Each Seller shall indemnify, on a pro rata basis (based on such Seller's Pro Rata Share), the Representative against all Losses (including any and all expense whatsoever reasonably incurred in investigating, preparing or defending against any litigation, commenced or threatened or any claims whatsoever), arising out of or in connection with any claim, investigation, challenge, action or proceeding or in connection with any appeal thereof, relating to the acts or omissions of the Representative hereunder. The foregoing indemnification shall not apply in the event of any action or proceeding which finally adjudicates the liability of the Representative hereunder for its willful misconduct.

- (c) All of the indemnities, immunities and powers granted to the Representative under this Agreement shall survive the Closing Date and/or any termination of this Agreement. Each of Parent and Merger Sub shall have the right to rely upon all actions taken or omitted to be taken by the Representative pursuant to this Agreement, all of which actions or omissions shall be legally binding upon the Sellers. The grant of authority provided for herein (i) is coupled with an interest and shall be irrevocable and survives the death, incompetency, bankruptcy or liquidation of any of the Sellers, and (ii) shall survive the Closing. Any amounts received by the Representative on account of the Sellers, whether pursuant to Section 1.9 or otherwise, shall be distributed to the Sellers, net of any reserve the Representative may deem necessary in its reasonable discretion, in accordance with Section 1.11.
- (d) The Parties acknowledge and agree that the Representative shall have no liability to, and shall not be liable for any Losses of, any Party in connection with any obligations of the Representative under this Agreement or otherwise in respect of this Agreement or the Transaction.
- (e) In the event of the death, incapacity, liquidation, dissolution or resignation of any Person serving as the Representative, as applicable, within twenty (20) days of such death, incapacity, liquidation, dissolution or resignation, the Sellers shall choose the successor representative by affirmative vote of the Sellers who hold a majority of the voting power of the Company based on their Pro Rata Share. Following such resignation, any reference to the Representative herein shall be deemed to include such successor representative. The appointment of a successor Representative shall not be effective unless and until Parent receives written notice thereof; <u>provided</u> that under no circumstances shall the effectiveness of any successor Representative appointment be subject to Parent's consent.
- 9.14 <u>Attorney Conflict Waiver</u>. Recognizing that Sheppard Mullin Richter & Hampton LLP has acted as legal counsel to the Representative and its Affiliates and the Group Companies in connection with the negotiation, documentation and the consummation of the Transaction prior to the Closing (the "<u>Transaction Engagement</u>"), and that Sheppard Mullin Richter & Hampton LLP intends to act as legal counsel to the Representative and its Affiliates (which will no longer include the Group Companies) after the Closing, each of Parent and the Company hereby waives, on its

own behalf and agrees to cause its Affiliates to waive, any conflicts that may arise in connection with Sheppard Mullin Richter & Hampton LLP representing the Representative and/or its Affiliates (including any of the Sellers) after the Closing as such representation may relate to the Transaction. In addition, all communications involving attorney-client confidences between the Representative, its Affiliates or any Group Company and Sheppard Mullin Richter & Hampton LLP in the course of the negotiation, documentation and consummation of the Transaction (the "Privileged Communications") shall be deemed to be attorney-client confidences that belong solely to the Representative and its Affiliates (and not the Group Companies). Accordingly, the Group Companies shall not have access to any such Privileged Communications, or to the files of Sheppard Mullin Richter & Hampton LLP relating to the Transaction Engagement, whether or not the Closing shall have occurred. Without limiting the generality of the foregoing, upon and after the Closing, (i) the Representative and its Affiliates (and not the Group Companies) shall be the sole holders of the attorney-client privilege with respect to such engagement, and none of the Group Companies shall be a holder thereof, (ii) to the extent that files of Sheppard Mullin Richter & Hampton LLP in respect of such engagement constitute property of the client, only the Representative and its Affiliates (and not the Group Companies) shall hold such property rights and (iii) Sheppard Mullin Richter & Hampton LLP shall have no duty whatsoever to reveal or disclose any such attorney-client communications or files to any of the Group Companies by reason of any attorney-client relationship between Sheppard Mullin Richter & Hampton LLP and any of the Group Companies or otherwise. Notwithstanding the foregoing, in the event that a dispute arises between Parent or its Affiliates (including the Surviving Company), on the one hand, and a third party other than any of the Group Companies or the Sellers, on the other hand, Parent and its Affiliates (including the Surviving Company) may assert the attorney-client privilege to prevent disclosure of confidential communications to such third party; provided, however, that neither Parent nor any of its Affiliates (including the Surviving Company) may waive such privilege without the prior written consent of the Representative, which shall not be unreasonably withheld, conditioned or delayed.

9.15 No Recourse to Debt Financing Sources. In furtherance of the foregoing and without limiting the generality of Sections 9.10 (Specific Performance), 9.11 (No Third-Party Beneficiaries), 9.12 (Assignments), and 9.7 (Amendments), the parties hereto agree that, notwithstanding anything in this Agreement to the contrary, the Sellers and the Company, in each case on behalf of itself and each of its Affiliates (including each subsidiary of the Company), and each of its and their controlling persons, shareholders, partners, members, directors, officers, employees, agents, managers, representatives and successors and assigns, shall not have, and hereby waives, any rights or claims against any Debt Financing Source in connection with this Agreement, the Financing, any commitment or any transaction contemplated hereby or thereby, whether at law or equity, in contract, in tort or otherwise, and the Seller and the Company, in each case on behalf of itself and each of its Affiliates (including each subsidiary of the Company), and each of its and their controlling persons, shareholders, partners, members, directors, officers, employees, advisors, agents, attorneys, trustees, administrators, managers, representatives and successors and assigns, agrees not to commence (and if commenced agrees to dismiss or otherwise terminate) any claim, action, suit, litigation, or other proceeding (including any civil, criminal, administrative, investigative or appellate proceeding) against any Debt Financing Source in connection with this Agreement, the Financing, any commitment or any transaction contemplated hereby or thereby (including any claim, action, suit, litigation, or other proceeding (including any civil, criminal, administrative, investigative or

appellate proceeding) relating to the Financing or any commitment). Without limiting the foregoing, no Debt Financing Source shall be subject to any special, consequential, punitive or indirect damages or damages of a tortious nature to the Seller or the Company (or, in each case, any of their respective Affiliates (including the subsidiaries of the Company) or any of their respective controlling persons, shareholders, partners, members, directors, officers, employees, agents, managers, representatives and successors and assigns). "Debt Financing Sources" means (i) each of the Persons that have committed to provide, or otherwise entered into agreements in connection with, the Financing or other debt financings in connection with the transactions contemplated by this Agreement, including any joinder agreements, credit agreements or loan agreements (or other definitive documentation) relating thereto, (ii) any former, current or future Affiliate of any Person described in clause (i) above and (iii) the former, current and future partners, directors, officers, employees, agents, trustees, administrators, managers, advisors and representatives, shareholders, members, successors and assigns of any Person described in clauses (i) or (ii) above. For the avoidance of doubt, the waivers contained in this Section 9.15 shall not be deemed to extend to, and the definition of "Debt Financing Sources" shall not be deemed to include, Parent and its Affiliates.

9.16 <u>Counterparts</u>. This Agreement may be executed in one or more counterparts, including by facsimile or other electronic transmission (including e-mail), each of which shall be deemed to be an original copy of this Agreement and all of which, when taken together, shall be deemed to constitute one and the same agreement. Such delivery of counterparts shall be conclusive evidence of the intent to be bound hereby and to the extent applicable, the foregoing constitutes the election of the Parties to invoke any Law authorizing electronic signatures.

## ARTICLE X

### **DEFINITIONS AND INTERPRETATIONS**

### 10.1 Certain Definitions.

(a) For purposes of this Agreement, the following terms shall have the meanings specified in this <u>Section 10.1</u>:

"Accounts Receivable" means all accounts receivable billed and unpaid for services provided by the Group Companies prior to the Closing, net of reserves for uncollectible accounts receivable, as determined in the calculation of Working Capital.

"Accounting Referee" means BDO, provided that such Person is independent of both Parties, and if such Person is not independent of both Parties, then such Person as mutually agreed by the Parties. For purposes of this definition, "independent" means a Person who is not an Affiliate of such Person or providing services to such Person at such time and was not an Affiliate of or did not provide services to such Person at any time within the prior two (2) year period.

"Accounting Rules" means, collectively, (i) the rules, principles and sample calculation of Working Capital set forth on Exhibit B (collectively, the "Agreed Principles"), (ii) the accounting principles, methods and practices used in preparing the Company Financial Statements (collectively, the "Historical Principles") and (iii) GAAP, applied in a manner consistent with its

application to the preparation of the Company Financial Statements ("<u>Historical GAAP</u>"); <u>provided</u>, that notwithstanding any provisions or concepts of GAAP, no developments or events taking place after the Closing Date shall be taken into account; provided, further, that in the event of any conflict among the Agreed Principles, the Historical Principles and Historical GAAP, then the Agreed Principles shall take precedence, followed by the Historical Principles, followed by Historical GAAP.

"Adjustment Time" means immediately before the Effective Time without regard to the effects of the Transaction.

"Affiliate" means, with respect to any Person, any other Person that, directly or indirectly through one or more intermediaries, controls, is controlled by or is under common control with, such Person, and the term "control" (including the terms "controlled by" and "under common control with") means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through ownership of voting securities, by contract or otherwise.

"Applicable Privacy and Security Laws" means all applicable Laws (including without limitation CCPA, GDPR and the European Union Data Protection Directive 95/46/EC) and guidance issued by a Governmental Authority (i) concerning the protection, Processing, privacy and/or security of Personal Information or other confidential data, (ii) that triggers a duty to protect the rights of an individual whose Personal Information is being Processed or (iii) that triggers a duty to notify an individual whose Personal Information has been, or may have been, the subject of unauthorized access, and all regulations promulgated and guidance issued by Governmental Authorities under any of the foregoing.

"Arlington Fees" means all accrued and unpaid fees and expenses payable to Arlington Capital Partners III, L.P. as of the Closing Date (including as a result of the consummation of the Transaction) pursuant to the Management Services Agreement.

"Business Day" means any day of the year on which national banking institutions in New York are open to the public for conducting business and are not required or authorized to close.

"Cash and Cash Equivalents" means the sum of the fair market value (expressed in United States dollars) of (i) all cash and (ii) all cash equivalents (including deposits, amounts held in escrow, marketable securities and short term investments) of the Group Companies, in each case, determined in accordance with GAAP as of the Adjustment Time. Cash and Cash Equivalents shall (i) be reduced by issued but uncleared checks and drafts of the Group Companies, and (ii) be increased by inbound checks and drafts deposited for the account of the Group Companies, in each case as of the Adjustment Time.

"CCPA" means the California Consumer Privacy Act of 2018 (California Civil Code §§ 1798.100 to 1798.198), as updated or amended from time to time, and any legally binding requirements, orders, directives or decisions of any regulatory, judicial, or governmental authority in connection with the enforcement thereof.

"Closing Cash" means the aggregate amount of all Cash and Cash Equivalents of the Group Companies as of the Adjustment Time, as may be adjusted pursuant to Section 1.10.

"Closing Date Indebtedness" means all Indebtedness of the Group Companies as of the Adjustment Time, as may be adjusted pursuant to Section 1.10.

"Code" means the Internal Revenue Code of 1986.

"Commercial-Off-The-Shelf Software" means Software that was licensed from a third party on general commercial terms widely and readily available for purchase by the general public on such commercial terms, and was licensed on a non-exclusive basis for fixed payments of less than \$85,000 per year.

"Common Stock" means the common stock of the Company, par value \$0.01, per share.

"Company Charter" means that certain Amended and Restated Certificate of Incorporation of the Company, filed with the Secretary of State of the State of Delaware on August 24, 2012, as amended pursuant to that certain Certificate of Amendment of Certificate of Incorporation filed with the Secretary of State of the State of Delaware on January 16, 2014, as modified by that certain Certificate of Designation of Series B Preferred Stock, filed with the Secretary of State of the State of Delaware on May 12, 2015.

"Company Material Adverse Effect" means an event, occurrence, fact, condition or change that has (a) had a material adverse effect on the business, assets, properties, financial condition or results of operations of the Group Companies, taken as a whole, or (b) materially and adversely impaired, or would reasonably be expected to materially and adversely impair or delay, the ability of the Company, (i) to perform its obligations under the Transaction Agreements or (ii) to consummate the Transaction prior to the Termination Date; provided that no event, change, occurrence, circumstance or effect (by itself or taken together with any and all other events, changes, occurrences, circumstances or effects) that results from or arises out of or is related to any of the following shall constitute or be deemed to contribute to a "Company Material Adverse Effect", or be taken into account in determining whether a "Company Material Adverse Effect" has occurred or may, would or could occur to the extent (but only to the extent) resulting from, arising out of or related to: (i) changes in general economic conditions in the United States or any other country or region in the world, or changes in conditions in the global economy generally; (ii) changes in conditions in the financial markets, credit markets or capital markets in the United States or any other country or region in the world; (iii) changes in political conditions in the United States or any other country or region in the world, acts of war, sabotage or terrorism (including any escalation or general worsening of any such acts of war, sabotage or terrorism), earthquakes, hurricanes, tsunamis, tornadoes, floods, mudslides, wildfires or other natural disasters, weather conditions and other force majeure events, in each case in the United States or any other country or region in the world; (iv) changes affecting the industry generally in which the Group Companies operate; (v) the announcement of this Agreement, the pendency of the Transaction or any investigation or challenge to the Transaction, or the consummation of the Transaction (including but not limited to, the loss of any employees, suppliers, customers, advertisers, assets, or property interests resulting from the

identity of Parent or Merger Sub); (vi) the taking of any action required or contemplated by this Agreement or undertaken with Parent's consent pursuant to the terms of this Agreement, or the failure to take any action prohibited by this Agreement or to which Parent refused to provide consent pursuant to the terms of this Agreement; (vii) changes in Law or other legal or regulatory conditions (or the interpretation thereof); (viii) changes in GAAP or other accounting standards (or the interpretation thereof); (ix) any failure of Parent to obtain any waiver or consent from any Person required in connection with this Agreement, or (x) any matters expressly set forth in the Company Disclosure Schedule; except, in the case of clauses (i) through (iv), (vii) and (viii) above, to the extent that such event, occurrence, fact, condition or change has a disproportionate and adverse effect on the business, assets, properties, financial condition or results of operations of the Group Companies, taken as a whole, relative to the effects on such other comparable participants in the industries in which the Group Companies conduct their business.

"Company Stockholder Agreement" means that certain Stockholder Agreement by and among the Company and each Person signatory thereto, dated as of August 27, 2012, as amended as of July 24, 2013, and as amended further as of September 30, 2013.

"Company Technology" means any and all Technology used by the Group Companies in the operation of the business of the Group Companies.

"Contract" means any written agreement, commitment, undertaking, contract, indenture, note, mortgage bond, lease or license and all other written legally binding arrangements.

"<u>Data Room</u>" means the electronic documentation site established by Houlihan Lokey Capital, Inc. on behalf of the Company.

"Environmental Laws" means as enacted and in effect on or prior to the Closing Date, any Laws concerning pollution or protection of the environment or natural resources, endangered or threatened species, human health or safety (to the extent related to exposure to Hazardous Materials), or the release, treatment, storage, transportation, remediation, disposal of, exposure to, or the management, manufacture, use, containment, recycling, reclamation, reuse, treatment, generation, discharge, processing or production of Hazardous Materials.

"Equitable Principles" means (i) bankruptcy, insolvency, reorganization, moratorium and similar Laws, in each case, affecting creditors' rights and remedies generally, and (ii) general principles of equity (regardless of whether enforcement is sought in a proceeding at law or in equity).

"Equity Incentive Plan" means the Company's 2012 Equity Incentive Plan, as amended as of June 2014, or any other equity incentive plan or arrangement adopted or approved by the board of directors of the Company, and any award agreements issued pursuant thereto, in each case, as amended from time to time, and any successor equity incentive plans thereto.

"ERISA" means the Employee Retirement Income Security Act of 1974.

"Escrow Agent" means U.S. Bank National Association.

"Escrow Agreement" means that an Escrow Agreement, in a form mutually agreed by the Parties, by and among the Representative, on behalf of the Stockholders and the Optionholders, Parent and the Escrow Agent.

"Escrow Amount" means \$500,000.

"Escrow Funds" means, at any time, the portion of the Escrow Amount then remaining in the Escrow Account plus any interest accrued thereon.

"Fully Diluted Participating Shares" means the sum of (x) the aggregate number of shares of Common Stock issued and outstanding immediately prior to the Effective Time (including for these purposes the Dissenting Shares), <u>plus</u> (y) the aggregate number of shares of Common Stock issuable upon conversion of the Series A Preferred Stock issued and outstanding immediately prior to the Effective Time (including for these purposes the Dissenting Shares), <u>plus</u> (z) the aggregate number of shares of Common Stock issuable upon exercise in full of all vested in-the-money Options immediately prior to the Adjustment Time.

"<u>Fundamental Representations</u>" means the representations and warranties provided in <u>Section 3.1</u> (Organization and Power), <u>Section 3.2</u> (Authorization of Agreement), <u>Section 3.4</u> (Capitalization; Operating Subsidiaries) and <u>Section 3.21</u> (Financial Advisors).

"GAAP" means generally accepted accounting principles in the United States of America in effect from time to time.

"GDPR" means the European Union General Data Protection Regulation (EU) 2016/679.

"Government Bid" means any quotation, bid or proposal by any Group Company that, if accepted or awarded, would lead to a Contract with a Governmental Authority, including a prime contractor or a higher tier subcontractor to the United States government or any foreign government, for the design, manufacture or sale of products or the provision of services by the Company or any of its Subsidiaries.

"Government Contract" means any Contract that (i) is between the Company or any of its Subsidiaries, on the one hand, and a Governmental Authority, on the other hand, or (ii) is entered into by the Company or any of its Subsidiaries as a subcontractor (at any tier) in connection with a Contract between another Person and a Governmental Authority. For the avoidance of doubt, Contracts with public utilities, unless such public utilities are majority-owned by a Governmental Authority, shall not be deemed to be Government Contracts.

"Governmental Antitrust Authority" shall mean any Governmental Authority with regulatory jurisdiction over any consent required for the consummation of the Transaction, under the HSR Act or under Other Competition Laws.

"Governmental Authority" means any government or governmental, tribunal, judicial, administrative or (self-)regulatory body thereof, or political subdivision thereof, whether

U.S. foreign, federal, state, local, or foreign or any agency, bureau, board, commission instrumentality or authority thereof, including any state's attorney general or Supervisory Authority, or any court or arbitrator (public or private), excluding, in each case (except for purposes of the definitions of "Government Contract" and Section 3.14), Governmental Authorities in their capacities as customers of the Company or its Subsidiaries.

"Group Companies" means the Company and the Operating Subsidiaries.

"<u>Hazardous Materials</u>" means any substances, wastes, chemicals, compound, derivative or materials that are listed, regulated or defined as hazardous, toxic, pollutants, or contaminants under any Environmental Laws, including materials defined as "hazardous substances" under the Comprehensive Environmental Response, Compensation, and Liability Act, 42 U.S.C. §§ 9601 et seq., and petroleum or petroleum products (including, crude oil or any fraction thereof).

"HSR Act" shall mean the Hart-Scott-Rodino Antitrust Improvements Act of 1976, and the rules and regulations promulgated thereunder.

"Indebtedness" of any Person means, without duplication, (i) the outstanding principal amount of and accrued and unpaid interest of (A) indebtedness of such Person or its Subsidiaries for borrowed money and (B) indebtedness evidenced by notes, debentures, bonds or other similar instruments for the payment of which such Person or its Subsidiaries is responsible or liable; (ii) all obligations of such Person or its Subsidiaries issued or assumed as the deferred purchase price of property, all conditional sale obligations of such Person and all obligations of such Person under any title retention agreement (but excluding accounts payable and other current liabilities arising in the Ordinary Course of Business); (iii) obligations under any interest rate, currency swap or other hedging agreement or arrangement; (iv) capital lease obligations; (v) reimbursement obligations under any letter of credit, banker's acceptance or similar instrument; (vi) all obligations of the type referred to in clauses (i) through (v) of other Persons for the payment of which such Person or its Subsidiaries is responsible or liable, directly or indirectly, as obligor, guarantor or surety; and (vii) all obligations of the type referred to in clauses (i) through (v) of other Persons secured by any Lien on any property or asset of such Person or its Subsidiaries; provided, however, that Indebtedness shall not include (1) any amounts taken into account in the calculation of the Working Capital as of the Adjustment Time or Transaction Expenses, (2) any obligations under leases (other than capital lease obligations), or (3) any undrawn letter of credit, banker's acceptance or similar instrument.

"Intellectual Property" means any and all right, title, and interests in and to all intellectual property rights of every kind and nature however denominated, throughout the world, including rights in (i) works of authorship (whether or not published), copyrighted works, which works include all compilations, databases, computer programs (source code and object code versions), manuals and other documentation and all derivatives, translations, adaptations and combinations of the foregoing, and all applications, registrations, and renewals in connection therewith; (ii) inventions (whether or not patentable), and all patents, patent applications, and patent disclosures, together with all reissuances, continuations, continuations-in-part, divisionals, re-issues, revisions, extensions, and reexaminations thereof; (iii) trade names, trademarks, service

marks, brands, packaging designs, slogans, and trade dress, including all goodwill associated therewith, and all applications, registrations, and renewals in connection therewith; (iv) trade secrets, confidential information, mask work rights and database rights, and all other proprietary rights in Technology; (v) Internet domain names, social media accounts, rights of privacy and publicity, and moral rights; (vi) any and all registrations, applications, recordings, licenses, common-law rights, statutory rights, and contractual rights relating to any of the foregoing; and (vii) all Claims and rights to sue at law or in equity for any past or future infringement or other impairment of any of the foregoing, including the right to receive all proceeds and damages therefrom, and all rights to obtain renewals, continuations, divisions, or other extensions of legal protections pertaining thereto.

"IRS" means the United States Internal Revenue Service.

"Key Personnel" means Michael B. Shillenn, Kurt Allen, Robert VanderMeer, Jacob Hall, Jennifer Whitacre, Jonathon Wittman, Eric Merten, Robert Hickey, Anand Iyer, Mark Meade, and Russ Faux.

"Knowledge" means, with respect to the Company, the actual knowledge (after due inquiry) of Peter LaMontagne, Mark Abatto, Robert Hickey and Eric Merten.

"<u>Law</u>" means all U.S. federal, state, local or foreign laws statutes, codes, ordinances, rules, regulations, resolutions, and Orders.

"Leases" means any lease, license, sublease, sublicense, franchise, easement or other Contract pursuant to which a Person has the right to use any real, personal or intangible property. When used as a verb, the word "Lease" or "Leased" (or words having correlative meanings) means to lease, license, sublease, sublicense, obtain a franchise, acquire an easement or otherwise use any real, personal or intangible property.

"<u>Legal Proceeding</u>" means any judicial, administrative or arbitral actions, suits, proceedings, claims, actions, cause of action, lawsuit, arbitration, inquiry, demands, audits, notices of violation, litigation, or investigations of any nature, civil, criminal, administrative, regulatory or otherwise, whether at law or in equity, whether public or private, in each case, by or before a Governmental Authority or arbiter.

"Lien" means any lien, claim, encumbrance, pledge, mortgage, deed of trust, right of first refusal or other security interest or similar restriction.

"Loss" or "Losses" means all damages, losses, claims, liabilities, demands, judgments, charges, suits, awards, penalties, fines and expenses of any kind (including reasonable attorneys' and other professionals' fees and disbursements).

"<u>Management Services Agreement</u>" means that certain Management Services Agreement by and between the Company and Arlington Capital Partners III, L.P., dated as of August 27, 2012 (as may be amended from time to time).

"Optionholders" means the holders of Options as of immediately prior to cancellation in accordance with the terms of this Agreement.

"Order" means any order, injunction, judgment, decree, determination, ruling, writ, assessment or arbitration or other award of a Governmental Authority.

"Ordinary Course of Business" means the ordinary course of business of the Group Companies consistent with past practices.

"Organizational Documents" means, with respect to a particular Person (other than a natural person), the certificate/articles of formation/incorporation/organization, bylaws, partnership agreement, limited liability company agreement, trust agreement or other similar organizational document or agreement, as applicable, of such Person.

"Other Competition Laws" shall mean all non-U.S. Laws intended to prohibit, restrict or regulate actions having an anti-competitive effect or purpose, including competition, restraint of trade, anti-monopolization, merger control or antitrust Laws.

"Other Seller Payments" means any additional cash amounts (without interest) payable from time to time to the Sellers pursuant to Section 1.10(e) and Section 9.13(b) or any other funds payable to the Sellers hereunder after the Closing Date.

"<u>Parent Closing Date Transaction</u>" means any transactions or elections, including Tax elections, made on the Closing Date, after the Closing, by the Group Companies that are outside the Ordinary Course of Business.

"Parent Material Adverse Effect" means any event, change, occurrence, circumstance or effect that, when taken individually or together with all other adverse events, changes, occurrences, circumstances or effects, would, or is reasonably expected to, prevent or materially delay Parent or its Affiliates from consummating the Transaction or performing its obligations under this Agreement.

"<u>Per Participating Share Merger Consideration</u>" means an amount equal to the quotient of (x) the Merger Consideration <u>less</u> the aggregate Series A Preferred Amount <u>divided by</u> (y) the Fully Diluted Participating Shares.

"Per Participating Share Portion" means the quotient (expressed as a percentage) of (x) one (1), divided by (y) the Fully Diluted Participating Shares.

"<u>Permits</u>" means any approvals, authorizations, franchises, registrations, consents, licenses, permits or certificates and similar rights obtained or required to be obtained from a Governmental Authority.

"<u>Permitted Liens</u>" means (i) all Liens disclosed in policies of title insurance and/or recorded in public records; (ii) Liens for Taxes, assessments or other governmental charges not yet due and payable or not yet delinquent (or which may be paid without interest or penalties) or the amount or validity of which is being contested in good faith by appropriate proceedings;

(iii) mechanics', carriers', workers', repairers' and similar Liens arising or incurred in the Ordinary Course of Business or the amount or validity of which is being contested in good faith by appropriate proceedings; (iv) pledges, deposits or other Liens to the performance of bids, trade contracts (other than for borrowed money), Leases or statutory obligations (including, workers' compensation, unemployment insurance or other social security legislation, but excluding Liens for Taxes); (v) zoning, entitlement and other land use or Environmental Laws by any Governmental Authority; (vi) survey exceptions and matters as to the Real Property which would be disclosed by an accurate survey or inspection of such real property and which do not materially impair the current occupancy or current use of such Real Property; (vii) any Lien affecting the fee interest of any Real Property not created by a Group Company; (viii) title of a lessor under a capital or operating Lease; (ix) any Liens discharged or released at or in connection with Closing; and (x) such other imperfections in title, charges, covenants, conditions, easements, restrictions and encumbrances that would not be reasonably expected to materially and adversely affect the business of the Group Companies.

"Person" means any individual, corporation, partnership, firm, joint venture, association, joint-stock company, trust, unincorporated organization, Governmental Authority or other entity.

"Personal Information" means any piece of information that is considered "personally identifiable information", "personally identifiable health information," "personal data," "protected data," "non-public personal information" or any similar category of protected information or data under any Law, including any information that (i) by itself or in combination with other information, is capable of, directly or indirectly, identifying, locating or contacting a natural person; or (ii) is considered "personally identifiable information", "personal information", "personal data", "protected data," "non-public personal information" or any similar category of protected information or data under any applicable Law.

"Pre-Closing Tax Period" means (i) any taxable period ending on or before the Closing Date, and (ii) with respect to any Straddle Period, the portion of such Straddle Period ending on the Closing Date.

"Preferred Stock" has the meaning set forth in the Company Charter.

"Process", "Processed" or "Processing" means, with respect to data, the use, collection, processing (including as that term is defined under GDPR), storage, recording, organization, adaption, alteration, transfer, retrieval, consultation, disclosure, dissemination or combination of such data.

"Pro Rata Share" means, with respect to each Seller, a percentage obtained by dividing (i) the aggregate number of shares of Common Stock held by such Person immediately prior to the Effective Time (including (x) any shares of Common Stock issuable upon exercise of vested in-the-money Options granted to such Person and (y) any shares of Common Stock issuable upon conversion of the Series A Preferred Stock held by such Person), by (ii) the total number of Fully Diluted Participating Shares outstanding immediately prior to the Effective Time. The respective Pro Rata Shares of the Sellers are set forth in Exhibit D.

"R&W Insurance Policy" means a representation and warranty insurance policy between Parent and the underwriter of such policy (the "Insurer").

"Schedules" means the Company Disclosure Schedule and/or the Parent Disclosure Schedule, as the case may be.

"Securities Act" means the Securities Act of 1933.

"Sellers" means the Stockholders and the Optionholders, collectively.

"Series A Preferred Amount" means, with respect to any share of Series A Preferred Stock, an amount equal to the sum of (A) the accrued or declared but unpaid "Senior Dividends" (as such term is defined in the Company Charter) with respect to each such Share and (B) any declared but unpaid "Participating Dividends" (as such term is defined in the Company Charter) with respect to each such Share.

"Series A Preferred Stock" has the meaning set forth in the Company Charter.

"Series B Preferred Stock" has the meaning set forth in the Company Charter.

"Software" means computer software programs and software systems, including all databases, compilations, compilers, higher level or "proprietary" languages, macros, related documentation and materials, whether in source code, object code or human readable form (excluding Commercial-Off-The-Shelf Software).

"Stockholders" means the holders of Shares as of immediately prior to the Effective Time.

"Straddle Period" means any taxable period beginning on or before the Closing Date and ending after the Closing Date.

"Subsidiary" means any Person of which a majority of the outstanding share capital, voting securities or other equity interests is owned, directly or indirectly, by another Person.

"Supervisory Authority" has the meaning ascribed to that term in the GDPR.

"Surrendered Certificate(s)" means one or more stock certificates representing the applicable Shares surrendered in accordance with this Agreement or, if any stock certificate representing the Shares has been lost, stolen or destroyed, compliance with Section 1.13(b) with respect to such applicable Shares.

"<u>Tax</u>" or "<u>Taxes</u>" means any U.S. federal, state, local or foreign taxes, including all net income, gross receipts, license, payroll, employment, escheat, excise, severance, stamp, occupation, premium, windfall profits, environmental, customs duties, equity, franchise, profits, withholding, social security, employment, unemployment, disability, real property, personal property, intangible property, sales, use, transfer, registration, recording, documentary, occupancy, occupation value added, alternative or add-on minimum, or other tax of any kind whatsoever,

including any interest, addition to tax, penalty or addition thereto imposed by a Taxing Authority, including any liability therefor as a result of being a member of a group, as a transferee, as a result of a Tax allocation agreement, Tax sharing agreement, or Tax indemnity agreement.

"<u>Tax Return</u>" means any return, declaration, report, claim for refund or information return or statement or attachment thereto, and including any amendment thereof required to be filed with a Taxing Authority in respect of any Taxes.

"<u>Taxing Authority</u>" means the IRS or any U.S. federal, state, local or foreign Governmental Authority having or purporting to exercise jurisdiction with respect to any Tax.

"Technology" means all inventions, works, discoveries, innovations, know-how, information (including ideas, research and development, formulas, algorithms, compositions, processes and techniques, methodologies, data, designs, drawings, schematics, blueprints, specifications, customer and supplier lists, pricing and cost information, business and marketing plans and proposals, graphics, flow charts, models, prototypes, illustrations, artwork, documentation, and manuals), databases, computer software, firmware, computer hardware, electronic, electrical, and mechanical equipment, and all other forms of technology, including improvements, modifications, works in process, derivatives, or changes, whether tangible or intangible, embodied in any form, whether or not protectable or protected by patent, copyright, trade secret law, or otherwise, and all documents and other materials recording any of the foregoing.

"Transaction(s)" means the transactions contemplated by this Agreement and the other Transaction Agreements.

"<u>Transaction Agreements</u>" means this Agreement, the Escrow Agreement and each other agreement, document, instrument or certificate contemplated by this Agreement to which Parent, Merger Sub or the Company is a party or to be executed by Parent, Merger Sub or the Company in connection with the consummation of the Transaction.

"Transaction Deduction" means any deduction permitted for income Tax purposes attributable to any payments (or deemed payments) of (i) Transaction Expenses or other similar expenses paid on, prior to or after the Closing Date or included in the computation of the Working Capital; (ii) the payment of the Merger Consideration with respect to the Options (and related employer portion of employment Taxes); (iii) the exercise of an Option between (and including) the date hereof and the Closing Date (and related employer portion of employment Taxes); (iv) the vesting of any Securities in connection with the Transaction; and (v) any fees, expenses and interest (including amounts treated as interest for income Tax purposes) that were included in the computation of the Working Capital.

"Transaction Expenses" means, without duplication and only to the extent not paid as of the Adjustment Time, as may be adjusted pursuant to Section 1.10, the collective amount of all (i) out-of-pocket costs and expenses incurred by the Group Companies in connection with the Transaction payable by the Group Companies to outside legal counsel, accountants, advisors, brokers and other third parties, including, without limitation, the Arlington Fees and (ii) transaction bonuses, change of control or similar payments (other than those arising from actions of Parent

taken after the Closing) that are due to any current or former employee, officer or director of the Group Companies directly as a result of the consummation of the Transaction pursuant to any Contract entered into by any Group Company prior to the Closing; provided, however, that Transaction Expenses shall not include (1) any amounts taken into account in the calculation of the Closing Date Indebtedness or (2) any prepayment penalties, redemption premiums, call premiums, make-whole payments or similar fees, costs, expenses and/or penalties incurred in relation to the payment of any Indebtedness.

"<u>Transfer Tax</u>" means any sales, use, transfer, registration, recording, documentary or other similar Tax payable in connection with the Transactions.

"Working Capital" means, with respect to the Group Companies, on a consolidated basis, (i) current assets of the Group Companies, as of the Adjustment Time, that are included in the line item categories of current assets specifically identified on Exhibit B reduced by (ii) those current liabilities of the Group Companies, as of the Adjustment Time, that are included in the line item categories of current liabilities specifically identified on Exhibit B in each case, without duplication, and as determined in a manner strictly consistent with the Accounting Rules, as may be adjusted pursuant to Section 1.10. Notwithstanding anything to the contrary contained herein, in no event shall "Working Capital" include any amounts with respect to (A) any fees, expenses or liabilities related to any financing by Parent and its Affiliates of the Transaction, (B) any intercompany accounts and transactions between or among the Group Companies, (C) any Transaction Expenses, Closing Date Indebtedness or Closing Cash, (D) any liabilities of the Group Companies or any of their respective Affiliates which are being discharged, terminated or cancelled pursuant to Section 1.9, or (E) any prepayment penalties, redemption premiums, call premiums, make-whole payments or similar fees, costs, expenses and/or penalties incurred in relation to the repayment of any Indebtedness. For purposes of this definition, including the calculation of current assets and current liabilities, the Parties shall disregard any adjustments arising from purchase accounting or otherwise arising out of the Transaction.

"Working Capital Target" means \$19,882,174.

"Work-in-Progress" means services provided by the Group Companies prior to the Closing Date for which payment invoices have not been remitted to the applicable customer as of the Closing Date.

(b) <u>Terms Defined Elsewhere in this Agreement</u>. For purposes of this Agreement, the following terms have the meanings set forth in the sections indicated:

Access Limitations Section 5.2(a)
Agreed Principles Section 10.1(a)
Agreement Preamble
Backlog Contracts Section 3.25(a)
Balance Sheet Section 3.5(a)(ii)
Balance Sheet Date Section 3.5(a)(ii)
Base Price Section 1.8
Benefit Plan(s) Section 3.17(a)

Business Systems Section 3.22(b)

Certificate of Merger Section 2.2

Claim Section 9.13(a)

Closing Section 2.1

Closing Date Section 2.1

Closing Date Merger Consideration Section 1.10(c)

Closing Payment Section 1.9

Closing Statement Section 1.10(c)

Company Preamble

Company Disclosure Schedule ARTICLE III

Company Financial Statements Section 3.5(a)

Company Intellectual Property Section 3.16(a)

Company Software Section 3.16(c)

Company's Representative Section 5.9(a)

Confidentiality Agreement Section 5.4

Delaware Courts Section 9.4

DGCL Recitals

Dispute Notice Section 1.10(d)(i)

Disputed Items Section 1.10(d)(i)

Dissenters' Rights Statute Section 1.6

Dissenting Shares Section 1.6

DOJ Section 5.3(b)

D&O Expenses Section 5.7(a)

D&O Indemnifiable Claim Section 5.7(a)

D&O Indemnified Person Section 5.7(a)

Effective Time Section 2.2

Environmental Permits Section 3.12(a)(ii)

Escrow Account Section 1.9(b)

Estimated Merger Consideration Section 1.10(a)

Final Closing Date Merger Consideration Section 1.10(e)

Financing Section 5.9(a)

FTC Section 5.3(b)

Governmental Approval Section 3.3(b)

Historical GAAP Section 10.1(a)

Historical Principles Section 10.1(a)

Increase Amount Section 1.10(e)(i)

Insurance Policies Section 3.20

Insurer R&W Insurance Policy

Increase Amount Section 1.10(e)(i)

Lease(d) Section 10.1(a)

Leased Real Property Section 3.11(b)

Letter of Transmittal Section 1.13

Material Contracts Section 3.13(a)

Material Government Contracts(s) Section 3.14(a)

Merger Recitals

Merger Consideration Section 1.8

Merger Sub Preamble

Open Source Materials Section 3.16(a)

Operating Subsidiaries Section 3.4(b)

Option(s) Recitals

Outbound IP Contractual Obligations Section 3.16(f)

Owned Intellectual Property Section 3.16(a)

Owned Real Property Section 3.11(a)

Parent Preamble

Parent Disclosure Schedules ARTICLE IV

Parties Preamble

Party Preamble

Paying Agent Section 1.9(e)

Paying Agent Agreement Section 2.3(e)

Pre-Closing Statement Section 1.10

Privileged Communications Section 9.14

Real Property Section 3.11(b)

Real Property Lease Section 3.11(b)

Referred Disputed Items Section 1.10(d)(ii)

Related Party Section 3.19

Representative Preamble

Representative Actions Section 9.13(a)

Representative Expense Fund Amount Section 1.9(c)

Scheduled IP Section 3.16(b)

Securities Recitals

Seller Related Persons Section 9.1(b)

Share(s) Recitals

Surviving Company Recitals

Termination Date Section 7.1(a)(iii)

Transaction Engagement Section 9.14

- 10.2 <u>Certain Interpretive Matters</u>. Unless otherwise expressly provided, for purposes of this Agreement, the following rules of interpretation shall apply:
- (a) <u>Time of the Essence; Calculation of Time Periods</u>. Time is of the essence for each and every provision of this Agreement. When calculating the period of time before which, within which or following which any act is to be done or step taken pursuant to this Agreement, the date that is the reference date in calculating such period shall be excluded. If the last day of such period is a non-Business Day, the period in question shall end on the next succeeding Business Day. Any reference in this Agreement to a "day" (without explicit qualification as a Business Day) means a calendar day.
  - (b) <u>Dollars</u>. Any reference in this Agreement to "\$" or dollars shall mean U.S. dollars.

- (c) Exhibits/Schedules/Construction; Agreements. The exhibits and Schedules to this Agreement are an integral part of this Agreement and are hereby incorporated herein and made a part hereof as if set forth herein. Any capitalized terms used in any Schedule or exhibit but not otherwise defined therein shall be defined as set forth in this Agreement. If a subject matter is addressed in more than one representation and warranty in this Agreement, each of Parent and Merger Sub shall be entitled to rely only on the most specific representation and warranty addressing such matter. Any disclosure set forth in one section of the Schedules shall apply to (i) the representations and warranties or covenants contained in the Section of this Agreement to which it corresponds in number, (ii) any representation and warranty or covenant to which it is referred by cross reference, and (iii) any other representation or warranty or covenant to the extent it is reasonably apparent from the wording of such disclosure that such disclosure is applicable to such representation or warranty or covenant. The inclusion of any information in the Schedules shall not (i) be deemed to be an admission or acknowledgment, in and of itself, that such information is required by the terms hereof to be disclosed, is material, has resulted in or would reasonably be expected to have a Company Material Adverse Effect or Parent Material Adverse Effect, or is outside the Ordinary Course of Business, (ii) be deemed or interpreted to expand the scope of any representations and warranties, obligations, covenants, conditions or agreements contained herein, or (iii) constitute, or be deemed to constitute, an admission to any third party concerning such information. No disclosure in the Schedules relating to any possible breach or violation of any Contract or Law will be construed as an admission or indication that any such breach or violation exists or has actually occurred. Capitalized terms used in the Schedules and not otherwise defined therein have the meanings given to them in this Agreement. Any reference in this Agreement, unless the context would require otherwise, to an agreement, instrument or other document means such agreement, instrument or other document as amended, supplemented and modified from time to time to the extent permitted by the provisions thereof. Any reference to a Law in this Agreement, unless the context would require otherwise, means such Law as amended from time to time and includes any successor legislation thereto and any rules or regulations promulgated thereunder.
- (d) <u>Gender and Number</u>. Any reference in this Agreement to gender shall include all genders, and words imparting the singular number only shall include the plural and vice versa.
- (e) <u>Headings</u>. The provision of the Table of Contents, the division of this Agreement into Articles, Sections and other subdivisions and the insertion of headings are for convenience of reference only and shall not affect or be utilized in construing or interpreting this Agreement. All references in this Agreement to any "Article," "Section" or other subdivision are to the corresponding Article, Section or other subdivision of this Agreement unless otherwise specified.
- (f) <u>Herein</u>. The words such as "herein," "hereinafter," "hereof," "hereunder" and "hereto" refer to this Agreement as a whole and not merely to a subdivision in which such words appear unless the context otherwise requires.
- (g) <u>Including</u>. The word "including" or any variation thereof means "including, without limitation" and shall not be construed to limit any general statement that it follows to the specific or similar items or matters immediately following it.

- (h) <u>Made Available</u>. An item shall be considered "made available" to a Party hereto, to the extent such phrase appears in this Agreement, only if such item has been provided via electronic mail from Houlihan Lokey to Parent or its representatives or posted by the Company or its representatives in the Data Room.
- (i) <u>Reflected On or Set Forth In.</u> An item arising with respect to a specific representation or warranty shall be deemed to be "reflected on" or "set forth in" a balance sheet or financial statements, to the extent any such phrase appears in such representation or warranty, if (i) there is a reserve, accrual or other similar item underlying a number on such balance sheet or financial statements that related to the subject matter of such representation, (ii) such item is otherwise specifically set forth on the balance sheet or financial statements, or (iii) such item is reflected on the balance sheet or financial statements and is specifically set forth in the notes thereto.
- (j) <u>Accounting Terms</u>. All accounting terms used in this Agreement shall have the meanings or interpretation given to them in accordance with GAAP, unless the context would require otherwise.
- (k) Material. As used in this Agreement, unless the context would require otherwise, the term "material" and the concept of the "material" nature of an effect upon the Group Companies or their business shall be measured relative to the Group Companies, taken as a whole, as their business is currently being conducted. There have been included in the Company Disclosure Schedule or the Parent Disclosure Schedule and may be included elsewhere in this Agreement items which are not "material" within the meaning of the immediately preceding sentence for informational purposes and in order to avoid any misunderstanding, and such inclusion shall not be deemed to be an agreement by the Parties that such items are "material" or to further define the meaning of such term for purposes of this Agreement.
- (l) <u>Joint Negotiation and Drafting</u>. The Parties have participated jointly in the negotiation and drafting of this Agreement and, in the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as jointly drafted by the Parties and no presumption or burden of proof shall arise favoring or disfavoring any Party by virtue of the authorship of any provision of this Agreement.

[Remainder of Page Intentionally Left Blank]

**IN WITNESS WHEREOF**, the Parties hereto have caused this Agreement to be executed by their respective officers thereunto duly authorized, as of the date first written above.

#### **PARENT:**

# **NV5 GLOBAL INC.**

By: /s/ Dickerson C. Wright

Name: Dickerson C. Wright Title: Chief Executive Officer

# **MERGER SUB:**

# **NV5 MERGER SUB CORP.**

By: /s/ Dickerson C. Wright

Name: Dickerson C. Wright Title: Chief Executive Officer

# THE COMPANY:

#### GEOSPATIAL HOLDINGS INC.

By: [redacted]

Name: Title:

# **REPRESENTATIVE:**

# ARLINGTON CAPITAL PARTNERS III, L.P.

By: Arlington Capital Group III, LLC, its general partner

By: [redacted]

Name: Title:



November 6, 2019

NV5 Global, Inc. 200 South Park Road, Suite 350 Hollywood, Florida 33021 Attention: Edward Codispoti, Chief Financial Officer

Re: Credit Agreement dated as of December 7, 2016 (as amended prior to the date hereof, the "*Existing Credit Agreement*") by and among NV5 Global, Inc., a Delaware corporation ("*you*" or the "*Company*"), the Guarantors party thereto, the Lenders party thereto and Bank of America, N.A., as Administrative Agent

#### Ladies and Gentlemen:

Capitalized terms used herein and not defined herein shall have the meanings assigned thereto in the Existing Credit Agreement.

You have advised Bank of America, N.A. (through itself or one of its designated affiliates or branch offices, "Bank of America"), BofA Securities, Inc. (or any of its designated affiliates, "BofA Securities"), PNC Bank, National Association ("PNC") and PNC Capital Markets LLC ("PNC Capital Markets") that you intend to acquire (the "Target Acquisition"), through a merger transaction directly and/or with one or more of your direct or indirect wholly-owned Subsidiaries ("Merger Sub"), all of the outstanding capital stock of Geospatial Holdings Inc. (the "Target"), from the Sellers (as defined in the Target Acquisition Agreement (as defined in Exhibit C to this letter agreement)), pursuant to the Target Acquisition Agreement.

In connection with the foregoing, you have also advised Bank of America, BofA Securities, PNC and PNC Capital Markets (collectively, the "Commitment Parties") that you desire to:

(a) finance the payment of the cash consideration for the Target Acquisition, repay existing bank facility indebtedness of the Target and its subsidiaries, and pay fees and expenses incurred in connection with the Transaction (defined below) using proceeds from (i) a Term Facility incurred under the Amended Credit Agreement (defined below) in the principal amount of \$150 million (the "Term Facility"), which Term Facility shall be have pricing, tenor, amortization and such other terms and conditions as set forth in Exhibit A, and (ii) up to \$150 million of Loans under the Revolving Facility under the Amended Credit Agreement (such amount, the "Acquisition Revolver Draw" and together with the Term Facility, the "Acquisition Funds"), which Revolving Facility will be increased under Section 2.16 of the Amended Credit Agreement by an aggregate principal amount of \$90 million (the "Revolving Facility Increase" and together with the Term Facility, the "Incremental Facilities") so that the maximum aggregate principal amount of the Revolving Facility under the Amended Credit Agreement shall be \$215 million; and

122724333\_8

(b) effect an amendment (the "*Proposed Amendment*") to the Existing Credit Agreement (as amended by the Proposed Amendment, the "*Amended Credit Agreement*"), which shall be effective on the Effective Date (defined below), that contains modifications to the Existing Credit Agreement as set forth in <a href="Exhibit B">Exhibit B</a>;

The revolving credit and term loan facilities under the Amended Credit Agreement (including the Incremental Facilities) are referred to herein as the "Amended Credit Facilities". The Target Acquisition, entering into the Proposed Amendment, the documentation and funding of the Incremental Facilities, the repayment of existing third party bank indebtedness of the Target and its subsidiaries, the payment of fees and expenses incurred in connection with the Target Acquisition and the Amended Credit Facilities, and all related transactions, are referred to collectively herein as the "Transaction". This letter and the exhibits hereto are collectively referred to as the "Commitment Letter".

- 1. <u>Commitments</u>. In connection with the foregoing:
  - (a) Bank of America is pleased to provide its commitment to provide 56.1% of the principal amount of the Incremental Facilities and PNC is pleased to provide its commitment to fund 43.9% of the principal amount of the Incremental Facilities (in each case, on a pro rata basis between the Term Facility and the Revolving Facility Increase);
    - (b) each of Bank of America and PNC is pleased to advise you of its willingness to consent to the Proposed Amendment;
  - (c) each of BofA Securities and PNC Capital Markets, as a joint lead arranger and joint bookrunner (collectively in such capacities, the "*Lead Arrangers*") for the Amended Credit Facilities, is pleased to advise you of its willingness to form a syndicate of financial institutions for the Incremental Facilities (such financial institutions, together with Bank of America, PNC and the lenders under the Existing Credit Agreement, the "*Lenders*");
    - (d) Bank of America will continue to act as sole Administrative Agent for the Amended Credit Facilities; and
  - (e) you hereby accept and agree to the foregoing commitments and consents, and appoint BofA Securities, Bank of America, PNC Capital Markets and PNC to the roles set forth above.

The commitments of Bank of America and PNC hereunder are several and not joint

No additional agents, co-agents, arrangers or bookrunners will be appointed and no other titles will be awarded without the consent of the Lead Arrangers. BofA Securities shall have "left side" designation and shall appear on the top left of any Information Materials (as defined below) and all other offering or marketing materials in respect of the Amended Credit Facilities and shall perform the duties and exercise the authority customarily performed and exercised by it in such role.

2. <u>Conditions Precedent</u>. The commitments of Bank of America and PNC hereunder and the undertakings of the Lead Arrangers to provide the services described herein are subject only to the satisfaction of each of the conditions precedent set forth in <u>Exhibit C</u> attached hereto and forming part of this Commitment Letter.

Notwithstanding anything in this Commitment Letter, the Fee Letter (as defined below) or the definitive documentation for the Amended Credit Facilities to the contrary, (a) the only representations and warranties the accuracy of which shall be a condition to the funding of the Acquisition Funds on the Effective Date shall be the Specified Representations (as defined below) and the Target Acquisition Agreement Representations (as defined in Exhibit C), and (b) the terms of the definitive documentation for the Amended Credit Facilities shall be in a form such that they do not impair the availability or funding of the Acquisition Funds on the Effective Date if the conditions set forth in Exhibit C are satisfied (it being understood that, to the extent any collateral (including the grant or perfection of any security interest) is not or cannot be provided and/or perfected on the Effective Date (other than the grant and perfection of security interests in assets with respect to which a lien may be perfected solely by the filing of a financing statement (other than as a "fixture filing") under the Uniform Commercial Code ("UCC"), then the provision and perfection of such collateral shall not constitute a condition precedent to the availability and initial funding of the Acquisition Funds on the Effective Date, but may instead be provided within a time period to be agreed by the Administrative Agent pursuant to arrangements to be mutually agreed by the Administrative Agent and you. For purposes hereof, "Specified Representations" means the representations and warranties to be made by the Company and the Guarantors in the definitive documentation for the Amended Credit Facilities relating to (a) corporate or other organizational existence of the Company and the Guarantors, (b) organizational power and authority of the Company and the Guarantors to enter into and perform the definitive documentation for the Amended Credit Facilities, (c) due authorization, execution and delivery by the Company and the Guarantors of, and enforceability against the Company and the Guarantors of, the definitive documentation for the Amended Credit Facilities, (d) creation, validity and perfection of first priority (subject to permitted liens) liens under the security documents (subject, without limitation, to the limitations and exceptions set forth in the preceding sentence), (e) no conflicts of the definitive documentation for the Amended Credit Facilities with, or consents required under, the organizational documents of the Company and the Guarantors, material laws applicable to the Company or any of the Guarantors or material agreements (other than consents that have been obtained), (f) use of proceeds not violating margin regulations, (g) not violating anti-corruption laws, sanctions and the PATRIOT Act, (h) Investment Company Act, (i) absence of litigation with respect to the Amended Credit Facilities. (i) accuracy of financial statements in all material respects and (k) solvency as of the Effective Date (after giving effect to the Transaction) of the Company and its subsidiaries on a consolidated basis. This paragraph is referred to herein as the "Limited Conditionality Provision".

3. Syndication. The Lead Arrangers intend to commence syndication of the Amended Credit Facilities promptly upon your acceptance of this Commitment Letter and the Fee Letter. You agree to actively assist the Lead Arrangers in achieving a syndication of the Amended Credit Facilities that is satisfactory to the Lead Arrangers. Such assistance shall include your (a) providing and causing your advisors to provide (and using your commercially reasonable efforts to cause the Target and its advisors to provide) the Commitment Parties and the other Lenders upon request with all information reasonably deemed necessary by the Commitment Parties to complete syndication, including, but not limited to, information and evaluations prepared by you, the Target and your or its advisors, or on your or its behalf, relating to the transactions contemplated hereby (including the Projections (as hereinafter defined), the "Information"), (b) assisting in the preparation of materials to be used in connection with the syndication of the Amended Credit Facilities (collectively, the "Information Materials"), (c) using your commercially reasonable efforts to ensure that the syndication efforts of the Lead Arrangers benefit materially from your existing banking relationships and (d) otherwise reasonably assisting the Commitment Parties in their syndication efforts, including by making your officers and advisors available (and your using commercially reasonable efforts to make the officers and advisors of the Target available) from time to time to attend and make presentations regarding the business and prospects of the Company and its subsidiaries, the Target and its subsidiaries and the Transaction, as appropriate, at one or more meetings of prospective Lenders. You hereby agree that, following the Effective Date and at our own expense, we may place advertisements in financial and other newspapers

and periodicals, and circulate similar promotional materials, in the form of a "tombstone" or otherwise, containing information customarily included in such advertisements and materials, including (i) the name of the Company and its subsidiaries, (ii) our and our affiliates' titles and roles in connection with the Amended Credit Facilities, and (iii) the amount, type and closing date of the Amended Credit Facilities. You also authorize each of the Lead Arrangers and its affiliates to download copies of the Company's and its subsidiaries' logos from its website, use copies thereof in SyndTrak or similar workspaces established by BofA Securities in connection with the syndication of the Amended Credit Facilities and use such logos on any confidential information memoranda, presentations and other marketing materials prepared in connection with the syndication of the Amended Credit Facilities.

It is understood and agreed that the Lead Arrangers will manage and control all aspects of the syndication in consultation with you, including decisions as to the selection of prospective Lenders and any titles offered to proposed Lenders, when commitments will be accepted and the final allocations of the commitments among the Lenders (it being understood that the syndication of the Incremental Facilities shall be allocated 56.1% to reduce the commitment of Bank of America and 43.9% to reduce the commitment of PNC until a Successful Syndication (as defined in the Fee Letter) has occurred). It is understood that no Lender participating in the Incremental Facilities will receive compensation from you in order to obtain its commitment, except on the terms contained herein and in the Fee Letter. It is also understood and agreed that the amount and distribution of the fees among the Lenders will be at the sole and absolute discretion of the Lead Arrangers.

Notwithstanding any other provision of this Commitment Letter to the contrary or any syndication, assignment or other transfer of Bank of America's or PNC's commitments in respect of the Incremental Facilities herein, neither Bank of America nor PNC shall be relieved, released or novated from its obligations hereunder (including its obligation to fund its commitment to the Incremental Facilities on the Effective Date) as a result of such syndication, assignment or other transfer until after the funding of the Incremental Facilities (or any portion thereof) on the Effective Date.

Without limiting your obligations to assist with syndication efforts as set forth herein, it is understood that our commitments with respect to the Incremental Facilities are not conditioned upon the syndication of, or receipt of commitments in respect of, the Incremental Facilities and in no event shall the commencement or successful completion of syndication of the Incremental Facilities nor the obligation to assist with syndication efforts as set forth herein (including, without limitation, any of your agreements in this paragraph or the following paragraph), constitute a condition to the commitment hereunder to fund the Incremental Facilities on the Effective Date.

4. <u>Information Requirements</u>. You represent, warrant and covenant (subject, however, to the best of your knowledge with respect to Information relating to the Target) that (a) all financial projections concerning the Company and its subsidiaries and the Target and its subsidiaries that have been or are hereafter made available to any of the Commitment Parties or the Lenders by you, the Target or any of your or its representatives (or on your or its behalf) (the "*Projections*") have been or will be prepared in good faith based upon assumptions believed by you to be reasonable at the time furnished (it being understood that such Projections are as to future events and are not to be viewed as facts, that actual results during the period or periods covered by any such Projections may differ significantly from the projected results and that such differences may be material, that such Projections are subject to significant uncertainties and contingencies many of which are beyond your control, and that no assurance can be given that the projected results will be realized) and (b) all Information, other than Projections, which has been or is hereafter made available to any of the Commitment Parties or the Lenders by you, the Target or any of your or its representatives (or on your or its behalf) in connection with any aspect of the Transaction, as and when furnished, is and will

be complete and correct in all material respects and does not and will not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements contained therein not misleading. You agree to furnish us with further and supplemental information from time to time until the effectiveness of the Proposed Amendment and funding of the Incremental Facilities (the "Effective Date") and, if requested by BofA Securities or PNC, for such period thereafter as is necessary to achieve a Successful Syndication (as defined in the Fee Letter) so that such representation and warranty is correct on the Effective Date and on such later date on which a Successful Syndication occurs as if the Information were being furnished, and such representation and warranty were being made, on such date. In issuing this commitment and in arranging and syndicating the Amended Credit Facilities, the Commitment Parties are and will be using and relying on the Information without independent verification thereof.

You acknowledge that the Commitment Parties on your behalf will make available Information Materials to the proposed syndicate of Lenders by posting the Information Materials on SyndTrak or another similar electronic system. In connection with the syndication of the Amended Credit Facilities, unless the parties hereto otherwise agree in writing, you shall be under no obligation to provide Information Materials suitable for distribution to any prospective Lender (each, a "Public Lender") that has personnel who do not wish to receive material non-public information (within the meaning of the United States federal securities laws, "MNPF") with respect to the Company or its affiliates, the Target and its affiliates or the respective securities of any of the foregoing. You agree, however, that the definitive documentation for the Amended Credit Facilities will contain provisions concerning Information Materials to be provided to Public Lenders and the absence of MNPI therefrom. Prior to distribution of Information Materials to prospective Lenders, you shall provide us with a customary letter authorizing the dissemination thereof. In any event, the Commitment Parties shall not disclose to any prospective Lender any Information Materials which are confidential unless such prospective Lender has agreed (whether by execution and delivery of a confidentiality agreement or by agreement to terms and conditions required to access information via electronic means) to keep such information confidential on terms which are customary in the syndication market for transactions of the nature described herein.

Confidentiality. Each of the Commitment Parties and their respective affiliates will use all confidential information provided to them or such affiliates by or on behalf of you hereunder (including, without limitation, the Projections, the information contained in the Information Materials and any other Information provided by you that is not identified or marked clearly by you as "PUBLIC") solely for the purpose of providing the services which are the subject of this Commitment Letter in connection with the transactions contemplated hereby and shall treat confidentially all such information; provided that nothing herein shall prevent the Commitment Parties from disclosing any such information (a) pursuant to the order of any court or administrative agency or in any pending legal or administrative proceeding, or otherwise as required by applicable law or compulsory legal process (in which case the applicable Commitment Party, to the extent permitted by law, will use commercially reasonable efforts to notify you prior to any such disclosure to the extent not prohibited by law, rule or regulation and shall comply with any applicable protective order (or equivalent) obtained by you or on your behalf and, in the event no such order (or equivalent) is obtained, shall use commercially reasonable efforts to disclose only such of the confidential information as they are advised by counsel is necessary to comply with applicable law, regulation or court or administrative agency order), (b) upon the request or demand of any regulatory authority having jurisdiction over any of the Commitment Parties or any of their respective affiliates, (c) to the extent that such information becomes publicly available other than by reason of disclosure by any of the Commitment Parties or any of their respective affiliates or Representatives (as defined below) in contravention of this paragraph, (d) to the extent that such information is received by any of the Commitment Parties from a third party that is not to its knowledge subject to confidentiality obligations to you, (e) to the extent that such information is independently developed by any of the Commitment Parties, (f) to any of the Commitment Parties respective

affiliates and their respective employees, legal counsel, independent auditors and other experts or agents on a need-to-know basis in connection with the Amended Credit Facilities and the transactions contemplated hereby (collectively, the "*Representatives*"), and provided such persons and entities are informed of the confidential nature of such information and instructed to keep such information confidential, (g) to prospective Lenders and prospective arrangers, participants or assignees in respect of the Amended Credit Facilities or potential counterparties to any swap or derivative transaction relating to the Company or any of its subsidiaries or any of their respective obligations, in each case who agree to be bound by standard confidentiality provisions in accordance with the customary loan market syndication practices for dissemination of such information, (h) for purposes of establishing a "due diligence" defense or (i) in connection with the exercise of any remedies hereunder or under the Fee Letter or any action or proceeding relating to this Commitment Letter or the Fee Letter or the enforcement of rights hereunder or thereunder.

6. Expenses and Indemnification. By executing this Commitment Letter, you agree to reimburse each of the Commitment Parties from time to time on written demand accompanied by reasonable supporting detail for all reasonable out-of-pocket fees and expenses (including, but not limited to, (a) the reasonable fees, disbursements and other charges of McGuireWoods LLP, as counsel to BofA Securities and the Administrative Agent, of Moore & Van Allen, PLLC, as counsel to PNC and PNC Capital Markets, and of special and local counsel to the Lenders retained by BofA Securities or the Administrative Agent, and (b) due diligence expenses) incurred in connection with the Amended Credit Facilities, the syndication thereof, the preparation of the definitive documentation therefor and the other transactions contemplated hereby. You acknowledge that we may receive a benefit, including without limitation, a discount, credit or other accommodation, from any of such counsel based on the fees such counsel may receive on account of their relationship with us including, without limitation, fees paid pursuant hereto.

You agree to indemnify and hold harmless each of the Commitment Parties, each Lender and each of their affiliates and their respective partners, officers, directors, employees, agents, trustees, administrators, managers, advisors and other representatives (each, an "Indemnified Party") from and against (and will reimburse each Indemnified Party as the same are incurred for) any and all claims, damages, losses, liabilities and expenses (including, without limitation, the reasonable fees, disbursements and other charges of counsel) that may be incurred by or asserted or awarded against any Indemnified Party, in each case arising out of or in connection with or by reason of (including, without limitation, in connection with any investigation, litigation or proceeding or preparation of a defense in connection therewith) (a) any matters contemplated by this Commitment Letter or any related transaction or (b) the Amended Credit Facilities or any other similar or replacement financings in which the Commitment Parties or any of their Affiliates are involved or any use made or proposed to be made with the proceeds thereof, except to the extent such claim, damage, loss, liability or expense is found in a final, non-appealable judgment by a court of competent jurisdiction to have resulted from such Indemnified Party's gross negligence, willful misconduct or breach in bad faith by such Indemnified Party of its obligations under this Commitment Letter. In the case of an investigation, litigation or proceeding to which the indemnity in this paragraph applies, such indemnity shall be effective whether or not such investigation, litigation or proceeding is brought by you, your equity-holders or creditors or an Indemnified Party, whether or not an Indemnified Party is otherwise a party thereto and whether or not the transactions contemplated hereby are consummated. You also agree that no Indemnified Party shall have any liability (whether direct or indirect, in contract or tort or otherwise) to you or your subsidiaries or affiliates or to your or their respective equity holders or creditors arising out of, related to or in connection with any aspect of the transactions contemplated hereby, except to the extent of direct, as opposed to special, indirect, consequential or punitive, damages determined in a final, non-appealable judgment by a court of competent jurisdiction to have resulted from such Indemnified Party's gross negligence, willful misconduct or breach in bad faith by such Indemnified Party of its obligations under this Commitment Letter. Notwithstanding any other provision of this Commitment Letter, no Indemnified Party shall be liable for any damages arising

from the use by others of information or other materials obtained through electronic telecommunications or other information transmission systems, other than for direct or actual damages resulting from the gross negligence or willful misconduct of such Indemnified Party as determined by a final and non-appealable judgment of a court of competent jurisdiction. Notwithstanding the foregoing, each Indemnified Party shall be obligated to refund and return any and all amounts paid by you under this paragraph to such Indemnified Party for any such losses, claims, damages, liabilities and expenses to the extent it has been determined by a final, non-appealable judgment by a court of competent jurisdiction that such Indemnified Person is not entitled to payment of such amounts in accordance with the terms hereof.

This Commitment Letter, the fee letter among you and the Commitment Parties of even date herewith (the "Fee Letter"), and the contents hereof and thereof are confidential and, except for disclosure hereof or thereof (i) on a confidential basis to your accountants, attorneys and other professional advisors retained by you in connection with the Amended Credit Facilities, (ii) on a confidential basis to the Sellers and their advisors (provided that fee percentages and amounts and economic terms of the market flex provisions in the Fee Letter are redacted to the reasonable satisfaction of BofA Securities and/or the Administrative Agent, as applicable), (iii) of this Commitment Letter and the existence and contents hereof (but not the Fee Letter or the contents thereof, other than the aggregate fee amount contained in the Fee Letter as part of the Projections, pro forma information, or a generic disclosure of aggregate sources and uses related to fee amounts in connection with the Transaction to the extent such disclosure would be customary or required in offering and marketing materials for the Amended Credit Facilities (and only to the extent aggregated with all other fees and expenses in connection with the Transaction and not presented as an individual line item unless required by applicable law, rule or regulation)) in any syndication or other marketing material in connection with the Amended Credit Facilities, or (iv) in any judicial proceeding in connection with the enforcement of your rights hereunder or under the Fee Letter, may not be disclosed by you in whole or in part to any person or entity without our prior written consent; provided, however, it is understood and agreed that you may disclose (x) this Commitment Letter but not the Fee Letter after your acceptance of this Commitment Letter and the Fee Letter, in filings with the Securities and Exchange Commission and other applicable regulatory authorities and stock exchanges and (y) the aggregate amount of the fees or other payments in the Fee Letter (provided, that, the fee percentages and amounts (other than aggregate amounts) and economic terms of the market flex provisions in the Fee Letter are redacted to the reasonable satisfaction of BofA Securities and/or Administrative Agent, as applicable) may be disclosed in public filings, included in projections and pro forma information, in disclosure of aggregate sources and uses and similar generic disclosures. The Commitment Parties hereby notify you that pursuant to the requirements of the USA PATRIOT Act, Title III of Pub. L. 107-56 (signed into law October 26, 2001) (the "Act"), each of them is required to obtain, verify and record information that identifies you and each guarantor under the Amended Credit Facility, which information includes your and their name and address and other information that will allow the Commitment Parties to identify you and each such guarantor in accordance with the Act. As a condition to entering into the Proposed Amendment, the Company must provide all requested KYC/AML information (including, without limitation, Beneficial Ownership Certifications) satisfactory to the Commitment Parties and each Lender.

7. No Fiduciary Duty. You acknowledge that each of the Commitment Parties or their affiliates may be providing financing or other services (including financial advisory services) to parties whose interests may conflict with yours. Each of the Commitment Parties agrees that it will not furnish confidential information obtained from you to any of its other customers and that it will treat confidential information relating to you and your affiliates with the same degree of care as it treats its own confidential information. Each of the Commitment Parties further advises you that it will not make available to you confidential information that it has obtained or may obtain from any other customer. In connection with the services and transactions contemplated hereby, you agree that each of the Commitment Parties is permitted to access, use

and share with any of its bank or non-bank affiliates, agents, advisors (legal or otherwise) or representatives any information concerning you or any of your affiliates that is or may come into the possession of the Commitment Parties or any of such affiliates, in each case subject to the confidentiality provisions contained in this Commitment Letter. You further acknowledge that each of the Commitment Parties is a full service securities or banking firm engaged in securities trading and brokerage activities as well as providing investment banking and other financial services. In the ordinary course of business, any such party may provide investment banking and other financial services to, and/or acquire, hold or sell, for its own accounts and the accounts of other customers, equity, debt and other securities and financial instruments (including bank loans and other obligations) of, you and other companies with which you may have commercial or other relationships. With respect to any securities and/or financial instruments so held by any such party or any of its customers, all rights in respect of such securities and financial instruments, including any voting rights, will be exercised by the holder of the rights, in its sole discretion.

In connection with all aspects of each transaction contemplated by this Commitment Letter, you acknowledge and agree that: (a) (i) the arranging and other services described herein regarding the Amended Credit Facilities are arm's-length commercial transactions between you and your affiliates, on the one hand, and the Commitment Parties, on the other hand, (ii) you have consulted your own legal, accounting, regulatory and tax advisors to the extent you have deemed appropriate, and (iii) you are capable of evaluating, and understand and accept, the terms, risks and conditions of the transactions contemplated hereby; (b) (i) each of the Commitment Parties has been, is, and will be acting solely as a principal and, except as otherwise expressly agreed in writing by the relevant parties, has not been, is not, and will not be acting as an advisor, agent or fiduciary for you, any of your affiliates or any other person or entity and (ii) none of the Commitment Parties has any obligation to you or your affiliates with respect to the transactions contemplated hereby except those obligations expressly set forth herein or in the definitive documentation for the Amended Credit Facilities; and (c) the Commitment Parties and their respective affiliates may be engaged in a broad range of transactions that involve interests that differ from yours and those of your affiliates, and none of the Commitment Parties has any obligation to disclose any of such interests to you or your affiliates. To the fullest extent permitted by law, you hereby waive and release any claims that you may have against any of the Commitment Parties with respect to any breach or alleged breach of agency or fiduciary duty in connection with any aspect of any transaction contemplated by this Commitment Letter.

- 8. <u>Survival</u>. The provisions of the immediately preceding five paragraphs shall remain in full force and effect regardless of whether the definitive documentation for the Amended Credit Facilities shall be executed and delivered, and notwithstanding the expiration or mutual termination of this Commitment Letter or any commitment or undertaking of any of the Commitment Parties hereunder; provided that the reimbursement and indemnity provisions set forth in <u>Section 6</u> of this Commitment Letter shall be superseded on the date of effectiveness of the definitive documentation for the Amended Credit Facilities that includes reimbursement and indemnity provisions. The provisions of <u>Section 3</u> and <u>Section 4</u> hereof shall remain in full force and effect, regardless of whether the definitive documentation for the Amended Credit Facilities shall be executed and delivered, until the earlier of (1) a Successful Syndication (as defined in the Fee Letter) shall have been achieved or (2) the occurrence of the Commitment Expiry Date without a closing of the Amended Credit Facilities having occurred.
- 9. <u>Miscellaneous</u>. This Commitment Letter and the Fee Letter may be executed in counterparts which, taken together, shall constitute an original. Delivery of an executed counterpart of this Commitment Letter or the Fee Letter by electronic transmission or facsimile shall be effective as delivery of a manually executed counterpart thereof. This Commitment Letter and the Fee Letter shall be governed by, and construed in accordance with, the laws of the State of New York; <u>provided</u> that the laws of the State of Delaware shall govern in determining (a) the interpretation of the definition of "Company Material Adverse Effect" (as

defined in Exhibit C) and whether or not a Company Material Adverse Effect has occurred, (b) the accuracy of any Target Acquisition Agreement Representation and whether as a result of any inaccuracy thereof, a condition to your (or any of your affiliates') obligations to close under the Target Acquisition Agreement has not been met or you (or any of your affiliates) have the right (without regard to any notice requirement but giving effect to any applicable cure provisions) to termination your (or any of your affiliates) obligations under the Target Acquisition Agreement and (c) whether the Target Acquisition has been consummated in accordance with the terms of the Target Acquisition Agreement (in each case without giving effect to the principles of conflicts of law thereunder). Each of you and the Commitment Parties hereby irrevocably waives any and all right to trial by jury in any action, proceeding or counterclaim (whether based on contract, tort or otherwise) arising out of or relating to this Commitment Letter, the Fee Letter, the transactions contemplated hereby and thereby or the actions of any of the Commitment Parties in the negotiation, performance or enforcement hereof.

This Commitment Letter and the Fee Letter embody the entire agreement and understanding among the Commitment Parties, you and your affiliates with respect to the Amended Credit Facilities and supersede all prior agreements and understandings relating to the specific matters hereof. No party has been authorized by any of the Commitment Parties to make any oral or written statements that are inconsistent with this Commitment Letter. This Commitment Letter is not assignable by any party hereto without the prior written consent of the other parties hereto and is intended to be solely for the benefit of the parties hereto and the Indemnified Parties. The terms of this Commitment Letter may be modified only pursuant to a writing signed by each party hereto specifically referencing this Commitment Letter.

This Commitment Letter and all commitments and undertakings of each of the Commitment Parties hereunder will expire at 11:00 p.m. (Eastern time) on November 6, 2019 unless you execute this Commitment Letter and the Fee Letter and return them to us prior to that time, whereupon this Commitment Letter and the Fee Letter shall become binding agreements. Thereafter, all commitments and undertakings of each of the Commitment Parties hereunder (except to the extent that any provisions expressly survive termination of this Commitment Letter) will expire on the earliest of (such earliest date referred to herein as the "Commitment Expiry Date"): (a) February 6, 2020 (the "Stated Termination Date"), (b) the closing of the Target Acquisition with or without the use of the Incremental Facilities, (c) the public announcement of the abandonment of the Target Acquisition by you (or any of your affiliates) and (d) the termination of the Target Acquisition Agreement prior to closing of the Target Acquisition or the termination of your (or any of your affiliates') obligations under the Target Acquisition Agreement to consummate the Target Acquisition in accordance with the terms thereof. In consideration of the time and resources that each of the Commitment Parties will devote to the Amended Credit Facilities, you agree that, until the Commitment Expiry Date, you will not, without BofA Securities' consent, solicit, initiate or permit, or enter into any discussions in respect of, any offering, placement or arrangement of any competing senior credit facility or facilities for the Company and its subsidiaries.

[Remainder of page intentionally left blank; signature pages follow]

We are pleased to have the opportunity to work with you in connection with this important financing.

Very truly yours,

# BANK OF AMERICA, N.A.

By:/s/ <u>Tony Keranov</u>
Name: Tony Keranov
Title: Senior Vice President

# **BofA SECURITIES, INC.**

By: /s/ Jonathan Stauning
Name: Jonathan Stauning

Title: Director

# PNC BANK, NATIONAL ASSOCIATION

By: /s/ <u>Brent Pockar</u> Name: Brent Pockar Title: Senior Vice President

# PNC CAPITAL MARKETS LLC

By: /s/ <u>Brian Prettyman</u>
Name: Brian Prettyman
Title: Managing Director

NV5 Global, Inc. 2019 Commitment Letter Signature Page

# ACCEPTED AND AGREED TO AS OF THE DATE FIRST ABOVE WRITTEN:

NV5 GLOBAL, INC., a Delaware corporation

By: /s/ Dickerson Wright

Name: Dickerson Wright

Title: Chairman and Chief Executive Officer

**EXHIBIT A** 

#### TERMS OF TERM FACILITY

Term Facility: A term loan facility in an aggregate principal amount of \$150 million. The Term Facility will be available to the

Company only in a single draw of the full amount of the Term Facility on the Effective Date.

Maturity: December 20, 2023 (same as Revolving Facility under the Existing Credit Agreement).

Applicable Rate: The interest rates per annum applicable to the Term Facility will be the same interest rates applicable to the

Revolving Facility under the Existing Credit Agreement, with a new pricing tier added as Level V in the definition of "Applicable Rate" equal to 2.25% for Eurodollar Rate Loans, 1.25% for Base Rate Loans, and a 0.250%

Commitment Fee when the Consolidated Senior Leverage Ratio is greater than or equal to 3.25 to 1.00.

Amortization: Quarterly amortization of principal in an amount equal to 1.875% of the initial aggregate principal amount of the

Term Facility, with the final payment of all amounts outstanding, plus accrued interest, being due at maturity.

Mandatory Prepayments: In addition to the amortization set forth above, (a) a percentage of all net cash proceeds to be agreed from sales of

property and assets of the Company and its subsidiaries (excluding sales of inventory in the ordinary course of business and other exceptions to be agreed upon in the definitive documentation), and (b) a percentage of all net cash proceeds to be agreed of Extraordinary Receipts (to be defined in the definitive documentation, to include tax refunds, insurance and condemnation proceeds and similar extraordinary items and to exclude cash receipts in the ordinary course of business and proceeds of offerings of equity securities) shall be applied to the prepayment of the

Term Facility (and to the principal installments thereof in inverse order of maturity).

Collateral, Covenants, Representations and Warranties, Events of

Default and other Terms: Same as Revolving Facility under the Existing Credit Agreement.

Other: The Term Facility shall rank pari passu in right of payment priority with the Revolving Facility, and share ratably in

rights in the Collateral and the Guaranty and in a manner consistent with the terms of the Loan Documents.

**EXHIBIT B** 

# PROPOSED AMENDMENT

The documentation for the Amended Credit Facilities (including the Incremental Facilities) will take the form of an amendment of, or amendment and restatement of, the Existing Credit Agreement and the Loan Documents executed in connection therewith, which shall be substantially consistent with the Existing Credit Agreement and the existing Loan Documents, with the modifications set forth below.

(i) Amend Section 2.16 (Increase in Revolving Facility) of the Existing Credit Agreement to (A) provide that an Incremental Facility thereunder may be one or more term loan facilities or an incremental revolving facility, (B) permit the Incremental Facilities to be incurred under Section 2.16 in an amount not to exceed \$240,000,000 (which will be in addition to, and not constitute usage of, the \$100,000,000 maximum aggregate principal amount for all incremental requests currently contained in the Existing Credit Agreement), (C) add customary provisions regarding the terms and conditions of any incremental term loan and to permit the proposed terms in Exhibit A hereto, and (D) provide for customary "limited conditionality" conditions for any incremental facilities incurred in connection with a limited condition acquisition;

- (ii) After giving effect to the amendment in clause (i) above, amend the Existing Credit Agreement to effectuate the Incremental Facilities in accordance with revised Section 2.16, including Section 2.16(f);
- (iii) Amend Section 7.01 (Liens) of the Existing Credit Agreement to permit scheduled Liens (subject to the approval of the Lead Arrangers) upon any of Target's property, assets or revenues existing on the Effective Date and disclosed in the schedules to the Target Acquisition Agreement (but which are not required to be satisfied or removed pursuant thereto nor were incurred in contemplation of the Target Acquisition);
- (iv) Amend Section 7.02 (Indebtedness) of the Existing Credit Agreement to permit scheduled Indebtedness of Target (subject to the approval of the Lead Arrangers) existing on the Effective Date and disclosed in the schedules to the Target Acquisition Agreement (but which is not required to be repaid pursuant thereto nor were incurred in contemplation of the Target Acquisition);
- (v) Amend Section 7.03 (Investments) of the Existing Credit Agreement to specifically permit the Target Acquisition;
- (vi) Waive Section 4.02 (Conditions to All Credit Extensions) of the Existing Credit Agreement with respect to the Acquisition Funds, such that the conditions precedent to the effectiveness of the Proposed Amendment and the funding of the Acquisition Funds on the Effective Date are limited to those set forth on Exhibit C to the Commitment Letter to which this Exhibit B is attached;
- (vii) Amend (A) <u>Section 6.11</u> (Use of Proceeds) of the Existing Credit Agreement to permit the Acquisition Funds to be used to consummate the Target Acquisition, (B) <u>Section 7.04</u> (Fundamental Changes) of the Existing Credit Agreement to permit the Target Acquisition in accordance with the Target Acquisition Agreement, (C) <u>Section 7.05</u> (Dispositions) of the Existing Credit Agreement to permit dispositions to comply with regulatory requirements pursuant to the Target Acquisition Agreement, and (D) <u>Section 7.07</u> (Change in Nature of Business) of the Existing Credit Agreement to permit the business conducted by the Target and its subsidiaries on the Effective Date;
- (viii) Amend the definition of "Applicable Rate" in <u>Section 1.01</u> (Defined Terms) of the Existing Credit Agreement to add a new pricing tier as Level V equal to 2.250% for Eurodollar Rate Loans, 1.250% for Base Rate Loans, and a 0.250% Commitment Fee when the Consolidated Senior Leverage Ratio is greater than or equal to 3.25 to 1.00;
- (ix) Amend the definition of "Change of Control" in <u>Section 1.01</u> (Defined Terms) of the Existing Credit Agreement to remove clause (c) from such definition;
- (x) Amend the definition of "Consolidated EBITDA" in Section 1.01 (Defined Terms) of the Existing Credit Agreement to (A) include an add-back for run-rate cost savings and cost synergies that are projected by the Company in good faith to be realized, within 12 months, as a result of the Target Acquisition in an aggregate amount not to exceed 10% of Consolidated EBITDA (calculated prior to giving effect to any such adjustment) and (B) update the add-back in clause (b)(iv) of such definition for charges and expenses incurred in connection with any permitted Acquisition or Disposition to exclude (and separately permit) up to \$5,000,000 of charges and expenses in connection with the Target Acquisition;
- (xi) Amend the definition of "Fee Letter" in <u>Section 1.01</u> (Defined Terms) of the Existing Credit Agreement to include the Fee Letter;
- (xii) Amend the definition of "Threshold Amount" in <u>Section 1.01</u> (Defined Terms) of the Existing Credit Agreement from "\$5,000,000" to "\$10,000,000";
- (xiii) Amend Section 1.02 (Other Interpretative Provisions) of the Existing Credit Agreement to clarify that references to "wholly-owned" Subsidiaries shall include those entities substantially 100% owned but for any director qualifying shares or similar arrangements;
- (xiv) Amend Section 6.13 (Covenant to Give Security) and Section 6.16 (Deposit Accounts) of the Existing Credit Agreement to specifically permit the Target to provide a Qualifying Control Agreement or to move its existing accounts to the Administrative Agent within 180 days of the Effective Date and to maintain its existing primary depositary and operating accounts for 180 days following the Effective Date;
- (xv) Amend Section 7.11 (Financial Covenants) of the Existing Credit Agreement to (A) change the maximum permitted Consolidated Leverage Ratio from 4.00 to 1.00 to (1) 4.25 to 1.00 at any time on or prior to March 31, 2020 (provided that clause (b)(i) of the definition of "Permitted Acquisition" shall also be amended to require that the pro forma Consolidated Leverage Ratio be no greater than the lesser of (x) 0.25 to 1.00 less than the applicable maximum Consolidated Leverage Ratio in Section 7.11(a) and (y) 3.75 to 1.00), (2) 4.00 to 1.00 at any time after March 31, 2020 and on or prior to June 30, 2020, (3) 3.75 to 1.00 at any time after June 30, 2020 and on or prior to September 30, 2020, and (4) 3.50 to 1.00 at any time

- after September 30, 2020, and (B) remove the maximum permitted Consolidated Senior Leverage Ratio covenant in clause (b) of such Section;
- (xvi) Amend <u>Section 8.01(m)</u> (Uninsured Loss) of the Existing Credit Agreement to increase the threshold amount in such Section from "\$5,000,000" to "\$10,000,000"; and
- (xvii) Add customary provisions reflecting the Administrative Agent's policy concerning market standards, including provisions regarding QFC stay rule.

**EXHIBIT C** 

#### CONDITIONS PRECEDENT

Capitalized terms not otherwise defined herein have the same meanings as specified therefor in the Commitment Letter (the "Commitment Letter") to which this Exhibit C is attached and forms part of.

The effectiveness of the Amended Credit Agreement and the extension of the Acquisition Funds under the Amended Credit Agreement will be subject to satisfaction of the following conditions precedent, in each case subject to the Limited Conditionality Provision:

- (i) <u>Definitive Documentation</u>. The negotiation, execution and delivery of definitive documentation for the Incremental Facilities and the Proposed Amendment, in each case consistent with the Commitment Letter and reasonably satisfactory to the Lead Arrangers, the Company and the Administrative Agent; it being understood that the Existing Credit Agreement and related existing Loan Documents shall be used as the basis for preparing such definitive documentation, and such documents shall be amended or amended and restated, as necessary, consistent with the Commitment Letter and reasonably satisfactory to the Lead Arrangers, the Company and the Administrative Agent.
- (ii) Specified Representations; Target Acquisition Agreement Representations. The Specified Representations shall be true and correct in all material respects (or, with respect to representations already qualified by concepts of materiality, in all respects) on and as of the Effective Date after giving effect to the Transactions. The Target Acquisition Agreement Representations shall be true and correct on and as of the Effective Date. As used herein and in the Commitment Letter, the term "Target Acquisition Agreement Representations" shall mean such of the representations and warranties made by or on behalf of the Sellers with respect to the Target or Target's subsidiaries in the Target Acquisition Agreement (defined below) as are material to the interests of the Administrative Agent, the Lead Arrangers and the Lenders (in their capacities as such), but only to the extent that the Company (or any of its affiliates) has the right (determined without regard to any notice requirement) to terminate the Company's (and, if applicable, its affiliates') obligations (or to refuse to consummate the Target Acquisition) under the Target Acquisition Agreement as a result of a breach of any of such representations and warranties.
- (iii) No Default. No Default or Event of Default shall have occurred and been continuing under the Existing Credit Agreement on the date of the Commitment Letter, and no Default or Event of Default shall exist under any of Sections 8.01(a), 8.01(f) and 8.01(g) of the Existing Credit Agreement on the Effective Date.
- (iv)<u>Legal Opinions</u>. The Administrative Agent shall have received customary opinions of counsel to the Loan Parties (including the New Guarantors (defined below)) reasonably acceptable to the Administrative Agent.
- (v)Organizational Documents; Resolutions; Good Standing Certificates. The Administrative Agent shall have received each of the following with respect to each Loan Party (including the New Guarantors), in each case certified by a secretary or assistant secretary of such Loan Party (including the New Guarantors) to be true and correct as of the Effective Date: (i) copies of its articles of incorporation or formation (or equivalent document) certified to be true and complete as of a recent date by the appropriate governmental authority of the state of its incorporation or organization, (ii) copies of its bylaws or operating agreement (or equivalent document), (iii) resolutions and incumbency certificates and (iv) a good standing certificate (or its equivalent) dated as of a recent date by the appropriate governmental authority of the state of its incorporation or organization, in each case, reasonably satisfactory to the Administrative Agent.
- (vi) Financial Statements. The Lead Arrangers shall have received (A) audited consolidated balance sheets and related statements of income, changes in equity and cash flows of the Company and its subsidiaries for the three most recent fiscal years ended at least 90 days before the Effective Date; (B) unaudited consolidated balance sheets and related statements of income, changes in equity and cash flows of the Company and its subsidiaries for each of the fiscal quarters ended after the last audited financial statements required pursuant to clause (A) but at least 45 days before the Effective Date; (C) audited consolidated balance sheets and related statements of income, changes in equity and cash flows of the Target and its subsidiaries for the three most recent fiscal years ended at least 90 days before the Effective Date; (D) unaudited consolidated balance sheets and related statements of income, changes in equity and cash flows of the Target and its subsidiaries for each of the fiscal quarters ended after the last audited financial statements required pursuant to clause (C) but at least 45 days before the Effective Date; and (E) a pro forma

consolidated balance sheet and related pro forma consolidated statement of income of the Company and its subsidiaries as of and for the most recent four fiscal quarter period ended at least 45 days (or, if such period is a fiscal year end, at least 90 days) prior to the Effective Date prepared after giving effect to the Transactions as if the Transactions had occurred as of such date (in the case of such balance sheet) or at the beginning of such period (in the case of such statement of income).

#### (vii)Target Acquisition.

- (A) <u>Target Acquisition Agreement</u>. To the extent not previously delivered, the Lead Arrangers shall have received true and correct fully-executed copies of the Target Acquisition Agreement (including all amendments or modifications thereto) and all material documentation for the Target Acquisition and the Transactions, each of which shall be in form and substance reasonably satisfactory to the Lead Arrangers (it being understood that the Target Acquisition Agreement executed by the parties on November 6, 2019 is satisfactory to the Lead Arrangers). "*Target Acquisition Agreement*" means that certain Agreement and Plan of Merger dated as of November 6, 2019 by and among the among the Company, Arlington Capital Partners III, L.P., a Delaware limited partnership, as representative for the Sellers (as defined therein), the Target and Merger Sub (as defined therein), along with all schedules and exhibits attached thereto.
- (B) Target Acquisition Consummation. The Target Acquisition shall have been, or shall concurrently with the effectiveness of the Amended Credit Agreement and the funding of the Acquisition Funds will be, consummated in accordance with the terms of the Target Acquisition Agreement, without giving effect to any amendment, modification, waiver or consent thereunder that is materially adverse to the interests of the Lenders unless such amendment or modification is approved by the Lead Arrangers. For purposes of the foregoing condition, it is hereby understood and agreed that (i) amendments, waivers and other changes to the definition of "Company Material Adverse Effect" (as defined in the Target Acquisition Agreement) (and consents and requests given or made pursuant to such definition), the "Xerox" provisions of the Target Acquisition Agreement or the governing law provisions in the Target Acquisition Agreement shall, in each case, be deemed to be materially adverse to the interests of the Lenders so long as such reduction is allocated dollar-for-dollar to reduce the principal amount of the Term Facility, (iii) any increase in the purchase price in connection with the Target Acquisition shall not be deemed to be material and adverse to the interests of the Lenders to the extent that such increase is not funded with indebtedness and (iv) any amendment, modification or waiver of (A) the definition of "R&W Insurance Policy" in the Target Acquisition Agreement, (B) the requirement to deliver the R&W Insurance Policy on or prior to the consummation of the Target Acquisition set forth in Section 2.4 of the Target Acquisition Agreement and/or (C) the covenant with respect to the R&W Insurance Policy set forth in Section 5.8 of the Target Acquisition Agreement shall, in any such case, be deemed to be materially adverse to the interests of the Lenders.
- (B) <u>Company Material Adverse Effect</u>. No Company Material Adverse Effect (as such term is defined in the Target Acquisition Agreement) shall have occurred from September 28, 2019 to and including the Effective Date.
- (C) <u>Refinancing of Third Party Bank Debt of Target and its Subsidiaries</u>. On the Effective Date (or concurrent with the funding of the Acquisition Funds) all third party indebtedness of the Target and its Subsidiaries shall have been repaid, all commitments thereunder shall have been terminated and all liens and security interests securing such indebtedness shall have been released or terminated, including all indebtedness and commitments outstanding under that certain Credit Agreement dated as of September 5, 2019 by and among Quantum Spatial, Inc. (a subsidiary of the Target), as borrower, Twin Brook Capital Partners, LLC as agent, and the lenders party thereto, and all liens and security interests securing any indebtedness thereunder.
- (viii) <u>Guaranty Joinder</u>. The Target and each of its Subsidiaries (and any new Subsidiaries of the Company formed in connection with the Target Acquisition) that are required to become a Guarantor under <u>Section 6.12</u> of the Existing Credit Agreement (collectively, the "<u>New Guarantors</u>") shall enter into a Joinder Agreement in form and substance reasonably satisfactory to the Administrative Agent to become a "Guarantor" under the Credit Agreement.
- (ix)Collateral. The Loan Parties (including the New Guarantors) shall enter into such amendments to, or amendments and restatements of, the Collateral Documents as reasonably requested by the Administrative Agent in form and substance reasonably satisfactory to Administrative Agent; it being understood that the terms of such Collateral Documents shall be substantially the same as the existing Collateral Documents relating to the Existing Credit Agreement, with such changes as are necessary in connection with the creation, continuation and/or perfection and priority of the security interest in the assets of the Company and its subsidiaries (including the Target and its subsidiaries) after giving effect to the Transactions. All actions necessary to establish that the Administrative Agent will have a perfected security interest in the Collateral of the Loan Parties (including the New Guarantors) under the Amended Credit Facilities shall have been taken.
- (x) <u>Solvency Certificate</u>. The Administrative Agent shall have received certification in the form of <u>Annex I</u> hereto from the chief financial officer of the Company as to the solvency of the Company and its subsidiaries on a consolidated basis as of the Effective Date after giving effect to the Transactions.
- (xi)<u>Inside Date</u>. The Closing Date shall not occur prior to December 20, 2019.
- (xii)<u>Clear Market</u>. There shall be no offering, placement or arrangement of any debt securities or bank financing by or on behalf of the Company or any of its Subsidiaries other than the Amended Credit Facilities contemplated herein.

- (xiii)KYC Information. Upon the reasonable request of any Lender made at least ten (10) days prior to the Effective Date, the Company shall have provided to such Lender the documentation and other information so requested in connection with applicable "know your customer" and anti-money-laundering rules and regulations, including the PATRIOT Act, in each case at least five (5) days prior to the Effective Date; provided that, at least five (5) days prior to the Effective Date, if the Company qualifies as a "legal entity customer" under 31 C.F.R. § 1010.230, then the Company shall deliver to any Lender that makes a request therefor a certification regarding beneficial ownership required by 31 C.F.R. § 1010.230.
- (xiv)<u>Loan Notice</u>. The Administrative Agent shall have received a customary borrowing notice and other customary closing certificates with respect to the borrowing of Acquisition Funds under each of the Revolving Facility and the Term Facility.
- (xv)<u>Payment of Fees and Expenses</u>. All fees payable to the Lead Arrangers and the Lenders on or before the Effective Date pursuant to the Fee Letter shall have been paid; and all expenses of the Lead Arrangers and the Administrative Agent required to be reimbursed by the Company pursuant to the Commitment Letter or the Existing Credit Agreement shall have been reimbursed to the extent invoiced in reasonable detail at least one (1) business day in advance of the Effective Date.

#### ANNEX I

# Form of Solvency Certificate

TO: Bank of America, N.A., as Administrative Agent

RE: Credit Agreement, dated as of December 7, 2016, by and among NV5 Global, Inc., a Delaware corporation (the "*Company*"), the Guarantors, the Lenders and Bank of America, N.A., as Administrative Agent, L/C Issuer and Swingline Lender (as amended, modified, extended, restated, replaced, or supplemented from time to time, the "*Credit Agreement*"; capitalized terms used herein and not otherwise defined shall have the meanings set forth in the Credit Agreement)

DATE: [Date]

The undersigned Chief Financial Officer of the Company is familiar with the properties, businesses, assets and liabilities of the Loan Parties and is duly authorized to execute this certificate on behalf of the Company and the other Loan Parties.

The undersigned certifies that [he/she] has made such investigation and inquiries as to the financial condition of the Loan Parties and their Subsidiaries as the undersigned deems necessary and prudent for the purpose of providing this Solvency Certificate (this "Certificate"). The undersigned acknowledges that the Administrative Agent and the Lenders are relying on the truth and accuracy of this Certificate in connection with the making of Credit Extensions and the other transactions contemplated under the Credit Agreement on the date hereof.

The undersigned certifies that the financial information, projections and assumptions which underlie and form the basis for the representations made in this Certificate were reasonable when made and were made in good faith and continue to be reasonable as of the date hereof.

BASED ON THE FOREGOING, the undersigned certifies on behalf of the Company that, both before and after giving effect to the consummation of the Target Acquisition (including the consummation of the Target Acquisition Agreement and the extensions of credit to be made under the Credit Agreement, the making of the Loans and the use of proceeds of such Loans on the date hereof):

- (a) The fair value of the property of the Company, together with its Subsidiaries on a Consolidated basis, is greater than the total amount of liabilities, including contingent liabilities, of the Company together with its Subsidiaries on a Consolidated basis.
- (b) The present fair salable value of the assets of the Company, together with its Subsidiaries on a Consolidated basis, is not less than the amount that will be required to pay the probable liability of the Company, together with its Subsidiaries on a Consolidated basis, on its debts as they become absolute and matured.
- (c) The Company, together with its Subsidiaries on a Consolidated basis, does not intend to, and does not believe that it will, incur debts or liabilities beyond such Person's individual or consolidated ability to pay such debts and liabilities as they mature.
- (d) The Company is not engaged in business or a transaction, and the Company is not about to engage in business or a transaction, for which the Company's property would constitute an unreasonably small capital.
- (e) The Company, together with its Subsidiaries on a Consolidated basis, is able to pay its individual and consolidated debts and liabilities, contingent obligations and other commitments as they mature in the ordinary course of business.

(f) The amount of contingent liabilities at any time have been computed as the amount that, in the light of all the facts and circumstances existing at such time, represents the amount that can reasonably be expected to become an actual or matured liability.

Delivery of an executed counterpart of a signature page of this Certificate by fax transmission or other electronic mail transmission (e.g. "pdf" or "tif") shall be effective as delivery of a manually executed counterpart of this Certificate.

[Remainder of page intentionally left blank; signature page follows]

IN WITNESS WHEREOF, I have executed this Certificate as of the date first written above.

NV5 GLOBAL, INC.

By:\_\_\_\_\_

Title: Chief Financial Officer

NV5 Global, Inc. 2019 Commitment Letter Signature Page

122724333 8

#### **CERTIFICATION**

#### I, Dickerson Wright, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q for the period ended September 28, 2019 of NV5 Global, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 7, 2019

/s/ Dickerson Wright

Dickerson Wright Chairman & Chief Executive Officer, (Principal Executive Officer)

#### **CERTIFICATION**

#### I, Edward Codispoti, certify that:

- 1. I have reviewed this Quarterly Report on Form 10-Q for the period ended September 28, 2019 of NV5 Global, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
  - a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
  - b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
  - c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
  - d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
  - a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
  - b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: November 7, 2019

/s/ Edward Codispoti

Edward Codispoti Chief Financial Officer (Principal Financial Officer)

# CERTIFICATION PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Quarterly Report of NV5 Global, Inc. (the "Company") on Form 10-Q for the quarter ended September 28, 2019, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), Dickerson Wright, Chief Executive Officer of the Company, and Edward Codispoti, Chief Financial Officer of the Company, each certify, to the best of his knowledge, pursuant to Rule 13a-14(b) or 15d-14(b) of the Securities Exchange Act of 1934, as amended, and 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that:

- (1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934, as amended; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: November 7, 2019

/s/ Dickerson Wright

Dickerson Wright

Chairman & Chief Executive Officer

Date: November 7, 2019

/s/ Edward Codispoti

Edward Codispoti Chief Financial Officer

This certification accompanies this Quarterly Report on Form 10-Q pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 and shall not, except to the extent required by such Act, be deemed filed by the Company for purposes of Section 18 of the Securities Exchange Act of 1934, as amended (the "Exchange Act"). Such certification will not be deemed to be incorporated by reference into any filing under the Securities Act of 1933, as amended, or the Exchange Act, except to the extent the Company specifically incorporates it by reference.

A signed original of this written statement required by Rule 13a-14(b) or 15d-14(b) of the Exchange Act and Section 906 has been provided to the Company and will be retained by the Company and furnished to the Securities and Exchange Commission or its staff upon request.